The effect of social deprivation on distal radius fracture incidence and treatment

Nicholas Alan Johnson MBCHB (Hons), FRCS (orth)

Submitted in fulfilment of the requirements for the Degree of PhD

June 2018

The effect of social deprivation on distal radius fracture incidence

and treatment

Nicholas Alan Johnson

Aim: Socioeconomic deprivation is known to be associated with many types of fracture but the reasons for this are not clearly understood. The primary aim of this thesis is to investigate if deprivation is associated with distal radius fracture and does it influence distal radius fracture treatment.

Methods: Regression modelling techniques are used to analyse four years of Leicestershire emergency department data (n=4278) and national Hospital Episodes Statistics data (n=59,315) from a similar period. Radiographic analysis and a prospective survey is carried out to identify factors responsible for the effects of deprivation on fracture incidence. Systematic review and meta-analysis is performed to quantify the risk of hip fracture after distal radius fracture.

Results: Deprivation was strongly associated with distal radius fracture in the whole population studied. Deprived patients sustained their injuries at an earlier age. The influence of deprivation was larger in white men. Incidence rate in the least deprived quintile was a third of that in the most deprived for white men and almost half for white women. Age 50 years and over and male gender was an independent risk factor for distal radius fracture in all ethnicities. Falls risk was associated with increasing deprivation. Significant variation is identified between hospital trusts treating distal radius fractures with a higher fixation rate seen in trusts with a smaller population. Deprivation was not related to any of the parameters of hospital care studied.

Discussion: Poverty is rising which will lead to higher levels of social deprivation and an increase in distal radius fracture rate. Fracture prevention strategies are required to combat this rise. Knowing which patients are at highest risk allows interventions to be efficiently targeted. Resources should be targeted to those at-risk patients from deprived areas and preventative strategies put in place.

Acknowledgements

I would like to thank my supervisor Professor Joe Dias for his guidance, planning, accessibility, support and kind advice throughout my PhD studies. I also wish to thank my co-supervisor Professor John Thompson for his support, availability and expert statistical advice.

I would also like to acknowledge the support and advice I received from the Leicester hand surgeons; Bhaskar Bhowal, Lucy Cutler, Aamer Ullah and Clare Widin.

Thanks to Elaine and the rest of the Academic Team of Musculoskeletal Surgery; Suzy, Chris and Jared, for their help, humour and constant cups of tea.

Euan, John, Neville and Nalin all worked extremely hard and diligently as members of the research team for parts of the work and I hope they found the experience valuable and that it may have stimulated interest in research for them.

Finally, many thanks are due to my wife and children for their patience and support over the years that this work was taking place.

Contents

List of ta	ables	9
List of fig	gures	12
Abbrevia	ations	15
Relevant	t publications	17
Chapter	1 Introduction	
1.1	The history of deprivation and health	
1.2	Measures of deprivation	
1.2.1	1 Index of Multiple Deprivation	20
1.2.2	2 Other measures of deprivation	23
1.2	2.2.1 Jarman score	23
1.2	2.2.2 Townsend Material Deprivation Score	23
1.2	2.2.3 Carstairs score	
1.3	Social deprivation and health	
1.3.1	1 Social deprivation and injury	
1.3.2	2 Deprivation and trauma	27
1.3.3	3 Adolescent fractures and deprivation	
1.3.4	4 Paediatric fractures and deprivation	
1.3.5	5 Distal radius fractures and deprivation	
1.3.6	6 Hand and upper limb fractures and deprivation	
1.4	Deprivation and osteoporosis	
1.5	Deprivation and falls	
1.6	Deprivation and access to health care	
1.7	Deprivation and health summary	
1.8		
1.8.1	1 Definition of osteoporosis	
1.8.2	2 Diagnosis and measurement of osteoporosis	
1.8.3	B Fragility fracture	
1.8	8.3.1 Hip fracture	
1.0 10	Distal radius fractures	
101	Enidemiology of distal radius fractures	43
1.9.1	2 Management of distal radius fractures	
1.5.2	9 2 1 Non -operative management	45
1.9	9.2.2 Percutaneous Pinning (Kirschner wires)	
1.9	9.2.3 Open Reduction and Internal Fixation	
1.9	9.2.4 External Fixation	47
1.9.3	3 DRAFFT (Distal Radius Acute Fracture Fixation Trial)	48
1.10	Distal radius fracture radiographic parameters	
1.10	0.1 Radiographic parameters and outcome	49
1.11	Variation in surgical treatment	
1.12	Quality of care	53
1.13	Aims of thesis	53
1.14	Overview of thesis	54
Chapter	2 Method	56

2.1	1 Settings and population56		
2.2	2.2 Epidemiology of distal radius fracture in Leicestershire		
2.2.	1	Patient identification	57
2.3	Effe	ct of deprivation on distal radius fracture rate in Leicestershire	58
2.3.	1	Index of Multiple Deprivation	59
2.3.	2	Incidence rate and deprivation guintile	59
2.3.	3	Surgical treatment	60
2.4	Radi	ographic measurements and deprivation	60
2.4.	1	Reliability and reproducibility of radiographic measurements	61
2	.4.1.1	Extra articular radiographic measurements	61
2	.4.1.2	Intra articular radiographic measurement	63
2.4.	2	Metacarpal Cortical Index	65
2	.4.2.1	MCI measurement	65
2	.4.2.2	MCI and bone fragility	66
2.5	Risk	of hip fracture after distal radius fracture	68
2.5.	1	Study group	68
2.5.	2	Search strategy and data sources	69
2.5	3	Study selection	
2.5	Δ	Data extraction and quality assessment	70
2.5.	5	Statistical analysis	70
2.5.	Nati	onal data	70 71
2.0	1	Hospital Enisode Statistics data	71
2.0.	יד ר	Diagnosis	/ 1 71
2.0.	2	Diagnosis.	1 /
2.6.	.3	Deprivation, ethnicity and rate of fixation	
2.6.	-	Analysis of variation	/3
2.6.	.5	Temperature	73
2.6.	.6	Hospital trust data	73
2.6.	7	Comorbidities and Charlson index	74
2.6.	8	Semi-elective trauma surgery	74
2	.6.8.1	NICE guidance	74
2.6.	9	Hospital Episode Statistics analysis	75
2.7	Pros	pective investigation of fracture risk	76
2.7.	1	Survey method	76
2.7.	2	Survey design	76
2	.7.2.1	Patient-Reported Wrist Evaluation (PRWE)	77
2	.7.2.2	EuroQol EQ-5D-5L	77
2	.7.2.3	FRAX [®]	78
2	.7.2.4	Falls risk assessment	78
2	.7.2.5	Analysis	79
2.8	Stat	istical analysis	79
2.8.	.1	Modelling	80
2	.8.1.1	Linear regression	80
2	.8.1.2	Generalised linear models	80
2	.8.1.3	Poisson regression	81
2	.8.1.4	KODUST STANDARD ERFORS	81
2	.8.1.5		82
2 ר	0.1.0 Q 1 7	Lugishi regression	2ة دو
2	81 x	Generalised linear models interpretation	20 دي
2	.0.1.0	Selectanded intent models interpretation and and and and and and and and and an	

2.8	8.1.9 Random effects	83
2.8	8.1.10 Overdispersion	
2.8	8.1.11 Under dispersion	85
2.8	8.1.12 Generalised additive models	
2.8	8.1.13 Model selection	
2.8.2		
2.8.3	3 Prediction intervals	
2.8.4	4 Other tests	8/
2.9	Permissions	8/
Chapter	3 Reliability and reproducibility of radiographic measurements	88
3.1	Extra-articular radiographic measurements	88
3.2	Intra-articular radiographic measurements	90
3.3	Metacarpal Cortical Index (MCI)	91
3.3.1	1 MCI measurement	91
3.3.2	2 MCI and age and fracture displacement	92
3.3.3	3 MCI and bone fragility	96
3.4	Summary of radiographic measurement results	98
Chapter	4 Epidemiology of distal radius fracture in Leicestershire	100
4.1	Population changes	100
4.2	Trends in distal radius fracture incidence	101
4.2.1	1 Overall Fracture Incidence	101
4.2.2	2 Age and Gender Specific Incidence	105
4.2.3	3 Seasonal Variation of Fracture Incidence	106
4.2.4	4 Weekly Variation in Fracture Incidence	107
4.3	Fracture Projection Analysis	109
Chapter	5 Effect of deprivation on distal radius fracture rate in Leicestershi	re.112
5.1	Total numbers of fractures	112
5.1.1	1 Observed and expected fracture rate	113
5.1.2	2 Deprivation and age of distal radius fracture patients	116
5.2	Relationship between deprivation and radiographic parameters	117
5.2.1	1 Dorsal tilt	117
5.2.2	2 Intra-articular step	118
5.2.3	3 Metacarpal Cortical Index	118
5.3	Patients with distal radius fracture aged 50 years and over	121
5.3.1	1 Relationship between deprivation and radiographic parameters in patient	nts
aged	d 50 years and over	125
5.3	3.1.1 Dorsal tilt	125
5.3	3.1.2 Intra-articular step	126
5.3	3.1.3 Metacarpal Cortical Index	
5.4	Patients with distal radius fracture aged between 18 and 49 years	128
5.4.1	Keiationship between deprivation and radiographic parameters in adult	
patie	ents aged between 18 and 49 years	132
5.4	4.1.1 Dorsal tilt	
5.4	4.1.2 Intra-articular step	133
5.4 5 5	Regression model results	134 124
5.5	Fthnicity age and radiographic findings	129
5.0	- connervy, age and radiographic infunity	T20

5.6.	.1 Patients aged 50 years and over	139
5	.6.1.1 Dorsal tilt	139
5	.6.1.2 MCI	
5.6.	.2 Patients aged between 18 and 49 years	143
5	.6.2.1 Dorsal tilt	143
5	.6.2.2 MCI	144
5.6.	.3 Comparison between ethnic groups	146
5.6.	.4 White patients	147
5	.6.4.1 Dorsal tilt	147
5	.6.4.2 MCI	148
5.6.	.5 Patients of other ethnicity	149
5	.6.5.1 Dorsal tilt	149
5	.6.5.2 MCI	
5.7	Effect of deprivation and ethnicity on national distal radius fracture surger	y rate
	151	
5.7.	.1 Patients aged under 50 years	151
5.7.	.2 Patients aged 50 years and over	151
Chanter	r 6 Influence of deprivation on treatment	154
6 1	Surgical fixation rate in Leicestershire	15/
6.2	National data Modele	154
0.2		155
0.2.		
6.2.	2 Wait for semi elective trauma surgery	
6.2.	.3 Daycase	
6.2.	A Pre operative wait for surgery	
6.2.	.5 Post operative stay	
6.3	National Variation	161
Chapter	r 7 Prospective investigation of fracture risk	165
7.1	Mechanism of fall	166
7.2	Osteoporosis risk	167
7.3	Falls risk assessment	169
Chapter	r 8 Risk of distal radius fracture after hip fracture	172
Chapter	r 9 Discussion	179
9.1	Accuracy of radiographic measurement	180
9.2	Measurement of displacement	180
9.2.	1 Extra-articular radiographic parameters	
9.2	2 Intra-articular radiographic parameters	181
9.3	Measurement of hone health	183
94	Is distal radius fracture associated with a higher risk of subsequent hin frac	turo?
5.4	190	cure.
0.4	1 Meta analysis	100
9.4. Q E	Enidemiology of dictal radius fracture in Leicestershire – a 10 year review	190 102
9.5	The influence of denrivation on distal radius fracture	193 107
J.U 0 6	1 Effect of deprivation on distal radius fracture rate	107
9.0. 0.6	2 Effect of deprivation on radiographic mascurements	100
9.0.		
~ ~ ~	The influence of deprivation on distal redive for store two states	190
9.7	The influence of deprivation on distal radius fracture treatment	198

9.7.2	2 Pi	e-and post-operative wait to surgery	201
9.7.3	3 D	ay case surgery	202
9.8	Nation	al variation in fixation rate	202
9.9	Investi	gation of fracture risk	205
9.9.1	1 0	steoporosis and deprivation	206
9.9.2	2 Fa	Ills risk and deprivation	207
9.10	Burde	n of falls	208
9.11	Increa	se of deprivation	210
9.12	Furthe	r projections and cost	
9.13	Fractu	re prevention strategies	212
Chapter	10	Summary	216
Chapter	11	Future directions	218
Appendi	ix 1 Sys	tematic review protocol	221
Appendi	ix 2 Sys	tematic review data collection form	223
Append	ix 3 Mo	dified Coleman score assessment	229
Append	ix 4 Ris	k of bias tool	230
Appendi	ix 5 Fra	cture risk assessment questionnaire	231
Appendi	ix 6 STI	ADI falls risk assessment	238
Appendi	ix 7 R p	ackages	240
Append	ix 8 Na	tional regression model results	241
Appendi	ix 9 Sui	gical fixation rate in Leicester	
Referen	ces		245

List of tables

Table 1 Measurement of inter-observer correlation between the two observers with
95% confidence intervals for common extra-articular radiographic parameters .89
Table 2 Intra-observer correlation between the two observers with a 10% retest
sample 4 weeks after initial measurements for common extra-articular
radiographic parameters89
Table 3 Measurement of inter-observer correlation between the two observers with
95% confidence intervals (whole sample) for intra-articular radiographic
Table 4 later observer correlation between the two observers with 05% confidence
intervale (freatures with displacement > 2mm) for intra erticular rediographic
intervals (fractures with displacement 2 2mm) for intra-articular radiographic
parameters
Table 5 Intra-observer correlation between the two observers with a 10% re-test
sample four weeks after initial measurements for intra-articular radiographic
parameters
Table 6: Intra-observer Correlation with 95% Confidence Intervals (CI) for MCI
measurement
Table / Comparative adult population profiles of Leicestershire and the whole United
Kingdom
Table 8: Total number of distal radius fractures and incidence (expressed per 100,000
person/years) in Leicestershire from 2007 to 2016102
Table 9 Mean (10-year) distal radius fracture incidence rates in Leicestershire by
gender and age group (expressed per 100,000 person/years)104
Table 10 Mean number and rate of distal radius fracture in Leicestershire (expressed
per 100,000 person/years) by gender and deprivation quintile from 2007 to 2010
Table 11 Observed number of distal radius fracture seen in Leicestershire and
expected number of distal radius fracture in the region for each deprivation
quintile from 2007 to 2010114
Table 12 Mean number of fractures per year and rate per deprivation quintile
(expressed per 100,000 person/years) in patients in Leicestershire aged 50 years
and over from 2007 to 2010121
Table 13 Observed number of distal radius fracture seen in Leicestershire and
expected number of distal radius fracture in the region for each deprivation
quintile from 2007 to 2010 in patients aged 50 years and over122
Table 14 Mean number of fractures per year and rate per deprivation quintile
(expressed per 100,000 person/years) in adult patients in Leicestershire aged
under 50 from 2007 to 2010128
Table 15 Observed number of distal radius fracture seen in Leicestershire and
expected number of distal radius fracture in the region for each deprivation
quintile from 2007 to 2010 in adult patients aged under 50 years
Table 16 P values obtained for each gender ethnicity subgroup from the generalized
additive model investigating relationship between IMD rank and dorsal tilt140

Table 17 Results of the comparison of mean dorsal tilt between the two most
deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each
gender ethnicity subgroup140
Table 18 P values obtained for each gender ethnicity subgroup from the generalized
additive model investigating relationship between IMD rank and MCI141
Table 19 Results of the comparison of mean MCI between the two most deprived
quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender
ethnicity subgroup142
Table 20 Results of the comparison of mean MCI between ethnicity groups of the
same gender in the two most deprived quintiles (Q1-2) and the three least
deprived quintiles (Q3-5)142
Table 21 P values obtained for each gender ethnicity subgroup from the generalized
additive model investigating relationship between IMD rank and dorsal tilt in
adult patients aged under 50 years144
Table 22 Results of the comparison of mean dorsal tilt between the two most
deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each
gender ethnicity subgroup in adult patients aged under 50 years
Table 23 P values obtained for each gender ethnicity subgroup from the generalized
additive model investigating relationship between IMD rank and MCI in adult
patients aged under 50 years145
Table 24 Results of the comparison of mean MCI between the two most deprived
quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender
ethnicity subgroup in adult patients aged under 50 years
Table 25 Patient and hospital trust factors which interact significantly with
deprivation (measured by IMD rank) in patients who required surgical treatment
for an acute distal radius fracture156
Table 26 Patient and hospital trust factors which interact significantly with wait for
semi elective trauma surgery in patients who required surgical treatment for an
acute distal radius fracture157
Table 27 Patient and hospital trust factors which interact significantly with the
likelihood of undergoing daycase surgery in patients who required surgical
treatment for an acute distal radius fracture159
Table 28 Patient and hospital trust factors which interact significantly with increasing
wait for surgery in patients who required surgical treatment for an acute distal
radius fracture160
Table 29 Patient and hospital trust factors which interact significantly with post
operative length of stay in patients who required surgical treatment for an acute
distal radius fracture161
Table 30: Negative binomial regression analysis of the relationship between hospital
trust factors and surgical fixation rate in patients who required surgical treatment
for an acute distal radius fracture163
Table 31 Mean values for each osteoporosis and falls risk score for patients in the two
most deprived quintiles(Q1-2) compared to those in the three least deprived
quintiles(Q3-5)168
Table 32 P values from individual regression models investigating the relationship
between osteoporosis risk factors and deprivation (as measured by IMD rank),
age and gender168

Table 33 P values from individual regression models investigating the relationship
between falls risk factors and deprivation (as measured by IMD rank), age and
gender170
Table 34 Description of Studies Included in the Meta-Analysis - Combined results for
men and women174
Table 35 Description of Studies Included in the Meta-Analysis – Women only
Table 36 Description of Studies Included in the Meta-Analysis – Men only176
Table 37 MCI values corresponding to T-score values used to diagnose osteopenia and
osteoporosis186

List of figures

Figure 1 Method of measurement for the extra-articular radiographic parameters of
dorsal tilt, radial inclination and radial neight
Figure 2 Lateral radiograph showing measurement of intra-articular step-off (AC) and
gap (BC) using the longitudinal axis method
Figure 3 Plain film AP radiograph of a distal radius and hand demonstrating how ICD
and OCD measurements are taken from the 2nd Metacarpal
Figure 4: Relationship between MCI and age in patients with distal radius fracture93
Figure 5: Relationship between MCI and age by gender in patients with distal radius
fracture94
Figure 6: Association between age and decreasing MCI in patients over 50 years of age
with distal radius fracture95
Figure 7: MCI and dorsal angulation of distal radius fractures in patients aged 50 years
and over96
Figure 8: Relationship between MCI and BMD in consecutive patients who had
undergone radiographs demonstrating the 2 nd metacarpal and a DXA scan97
Figure 9: Relationship between MCI and T-score in consecutive patients who had
undergone radiographs demonstrating the 2 nd metacarpal and a DXA scan98
Figure 10: Annual distal radius fracture incidence in Leicestershire from 2007 to 2016
by gender and age group103
Figure 11 Total number of distal radius fracture in Leicestershire stratified
cumulatively by gender and age from 2007 to 2016
Figure 12 Gender and age specific distal radius fracture incidence rates in
Leicestershire from 2007 to 2016
Figure 13 Seasonal variation of distal radius fracture incidence by gender and age
group in Leicestershire from 2007 to 2016
Figure 14 Weekly variation of distal radius fracture incidence by gender and age group
in Leicestershire from 2007 to 2016
Figure 15 Mean age of distal radius fracture patients each weekday in Leicestershire
from 2007 to 2016
Figure 16 Projected number of distal radius fracture per year in the United Kingdom
hy gender and age group from 2016 to 2036
Figure 17 Comparison of the expected and observed number of distal radius fracture
seen in Leicestershire in each denrivation quintile from 2007 to 2010
Figure 18 Mean age of distal radius fracture nations in Leicestershire in each quintile
with 0.5% confidence intervals from 2007 to 2010
Figure 10. Belationship between dersal tilt and IMD rank of distal radius fracture
Figure 19 Relationship between dorsal tilt and IND rank of distal radius fracture
patients in Leicestersnire from 2007 to 2010 investigated using a generalised
additive model
Figure 20 Relationship between intra-articular step and IVID rank of distal radius
tracture patients in Leicestersnire from 2007 to 2010 investigated using a
generalised additive model
Figure 21 Relationship between MCI and IMD rank of distal radius fracture patients in
Leicestershire from 2007 to 2010 investigated using a generalised additive model

Figure 22 Mean MCI of distal radius fracture patients in Leicestershire from 2007 to
2010 in each deprivation quintile with 95% confidence intervals
Figure 23 Comparison of the expected and observed number of fractures seen in
Leicestershire in each deprivation quintile from 2007 to 2010 in patients aged 50
years and over123
Figure 24 Mean age of distal radius fracture patients in Leicestershire from 2007 to
2010 aged 50 years and over in each deprivation quintile with 95% confidence
intervals124
Figure 25 Relationship between dorsal tilt and IMD rank of distal radius fracture
patients in Leicestershire from 2007 to 2010 investigated using a generalised
additive model in patients aged 50 years and over125
Figure 26 Relationship between intra-articular step and IMD rank of distal radius
fracture patients in Leicestershire from 2007 to 2010 investigated using a
generalised additive model in patients aged 50 years and over126
Figure 27 Relationship between MCI and IMD rank of distal radius fracture patients in
Leicestershire from 2007 to 2010 investigated using a generalised additive model
aged 50 years and over127
Figure 28 Comparison of the expected and observed number of fractures seen in
Leicestershire in each deprivation quintile from 2007 to 2010 in adult patients
aged under 50 years129
Figure 29 Mean age of distal radius fracture patients in Leicestershire from 2007 to
2010 in adult patients aged under 50 years in each deprivation quintile with 95%
confidence intervals131
Figure 30 Relationship between dorsal tilt and IMD rank of distal radius fracture
patients in Leicestershire from 2007 to 2010 investigated using a generalised
additive model in adult patients aged below 50 years
Figure 31 Relationship between intra-articular step and IMD rank of distal radius
fracture patients in Leicestershire from 2007 to 2010 investigated using a
generalised additive model in adult patients aged below 50 years
Figure 32 Relationship between MCI and IMD rank of distal radius fracture patients in
Leicestershire from 2007 to 2010 investigated using a generalised additive model
in adult patients aged below 50 years
Figure 33 distal radius fracture incidence rate ratio of white patients in each
deprivation quintile compared to the most deprived quintile in Leicestershire
from 2007 to 2010
Figure 34 distal radius fracture incidence rate ratio of patients of other ethnicity in
each deprivation quintile compared to the most deprived quintile in
Leicestershire from 2007 to 2010
Figure 35 Generalised additive model analysis comparing dorsal tilt and IVID rank by
gender and ethnicity in distal radius fracture patients in Leicestersnire from 2007
to 2010 aged 50 years and over
Figure 36 Generalised additive model analysis comparing IVICI and IVID rank by gender
and EQ years and over
ageu ou years dilu uver
rigure 57 Generalised additive model analysis comparing dorsal till and IND rank by
2007 to 2010 agod under EQuears
2007 to 2010 aged under 50 years

Figure 38 Generalised additive model analysis comparing MCI and IMD rank by gender and ethnicity in adult distal radius fracture patients in Leicestershire from 2007 to 2010 aged under 50 years
Figure 39 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and age group in white distal radius fracture patients in Leicestershire from 2007 to 2010
Figure 40 Generalised additive model analysis comparing MCI and IMD rank by gender and age group in white distal radius fracture patients in Leicestershire from 2007 to 2010
Figure 41 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and age group in distal radius fracture patients of other ethnicity in Leicestershire from 2007 to 2010
Figure 42 Generalised additive model analysis comparing MCI and IMD rank by gender and age group in distal radius fracture patients of other ethnicity in Leicestershire from 2007 to 2010
Figure 43 Incidence rate ratio of white patients who underwent surgery for distal radius fracture in each deprivation quintile compared to the most deprived quintile in the UK from 2009 to 2013
Figure 44 Mean wait for semi elective trauma surgery in patients who required surgical treatment for an acute distal radius fracture dependent on the day they were listed for surgery with 95% confidence intervals
Figure 45 Acute distal radius fracture fixation rate (expressed per 100,000 person/years) by hospital trust in England from 2009 to 2013162Figure 46 Plot of acute distal radius fracture fixation rate (expressed per 100,000
person/years) by hospital trust in England population with 95% prediction intervals163
Figure 47 Flow chart of study identification, screening and selection
Figure 49 Meta analysis of studies reporting results for risk of hip fracture after distal radius for men only
Figure 50 Meta analysis of studies reporting results for risk of hip fracture after distal radius for men and women combined
Figure 51 Example radiograph of an osteopenic patient (MCI of 0.50, equivalent to a T- score of -1)
Figure 52 Example radiograph of an osteoporotic patient (MCI of 0.44, equivalent to a T-score of -2.5)
Figure 53 Example radiograph of a patient with normal BMD (MCI of 0.53)187

Abbreviations

AP	Antero-posterior
BMD	Bone Mineral Density
CI	Confidence Interval
СТ	Computed Tomography
DASH	Disabilities of the Arm, Shoulder and Hand
DRAFFT	Distal Radius Acute Fracture Fixation Trial
distal radius fr	acture Distal Radius Fracture
DXA	Dual-energy X-ray Absorptiometry
FRAX	Fracture Risk Assessment tool
GAM	Generalised Additive Model
GLM	Generalised Linear Model
GLMM	Generalised Linear Mixed Model
HR	Hazard Ratio
ICC	Intraclass Correlation Coefficient
ICD	Inner Cortical Diameter
ICER	Incremental Cost Effectiveness Ratio
IMD	Index of Multiple Deprivation
IRR	Incidence Rate Ratio
LSOA	Lower Super Output Area
MCI	Metacarpal Cortical Index
NICE	National Institute of Clinical Excellence
NIHR	National Institute for Health Research
OCD	Outer Cortical Diameter
ONS	Office of National Statistics
OPCS	Office of Population Censuses and Surveys
ORIF	Open Reduction Internal Fixation
РА	Postero-anterior
PROM	Patient Reported Outcome Measure

- PRWE Patient Reported Wrist Evaluation
- QALY Quality Adjusted Life Year
- RR Relative Risk
- UHL University Hospitals of Leicester
- VLP Volar Locking Plate
- WHO World Health Organisation

Relevant publications

Risk of hip fracture following a wrist fracture-A meta-analysis. Johnson NA, Stirling ER, Divall P, Thompson JR, Ullah AS, Dias JJ. Injury. 2017 Feb;48(2):399-405.

Comparison of distal radius fracture intra-articular step reduction with volar locking plates and K wires: a retrospective review of quality and maintenance of fracture reduction. Johnson NA, Dias JJ, Wildin CJ, Cutler L, Bhowal B, Ullah AS.

J Hand Surg Eur Vol. 2017 Feb;42(2):144-150.

Are radiographic measurements of the displacement of a distal radial fracture reliable and reproducible?

Stirling ER, Jeffery J, Johnson NA, Dias J.

Bone Joint J. 2016 Aug;98-B(8):1069-73.

Conference presentations

Variation in surgical fixation rate of wrist fractures in England.Johnson NA, Stirling ER, Thompson JR, Dias JJ.58th Congress of the German Society of Hand Surgery, Munich, September 17.

Epidemiology of distal radius fractures in adults in a geographically defined population. Johnson NA, Stirling ER, Thompson JR, Dias JJ.

British Society for Surgery of the Hand, Cardiff, October 16.

Chapter 1 Introduction

1.1 The history of deprivation and health

The adverse association between social deprivation and numerous health outcomes has been recognised for many years. In 1843 a paper was published in the Lancet describing a strong relationship between life expectancy and occupation(1). The study recommended that employers and landlords should "make alterations for the comfort and health of those comparatively helpless classes of the community, the artisans and labourers."

Socioeconomic deprivation is a difficult concept which is not just related to financial poverty. In 1987 Peter Townsend wrote an article suggesting 'people can be said to be deprived if they lack the types of diet, clothing, housing, household facilities and fuel and environmental, educational, working and social conditions, activities and facilities which are customary(2).' Therefore, deprivation can relate to a lack of different types of resources, not just money. Whereas those in poverty lack financial resources to prevent deprivation.

In the United Kingdom there is a long history of research into deprivation and social inequality. Charles Booth used both quantitative and qualitative methods to investigate poverty in London in the late 19th century(3). His report titled 'Life and Labour of the People' was published in 1889 and is seen as one of the founding texts of British sociology. He noted that 35% of people in East End London were living in abject poverty and argued for the introduction of old age pensions and free school meals.

Seebohm Rowntree was a sociological researcher who carried out comparable work in York during a similar time period. His first study involved researchers visiting and surveying resident's homes and calculating a minimum weekly amount of money required for families to live a healthy life(4). This included funds for rent, fuel, light,

food, clothing, household and personal items. He applied new scientific methods such as analysing the cost of food necessary for minimum calorie requirements.

In the twentieth century Peter Townsend's work pioneered the relative deprivation approach to poverty. This takes into account a wide variety of factors including material and social living standards. He stated that for deprived people: "their resources are so seriously below those commanded by the average individual or family that they are, in effect, excluded from ordinary patterns, customs and activities."

This approach looks at factors which indicate deprivation. These identified factors can then be used to look at other variables such as income. Deprivation occurs because of lack of income and other resources. Deprivation measures are therefore related to how individuals live and not just their income. In 1968 to 1969 Townsend performed a large UK survey using this relative deprivation methodology and reported these results in 1979(5, 6). He selected 60 indicators of standard of living covering the following broad categories:

- diet
- clothing
- fuel and light
- home amenities
- housing and housing facilities
- the immediate environment of the home
- the general conditions and security of work
- family support
- recreation
- education
- health
- social relations

Absence of indicators suggested deprivation and deprivation scores were produced and compared with household income. This work identified levels of income at which standards of living would be seen as less than acceptable for modern society. Some critics of Townsend's work suggested that measurement of his chosen indicators would not allow for the difference in how individuals choose to live. The consensual method was later developed to address these concerns(7).

This involves looking at direct measures of standards of living with deprivation defined as an enforced lack of key "necessities". These necessities are identified by public opinion and not from the views of experts. This method also contains an option for those who do not possess the necessities due to choice rather than lack of resources. The initial step for this approach therefore involves a large amount of information gathering regarding what the public believe to be the necessities which everybody should be able to afford and not manage without. These items can include food, health, housing, relationships, possessions, hobbies and financial security.

In the next stage for this consensual method surveys are undertaken to establish who lacks these necessities and if this is due to choice. Increased number of 'enforced lack of necessities' is regarded as more deprivation. Analysis is then performed to identify a 'poverty threshold' which in most studies is related to the lack of two or three necessities. When comparing deprivation through time the lack of three necessities is used as the poverty threshold. This amount of deprivation is associated with poor physical and mental health, financial difficulties and individuals who are most likely to identify themselves as being poor. Over the past thirty years there has been a rise in this deprivation poverty from 14% in 1983 to 33% in 2012(8).

1.2 Measures of deprivation

1.2.1 Index of Multiple Deprivation

The Index of Multiple Deprivation(IMD) is the official UK government measure of deprivation. This was first used in 2000 and has been updated approximately every three years since. Multiple deprivation is measured at the small area level using seven separate recognisable and measurable dimensions of deprivation(9). Individuals living

within the area experience these dimensions. This builds on Townsend's concept of relative deprivation.

The seven domains included in the IMD are:

- Income Deprivation
- Employment Deprivation
- Health Deprivation and Disability
- Education, Skills and Training Deprivation
- Barriers to Housing and Services
- Crime
- Living Environment Deprivation.

Multiple deprivation is the cumulative effect of these domains.

A total of 38 different indicators are used to capture the deprivation data for the domains. To be included an indicator must be specific to that domain and as direct as possible a measure of that type of deprivation. It must measure the main components of that deprivation, be up-to-date, easily updated, statistically robust and available across England at a small area level.

As the IMD measures deprivation at a small area level some indicators have large standard errors due to small numbers. In these cases shrinkage methodology is used to shift scores towards local authority averages for that indicator. Following this the indicator scores are combined to form the domain scores. Units used depends on the domain characteristics. In the income and employment domains the results are presented as the proportion of households experiencing that type of deprivation by comparing households to a defined level of income or employment rate. This is not possible in the other domains were simple rates cannot be calculated. For these domains indicator measurements are ranked then transformed to a normal distribution and weightings applied to form a domain score. The seven individual domain scores must then be combined to create the overall IMD score. As the units of measurements vary the individual scores are standardised by ranking. These ranked scores are then exponentially transformed to allow continued identification of the most deprived areas. Before the domain scores for each spatial area are added together weighting is applied as some domains are understood to contribute to deprivation more than others. Income and employment are believed to be the most important and carry a domain weight of 22.5% each.

Small spatial areas called Lower Super Output Areas (LSOAs) are used to measure and compare deprivation. LSOAs are small areas of relatively equal size, usually equating to a few postcodes and consisting of about 1,500 people. Leicestershire and Rutland contains 606 LSOAs. These spatial measures have been used since IMD2004. Previously electoral ward based geographic areas had been used. These measures have weaknesses as electoral ward boundaries are frequently altered in relation to population size change to maintain a similar rate of elected local councillors to population. Therefore, changes in deprivation over time are difficult to accurately assess.

Ten indices are provided for each LSOA.

- Total Index of Multiple Deprivation score
- The seven individual domain indices
- Supplementary Income Deprivation Affecting Children Index
- Supplementary Income Deprivation Affecting Older People Index.

A higher total IMD score reflects greater deprivation. LSOAs are ranked on each indice score from 1 to 32,482 with rank 1 being the most deprived area and highest score. These rankings are split into 5 equal quintiles.

Complicated patterns of deprivation are seen across England. All nine regions contain some of the most and least deprived LSOAs. The East Midlands is comprised of 2732 LSOAs. 53.6% of these LSOAs are within the least 50% deprived LSOAs, 7.4% are in the 10% most deprived. These most deprived areas are focused around the cities of Leicester, Nottingham and Derby.

1.2.2 Other measures of deprivation

1.2.2.1 Jarman score

The Jarman score is also known as the underprivileged area score. It was originally devised in the mid eighties from a large survey of General Practitioners about the importance of variables affecting their workload. This information is used to give weightings to eight census variables which are added together to give the score for a geographical area. The initial purpose was to then give additional payments to GPs treating deprived patients.

The eight variables included are the proportion of:

- elderly living alone
- children aged under 5
- unskilled workers
- overcrowded households
- change of address in last year
- people living in a household where the head of household was born in a New Commonwealth country or Pakistan
- unemployed
- Single parents

Mean score is 0. A higher score suggests deprivation and a higher than expected demand for primary care. An area with a score of 30 or more is classified as deprived.

1.2.2.2 Townsend Material Deprivation Score

This was developed by Peter Townsend in 1988. It is based on four census variables:

• households without a car

- overcrowded households
- households not owner-occupied
- persons unemployed (of working age)

Scores are standardised then added together. Mean score is 0 and an increasing score indicates more deprivation. They can be calculated for LSOAs or larger areas.

1.2.2.3 Carstairs score

The Carstairs score were first developed in the eighties as an alternative to the Townsend score for Scotland. It can be calculated at ward level and is based on four census variables:

- unemployed males
- households without a car
- overcrowded households
- Low social class

Similar to the Townsend method scores are standardised then added together. Mean score is 0 and an increasing score indicates more deprivation.

1.2.3 Comparison of deprivation measures

All the measures above have the strength of quantifying deprivation at relatively small geographical levels. The Jarman score includes a diverse range of measures. Data used is linked to the 2001 census so may be out of date. It has been criticised for including the elderly, single parents and some ethnic groups in its definition of deprivation and implying they are a cause of the problem. The Townsend and Carstairs score are similar to each other. They are easy to calculate and shown to correlate with health measures better than the Jarman score. Disadvantages are the small number of indicators used and they also use census data which could be outdated. The Index of

Multiple Deprivation has now largely surpassed these measures. It uses more sophisticated statistical methods and a broad range of indicators. LSOAs are used to measure deprivation at smaller geographical levels than the other scores. Almost all recent research uses the IMD and for these reasons it was chosen to be used throughout this thesis.

1.3 Social deprivation and health

Social deprivation is associated with many major health problems including cancer, cardiorespiratory disease and diabetes but the potential reasons for this remain varied and largely unclear.

Lower socioeconomic status is associated with lower cancer survival rates and with presentation at a more advanced stage in the UK(10, 11). Lung cancer treatment rates are lower in those of lower socioeconomic status and they are more likely to present initially as an emergency and to die soon after diagnosis(12-15). These findings would suggest strategies to improve earlier diagnosis in deprived patients are required and would be beneficial.

In a study of 1417 patients presenting with an acute myocardial infarction in London it was found that there was minimal variation between deprivation groups in age, sex, smoking status or type of treatment(16). More patients from deprived areas were diabetic or south Asian and these patients were more likely to suffer more significant disease. Deprived patients did have a higher risk of early recurrent ischaemic events.

Kaplan et al carried out a population based study of 2679 men in Finland and found an association between socioeconomic state in childhood and ischaemic heart disease in later life(17). Other cardiac risk factors, measured as an adult, did not explain this difference. Further work by the same group attempted to find reasons for this(18). They reported that adult behaviours detrimental to health are related to low education, 'blue collar' employment and poor childhood conditions. They

recommended that governments must understand that economic policy is directly linked to public health policy.

Asthma rates in the US vary with race, being higher in Blacks and Hispanics(19). Odds ratios of 2.1 and 2.9 are seen in adults and children respectively compared to Whites. However, when three measures of socioeconomic status; family income, level of education and area of residence, were taken into account a large proportion of the differences between ethnicity decreased. A large study in Scotland using Carstairs score to measure deprivation showed a relationship between deprivation and type 2 diabetes and obesity(20). There was no relationship with type 1 diabetes. Several studies have shown an association between acute pancreatitis incidence and mortality with deprivation(21). This is most pronounced for acute pancreatitis caused by alcohol excess.

1.3.1 Social deprivation and injury

Many authors have attempted to study the relationship between social deprivation and injury, particularly for children. Dowswell et al carried out a systematic review looking at social deprivation and the prevention of injury in childhood(22). They found 32 studies which investigated this. However, the literature covered many different types of injury and social deprivation was defined in considerably different ways. They concluded therefore that it was currently difficult to design and target interventions to address inequalities in paediatric injury rates.

Dunn et al noted differences in the mechanism and patterns of injury in adult patients presenting with a head injury in Scotland(23). Deprived patients were more likely to suffer an injury due to an assault compared to less deprived patients who were more likely to sustain an injury through a road traffic accident or fall from height. Isolated injuries, normal physiology and Glasgow coma Score were more often seen in deprived patients. It was felt that these findings were due to the mechanism of injury. There was no difference in time to treatment, rate of transfer to a specialist centre or mortality. Another Scottish study reported that victims of assault attending

emergency departments were more likely to have been drinking alcohol, have suffered previous assault, be unemployed and from an area with higher deprivation score(24).

A study of over 100,000 patients with head injury in England again showed incidence rate was associated with socioeconomic factors such as unemployment, sickness from work and lone parent families(25). Interestingly access to good public transport for travel to work was linked to significantly reduced head injury rate. Therefore, some areas of London, with very high deprivation, had much lower head injury rates than expected due to the excellent public transport network.

Severe limb injury is also more common in deprived patients. A study of 601 patents admitted to Level 1 trauma centres in the US with high energy lower extremity injuries found they were more likely to live in households below the poverty line and be less likely to have health insurance compared to the general population(26). Heavy drinking was twice as high compared to national rates. Treatment decision in terms of amputation or limb salvage were not related to patient characteristics.

1.3.2 Deprivation and trauma

Several studies have investigated the relationship between fractures and deprivation. Curtis et al analysed data from UK General Practitioners covering 11.3 million people between 1988 and 2012(27). They compared fracture incidence with IMD quintiles. When comparing regions within the UK deprivation was found to be associated with increased fracture risk after adjustment for sex and age. Rates in Scotland were almost 50% higher than those in the less deprived South-East of England.

In men, the relative risk of vertebral, wrist and hip fractures increased with deprivation. The largest association was seen in men sustaining a hip fracture with those in the most deprived quintile having a relative risk of 1.3 (95% Cl 1.21 - 1.41) compared to those in the least deprived quintile. These findings were not observed in women. No increase in risk of wrist or hip fractures and a reduced vertebral fracture

risk of 0.86 (95% CI 0.79 – 0.94) was seen in the most deprived quintile compared to the least deprived for women.

They suggested that higher fracture incidence for the more deprived was due to lifestyle factors such as smoking, alcohol and poor diet which are seen more frequently in areas of lower socioeconomic status and known to have a negative effect on bone health(28-31). It was felt that these negative lifestyle factors are seen more often in men than women which may explain the difference between genders. A study from Australia also noted a noted a larger association between deprivation and fracture in men(32).

Another large study in Wales calculated fracture rates by Townsend deprivation quintiles for 60,166 people who attended an A&E department with a fracture(33). They found that fracture incidence was significantly higher in areas with higher deprivation scores. However, this effect diminished with age and they concluded that socioeconomic deprivation does not appear to be a risk factor for development of osteoporotic fractures. Their study was not intended to investigate the reason for this but they felt that adverse lifestyle factors which are likely to be seen in more deprived areas such as smoking, alcohol, poor diet and reduced HRT uptake may be offset by increased levels of physical activity in more deprived areas were people are less likely to have cars. It may be that these deprivation linked lifestyle factors are less important than falls, osteoporosis and other medical co-morbidities. Fractures occur more frequently in older people living in residential and nursing care(34). These care homes are generally in large converted houses in less deprived areas meaning there is a "socioeconomic drift" of deprived older people to care in more affluent areas. Jones et al analysed this effect in their study and found the number of cases was tiny and would not influence any relationship between deprivation and fracture(33).

Aitken et al reviewed 6,872 consecutive adult patients with fractures attending an Emergency Department in Edinburgh over a one year period. Carstairs score was used to investigate an association with deprivation deciles(35). An association with deprivation was seen in 22 of the 27 fracture types studied in men and 17 out of 27 in women. In this study the effect of deprivation was not linear and the most deprived 10% of the population had the highest fracture risk. When comparing the most

deprived deciles (9, 10) with the less deprived deciles (1-8) by calculating odds ratios, to investigate the effect of severe deprivation, only distal femur fractures in males were not associated with deprivation. In women, there were five fractures that were not associated with severe deprivation – calcaneal, talar, distal humerus, distal tibial and diaphyseal tibial fractures. Odds ratios were similar between genders for the majority of fractures. Five fractures (scapula, proximal ulna, midfoot, distal humerus, proximal tibia) in men had an odds ratio of over 7.5. The odds ratios for women with these fractures were much lower. Men in decile 9 or 10 who sustained any of those five fractures were compared to the comparable women patients. 50% of men admitted to drinking more than 28 units per week, 28% were smokers and 24% of the injuries were due to road traffic accidents. The findings for women were 0.6%, 12% and 5% respectively. This again suggests lifestyle factors may influence fracture rate in deprived patients and help to explain gender differences.

The percentage of the common osteoporotic fractures (proximal femur, proximal humerus, wrist) seen in the most deprived decile was compared with the least deprived decile. In the most deprived decile 17.7% of patients had a wrist fracture, 7.4% a proximal humerus fracture and 8.6 % a proximal femur fracture compared to 18.8%, 6.6% and 15.3% in the least deprived decile respectively. The only significant difference was the reduced incidence of proximal femur fracture in the most deprived group. Life expectancy is known to be lower in this decile and the authors felt this difference was due to the fact that more deprived patients died earlier and did not live long enough to suffer an osteoporotic proximal femur fracture.

The same authors also studied the relationship between adult tibial diaphyseal fractures and deprivation in Edinburgh(36). These injuries have a bimodal distribution and are generally due to high energy trauma in younger patients and secondary to minor trauma and osteoporosis in older patients. They demonstrated a clear relationship between tibial fractures and deprivation but the effect lessened with age. Tibial fractures in younger patients tended to occur in men from socially deprived areas after football injuries, direct blows or assaults and road traffic accidents. Overall those injured playing sport were less deprived. Football was associated with deprivation but other activities such as horse riding and skiing were associated with

affluence. These younger patients are unlikely to have reduced bone density due to alcohol, smoking or poor diet. In the older group, most injuries occurred after low energy injuries and simple falls. It was felt that the mechanism of injury was related to deprivation with falls affecting older people from all deprivation groups. This agrees with the study by Jones et al. (33).

Quah et al prospectively analysed incidence of hip fracture and deprivation in 7,511 patients using the IMD 2007(37). The study was carried out in Nottingham which has a similar population to Leicester but with slightly higher levels of deprivation. Hip fracture incidence was 1.3 times higher in the most deprived quintile compared to the least deprived. More co-morbidities were seen in the more deprived population agreeing with other findings that deprivation is associated with a less healthy lifestyle potentially leading to the development of health problems. No difference in mortality was seen at 30 days but those in the most deprived quintile had a higher mortality rate at one, three, five and seven years when compared to the least deprived quintile. These findings could be expected to occur regardless of hip fracture. The authors acknowledged the variation in the literature regarding the influence of deprivation on hip fracture. They thought the variety of measures of deprivation used in other studies could be an explanation for this.

Brennan et al performed a systematic review to attempt to address the question of whether socioeconomic status is an influence on osteoporotic fracture(38). They evaluated the evidence and found only 11 suitable studies suggesting a lack of good quality literature. There was strong evidence of an association between being married or living with someone and a reduced risk of osteoporotic fracture. Limited evidence was found for a relationship between occupation type or employment status and fracture, or for type of residence and fracture. There was conflicting evidence for the relationship between osteoporotic fracture and level of income and education. They suggested further high quality literature on the subject was required.

1.3.3 Adolescent fractures and deprivation

A further study from Edinburgh investigated the relationship between adolescent fractures and deprivation(39). They believed this was an important group to study who are often included in adult or paediatric studies despite being a separate group with specific characteristics. The population was subdivided into junior (10 - 14 years) and senior (15-19 years) adolescents. Weighted linear regression using Carstairs index as a factor demonstrated social deprivation was an independent and significant risk factor for hand fractures in senior adolescent males, upper limb fractures in junior adolescent males and upper limb and distal radius fractures in senior adolescent females. All deprivation associated injuries occurred in the upper limb. The authors hypothesised that the mode of injury is the main causative factor in the relationship between fracture and deprivation.

1.3.4 Paediatric fractures and deprivation

There is also evidence linking paediatric fractures with increased deprivation. Stark et al reviewed 2712 children's fractures in Glasgow from a population at risk of 167,972 in 1997(40). They used cluster analysis of demographic variables from census data to split postcodes into three groups; affluent, middle and deprived. Combined fracture rate was significantly higher for children from deprived areas. Prior to this Lyons et al used Townsend scores to investigate fracture rate and deprivation in 2399 children with fractures in South Wales(41). Fractures rates were similar across deprivation groups. More deprived children were more likely to sustain fractures due to assault whereas sports related fractures were more common in affluent areas. The authors commented that they thought the finding of no association with deprivation was surprising and still suggested that injury prevention strategies should be targeted towards deprived areas.

Bridgman et al studied paediatric femoral fractures in the West Midlands(42). Over a 10 year period 3272 children suffered a femoral fracture. When analysed by four year age bands and gender a significantly higher rate of fracture was seen with increasing deprivation for all age-gender groups. Fractures due to road traffic accidents and falls

were 3 times and 1.5 times more frequent respectively in the most deprived areas compared to the most affluent. Similar findings have also been reported for paediatric femoral shaft fractures in Maryland, USA(43). Fracture was associated with low household income, crowding of housing, single mother families and living below the property level.

Another study from the Edinburgh unit used the Scottish Index of Multiple Deprivation to assess for a relationship with deprivation in the 2,195 children who sustained fractures in the year 2000(44). Overall there was a correlation between increasing deprivation and rate of all fractures combined. Analysis by individual fracture types revealed an association with fractures of the metacarpal, distal humerus, clavicle, midshaft of the forearm and metatarsals when compared with more affluent children. Injuries due to falls, road traffic accidents and blow or crush injuries were more common with deprivation. Sports related fractures were more likely in the least deprived children. They felt this may be due to fewer safe areas to play, higher exposure to traffic and less adult supervision in deprived areas(45-48).

1.3.5 Distal radius fractures and deprivation

To my knowledge only one paper has specifically investigated the relationship between distal radius fracture and deprivation. In a study from Edinburgh, Clement et al sought to identify if socioeconomic status influenced the epidemiology and outcome of adult distal radius fracture(49). They identified 3,983 patients with a distal radius fracture over a seven year period and evaluated socioeconomic status with the Carstairs score, dividing the cohort into quintiles according to this. They assessed all initial radiographs and reviewed a further 770 radiographs at one year following injury. Functional assessment at one year was carried out in 644 patients. This involved measurement of range of movement, grip strength and subjective assessment by asking a standardised set of questions such as had the patient regained normal use of the hand. They found that those in the most deprived quintile were significantly younger and sustained the injury 3.1 years earlier than the most affluent

patients. They were also more likely to be male and sustain the injury by a high energy mechanism. There was no difference in functional assessment or radiographic position of the fracture between the quintiles. Interestingly those in the most affluent quintile were more likely to suffer with complex regional pain syndrome.

The authors commented that patients may sustain a fracture earlier due to reduced bone quality but also note that their radiographic findings do not support this as there was no difference in severity of fracture pattern, displacement or comminution between the groups. They conclude that it is not clear whether the age difference is related to diminished bone quality.

The study assessed the whole fracture group together by quintile and compares mean age, gender, mechanism of injury, treatment and radiographic findings. They therefore do not attempt to identify differences between the two peaks of patients seen in the bimodal distribution for distal radius fracture representing fragility fractures in older patients and higher energy injuries in younger patients. Rate of injury is not calculated and they do not make any comparison with the numbers and characteristics of fracture patients and those in the background population without a fracture.

Four other papers have reported some results regarding the association between distal radius fracture and deprivation. In their study from Edinburgh looking at a wide variety of fractures Aitken et al found distal radius fracture was associated with deprivation measured by Carstairs score, for men and women(35). Odds ratio of sustaining a fracture in the two most deprived deciles was 4.2 for men and 3.0 for women compared to those in the other eight deciles. However in a study by the same group looking at only falls related fractures there was no difference in distal radius fracture prevalence across deciles(50). Curtis et al found an association for men but not women using the IMD(27). Ong found no difference between number of patients with distal radius fracture and IMD 2010 quintile in those being treated in a fracture clinic in Nottingham(51).

1.3.6 Hand and upper limb fractures and deprivation

In their study from the Derby hand unit Horton et al reviewed data on 1,234 patients presenting over a 6 month period(52). They used the Index of Multiple Deprivation 2004 and found a significant association with hand trauma and deprivation. Odds ratio for sustaining a hand injury was 1.6 in the most deprived quintile compared to the least deprived. They found that the association was greatest in adults and older children but there was not a constant relationship between deprivation and incidence of hand trauma in older patients or young children. They felt this may be a type two error due to low incidence of trauma in the very young and old.

Sporting injuries were not associated with deprivation and were most common in the least and most deprived quintiles. The authors felt this may be due to the affluent and the unemployed having more time for these activities. Injuries in the more deprived patients required lengthier surgical time, suggesting greater complexity, and self reported physical outcome was worse.

Contrastingly Anakwe et al did not demonstrate an association between hand fractures and deprivation(53). They studied 1,569 fractures in 1,382 patients in Edinburgh and used the Carstairs score to measure socioeconomic status. Only fifth metacarpal fractures in men were associated with deprivation. They felt this was due to mechanism of injury as they were frequently sustained due to assaults, blunt trauma and crush injuries. In women, only the mechanism of injury was related to deprivation, with a significant association seen with an unclear mechanism or the patient being intoxicated when injured. Again, there was no association with deprivation and sporting injury. Rate of surgical intervention was higher in affluent patients suggesting socioeconomic status influences treatment.

Duckworth et al created a multivariate linear regression model to assess the influence of several variables on patient reported functional outcome after proximal radius fracture in 200 patients(54). Socioeconomic status, when assessed by the Scottish Index of Multiple Deprivation (IMD 2009), was a significant factor negatively influencing outcome. The other predictors were a compensation claim and severity of

fracture classification. The model took into account age, gender, mechanism of injury and co-morbidities. When reviewed separately mechanism of injury was not associated with deprivation in this group of patients.

1.4 Deprivation and osteoporosis

As mentioned previously authors of studies investigating fracture rate and deprivation have hypothesised that risk factors for poor bone health such as smoking, poor diet and alcohol would be seen more often in deprived populations. However, the evidence linking social deprivation and osteoporosis is varied. Most studies investigate the relationship between deprivation and fractures and draw conclusions from this rather than directly looking for an association between deprivation and measures of osteoporosis or bone health. In their large study Jones et al found that the risk of osteoporotic fractures in older patients was not related to socioeconomic status(33). A study of 6,160 Italian women who underwent bone mineral density assessment by DXA scan found that osteoporosis was prevalent in 28.7% of the least educated patients compared to 18.3% of the most educated(55). Four times as many women from the most educated group attended the clinic for assessment and more of these were smokers, making comparison with a UK population difficult.

A recent study from Nottingham used the IMD 2010 to assess deprivation and osteoporosis in patients that attended fracture clinic and underwent a DXA scan(51). Similar number and proportion of fracture type were seen across all quintiles. There was no difference in prevalence of osteoporosis between quintiles in those that attended for a DXA scan. There was a lower attendance for a scan in those referred from more deprived quintiles. When interpreting these results it is worth noting that only patients who could be managed as an outpatient are included. Patients were not referred for DXA scan if they did not sustain a low energy fracture, lived in a care home or were thought to be too frail, unwell or cognitively impaired to tolerate a scan, were already being treated for osteoporosis or had a DXA scan in the last three years.

Pearson et al measured heel bone mineral density of 1,187 women in Nottingham(56). Jarman score was used to measure social deprivation. Women in the two most deprived quartiles had a significantly higher prevalence of osteoporosis and lower bone mineral density. The authors felt Jarman score was likely to act as a composite surrogate for other risk factors of osteoporosis. Limitations of the study were a poor uptake of screening from those in deprived areas which led to a mean Jarman score of study participants being much higher than the average for the region. 6% of people in the health authority and 11% in the city are non white but uptake from non white participants was only 0.3%.

A study from Boston, US followed 692 men over 7 years and measured bone mineral density annually. Multivariate models were created to identify which factors influenced change in bone mineral density. Income was used as a measure of socioeconomic status. Lower income was strongly associated with higher loss of bone mineral density measured at the hip, independent of a large number of potential confounders included in the model. Decline in bone mass began relatively early and the rate did not vary with ethnicity or genetic ancestry. They concluded that early life exposures influence peak bone mass which then steadily declines, independent of ethnicity, but income related factors affect the rate of loss(57).

1.5 Deprivation and falls

A further reason for increased fracture rates in deprived patients could be a higher risk of falls. It could be expected that those from deprived communities may have living conditions with hazards like poorer lighting, cramped surroundings and unsafe stairs along with suffering from more medical problems such as visual disturbances, poor balance and postural hypotension, all of which are known falls risk factors. Evidence investigating this is scarce.

In the study by Pearson et al there was no relationship between a history of two or more falls in the past year and deprivation (56). Visual problems were significantly
higher in the three most deprived quartiles. West et al demonstrated a small but significant link between deprivation and falls requiring hospitalisation(58). Admission rate following a fall was 10% higher in the two most deprived quartiles measured by Townsend score. There was no relationship between deprivation and hip fracture. They hypothesised that deprived patients who suffer a minor fall may be more likely to require hospital admission due to medical co-morbidities or lack of social support but that falls severe enough to cause a hip fracture occur independently of deprivation.

In a further study by Court-Brown et al the association between deprivation and 3,843 falls from standing height was investigated(50). This showed an overall association between fall related fractures and deprivation but hip fracture prevalence declined with increasing deprivation. Age at fracture also reduced with increasing deprivation. The authors believed that these findings were due to reduced life expectancy in deprived patients. Gribbin et al investigated falls which occurred in 61,248 patients seen in primary care(59). Incidence of falls and recurrent falls was significantly associated with increased social deprivation measured by Townsend quintile.

1.6 Deprivation and access to health care

Understanding access to healthcare and the relationship with need and demand is complex. Need reflects the requirements for healthcare of a community. Demand is how these services are utilised. Inequality of access results from a discrepancy between need, demand and the supply of these services. Several studies have reported an association between higher levels of primary consultation and deprivation(60, 61). Higher use of inpatient services has been seen in unskilled manual workers and those without access to a car(62, 63). There is evidence that health promotion and preventative services are used less in deprived areas with manual workers 10% less likely to attend their GP for preventative interventions than nonmanual workers(64, 65).

It seems likely that the vast majority of patients with an acute distal radius fracture will attend their nearest Emergency Department. A large study of General Practices in England analysed data from 4.5 million patients who had self referred to an Emergency Department and been directly discharged without hospital follow up(66). A negative binomial regression model identified that increasing deprivation quintile was associated with increased attendance at an Emergency Department. It therefore seems unlikely that deprived patients with a distal radius fracture would be less likely to present for emergency treatment.

1.7 Deprivation and health summary

From reviewing the literature investigating the relationship between social deprivation and fractures some trends have emerged. For young patients, who sustain fractures not related to osteoporosis, there appears to be good evidence that increased risk of fracture in those from socially deprived backgrounds is likely to be due to the mechanism of injury. Deprivation seems to have a larger effect on fracture risk of men of all ages. This is thought to be due to lifestyle factors which have an adverse effect on bone health and also increase risk of injury.

The overall evidence investigating the relationship between social deprivation and fragility fractures such as those of the hip is varied. It could be expected that many risk factors for osteoporosis would be more likely to be seen in more deprived areas but in some studies this is not the case. Other factors such as increased activity may offset this. There is some evidence reduced bone mineral density is associated with lower socioeconomic status but several studies do not demonstrate this translates to an increase in osteoporotic fractures. Minimal evidence is available to investigate the relationship between falls and deprivation. There is good evidence that those from deprived areas will attend the Emergency Department when felt necessary so access to acute healthcare treatment seems to be unlikely to affect fracture rate however uptake of preventative strategies is low.

It should be noted that a large proportion of studies regarding fractures and deprivation come from the same trauma unit in Edinburgh which has a well established database for collecting information on all patients who attend. This population is mostly white with a much lower proportion of other ethnicities than Leicestershire. There is also considerable variation in the measures of deprivation used with few studies using the index of multiple deprivation.

1.8 Osteoporosis

1.8.1 Definition of osteoporosis

Osteoporosis is a chronic progressive and metabolic disease. It is the most common bone disease affecting men and women(67). It is most frequently seen in older white women but effects both sexes, all ethnicities and different age groups. Establishing any relationship between osteoporosis and deprivation is one of the key components of this thesis.

Osteoporosis will often present with fragility fractures. These are fractures which occur due to forces which would not normally cause a fracture, such as falls from standing height or less. Osteoporosis is a growing health problem and a huge economic burden across the world. It often leads to ongoing pain, disability and reduced quality of life(68).

It is characterised by reduced bone mass and micro architectural loss of bone tissue(69). This causes bone fragility. There are several different causes for osteoporosis and the pathogenetic mechanisms are complex. Throughout a person's life bone is continually remodelling. Bone is resorbed and this is then followed by bone formation. Osteoclasts resorb bone and control bone remodelling, while osteoblasts form new bone. Osteoblasts also mineralise osteoid and play a role in controlling osteoclast resorption. Resorption by osteoclasts occurs quicker than production of new bone by osteoblasts. Therefore increased bone remodelling leads

to a loss of bone(70). Following increased remodelling bone is initially less densely mineralised so fracture risk is higher(71).

Osteoporosis occurs due to the imbalance between resorption and formation of bone and leads to a reduction in bone mass. This may be due to an increase in resorption or reduction in formation. Oestrogen deficiency is an important factor in the development of osteoporosis. It is important in accelerated bone loss seen after menopause in women but also influences bone loss in men. Oestrogen receptors are found on osteoblasts, osteocytes and osteoclasts. Oestrogen also affects bone through growth factors and cytokines. With increasing age, there is a progressive reduction in osteoblast proportion and bone resorption exceeds formation. Vitamin D and calcium are required for normal bone homoeostasis. Insufficient dietary intake or other problems causing reduced intake or absorption can contribute to the development of osteoporosis. Other lifestyle factors such as inactivity, smoking and excessive alcohol consumption also increase the risk of osteoporosis. These factors have been shown to be associated with deprivation and have been the reason why many researchers have hypothesised that increased fracture rate in deprived areas is related to increased incidence of osteoporosis.

In the third decade of life bone mass is at its highest. If a high peak of bone mass is not obtained by this age then osteoporosis may develop later. Diet and exercise while young are therefore important for future bone health. However, genetics are thought to be the main determinant of peak bone strength. It has been suggested they are responsible for at least 80% of bone mass variance between different people. A large Norwegian study of 7,600 participants reported that forearm bone mineral density is relatively stable up until the age of 50 years(72). After 50 years, bone mineral density in males steadily decreased. In females, there was a decline from the age of 50, with a subsequent further decline after the age of 65. Many studies therefore use the age of 50 as the distinction between fragility fractures and non-osteoporotic fractures.

1.8.2 Diagnosis and measurement of osteoporosis

Bone mineral density is the amount of bone mass per unit volume or area. It can be measured by densitometric techniques. The commonest method uses dual energy x-ray absorptiometry (DXA). This utilises the absorption of x-rays which are sensitive to calcium content of tissue, which is largely bone. Commonly measurements are taken at the lumbar spine and proximal femur.

Bone mineral density is then used for the diagnostic criteria of osteoporosis. This was defined by the World Health Organisation. A Z score compares a patient's bone density with someone of the same age and sex. T score compares their bone density with somebody at their peak BMD. A T score for a woman of 2.5 standard deviations or more below that for a young female adult is classified as osteoporosis(73). The same threshold is used for men. Osteopenia, low bone mass, is defined as a BMD 1 to 2.5 standard deviations below the mean. Fracture risk increases 1.5 to 3 times with each 1 standard deviation reduction of BMD.

1.8.3 Fragility fracture

The classical sites for fragility fracture are the hip, vertebrae and forearm although they may occur almost anywhere in someone with low bone mineral density. Fracture risk exponentially rises with age. After the age of 50 years one in two women and one in five men will sustain a fracture(74). 70% of all fractures are seen in women aged over 65 years(75). Around 300,000 fragility fractures happen every year in the UK(76). This contributes to more hospital admissions than from stroke, myocardial infarctions and heart failure combined(77). With an ageing population, these numbers are rising.

A large report from the European union in 2010 aimed to characterise the burden of osteoporosis in Europe(78). They estimated that 22 million women and 5.5 million men suffered with osteoporosis and 3.5 million fragility fractures occurred in that year. 610,000 were hip fractures, 520,000 vertebral fractures and 560,000 fractures of the forearm. The cost of fractures was estimated at €37 billion. 66% of the cost was due to acute fractures, 29% to ongoing fracture care and 5% to medical therapy. Fractures caused a loss of 1.1 million guality adjusted life years. They also noted that

most individuals at high risk of fracture were untreated and there was a reduction in the number of patients on treatment. The estimated medical care costs for osteoporosis itself in Europe were €36.3 billion in 2000. By 2050 it is thought this will reach €76.8 billion. A study from the Ipswich fracture liaison service which serves a population of 320,000 estimated the cost of treating fragility fractures to be £5.3 million over a one-year period(79).

1.8.3.1 Hip fracture

Fractures of the hip are the fragility fracture which cause the most morbidity, mortality and the largest economic burden. It is estimated that worldwide there may be 21.3 million hip fractures per year by 2050, costing €100 billion. 46,000 hip fractures occurred per year in the 1980s. 85,000 emergency admissions to hospital now occur each year in the UK for hip fractures involving 1.8 million bed days and around £1.9 billion of costs for hospital care alone(80). It is expected that by 2036 there will be an increase of 65% of admissions for hip fracture to 140,000 admissions(81). Average length of hospital stay following a hip fracture is 20.4 days and the median cost for a hospital of treating the primary hip fracture is $\pm 8,049(82)$. In the Ipswich study mentioned above hip fractures accounted for £4 million of the total £5.3 million spent(79). One month mortality following a hip fracture is 8% and at this period only 50% of patients will have returned home(83, 84). Up to 25% of hip fracture patients enter long-term residential care and 60% have ongoing difficulties with daily activities of living such as feeding and dressing themselves(85). 50% of patients who previously walked independently are unable to walk unaided for the first year after a hip fracture(83).

1.8.3.2 Further fractures

Distal radius fracture is an injury of particular interest as it is known to have positive predictive value for subsequent fracture at other anatomical sites including vertebrae and the hip which have high associated morbidity, mortality and management costs(86, 87). Improved understanding of the groups most at risk of distal radius fracture may therefore benefit understanding of fracture epidemiology as a whole and also allow identification of groups at risk of further fracture in whom secondary preventative treatment strategies may be of benefit.

There is mounting evidence that individuals with a previous history of fragility fracture are at an increased risk of subsequent fracture(88, 89). Several studies have demonstrated that distal radius fracture pre-disposes to future hip fracture(90, 91). Haentjens et al observed that short term and long term risk of hip fracture was increased after distal radius fracture(92). Their findings suggested distal radius fracture is an early and sensitive marker of skeletal fragility, particularly in men. The greatest risk for hip fracture after distal radius fracture has been shown to be within the first year after the distal radius injury(93, 94). The incidence of osteoporotic fracture, including hip fracture, can be significantly reduced with appropriate treatment of high-risk groups(95-97). Evaluation of patients with a distal radius fracture, and initiation of secondary preventative strategies, represents an economically viable opportunity to reduce the incidence of subsequent, and potentially more serious fragility fractures.

1.9 Distal radius fractures

1.9.1 Epidemiology of distal radius fractures

Distal radius fracture is the commonest adult orthopaedic injury and the commonest fracture treated worldwide. It is generally defined as a fracture of the radius within 3 cm of the distal articular surface and this definition is used in this thesis. Distal radius fracture is a huge burden on health care resources and a large cause of morbidity for patients so an understanding of its epidemiology is of high importance(98).

Approximately 70,000 individuals sustain a distal radius fracture in the UK each year. Estimates for the direct medical cost alone for treatment of fractures of the forearm and wrist in the UK have been predicted to total over £35 million by 2020 and distal radius fracture accounts for the majority of these(99). \$170 million was paid out in US Medicare for distal radius fracture related payments in 2007. 2.5% of all emergency department attendances in Denmark are due to distal radius fracture(100). A bimodal age distribution is well described(101, 102). Injury may result from high-energy trauma in a young patient, but most commonly follows a fall onto the outstretched hand in older patients with underlying osteoporosis. Distal radius fracture is seen more frequently in younger male patients and older women patients. Up to 18% of all fractures in patients aged over 65 years of age are distal radius fractures(103).

Many epidemiological studies of distal radius fracture have been performed in several different countries with varied incidence rates reported, these are likely to reflect the differences in those populations studied along with the methods of patient identification and data capture(101, 104-108) . In a recent prospective multi-centre UK trial incidence rates of 9/10,000 and 36.8/10,000 person-years were reported for men and women respectively(98).

Several studies have suggested distal radius fracture incidence rate is increasing. Whereas others have not shown any significant change in incidence of distal radius fracture with time. In Oslo Lofthus et al. reported no rise in fracture incidence over a 20 year period while Jaglal et al. actually described a fall in the incidence of osteoporotic hip and distal radius fracture in Ontario over a 13 year period(109, 110). It was felt this was due to a simultaneous increase in preventative strategies such as BMD testing and bone-sparing medication treatment. Fracture liaison services have continued to develop significantly over recent years with publication of specific evidence based guidelines. The benefits of these services is well known and has been clearly demonstrated in the UK(111).

Distal radius fracture rate varies with season and temperature. During the summer months Emergency departments treat the most fractures overall due to paediatric injuries when children are involved in more outdoor activities and not at school(112, 113). In winter adult distal radius fracture rate increases(114, 115). This is most likely

due to falls on slippery outdoor surfaces but evidence is limited. An increased risk of distal radius fracture has been demonstrated in studies from Finland and Rochester, Minnesota(106, 116).

The studies from Rochester reported on the effect of weather and seasonality on hip and distal radius fractures over a 38 year period(116, 117). It was noted that wintery conditions were associated with the greatest increase in incidence of distal radius fracture for women aged 35-65. Increased risk of distal radius fracture was present in older women but was less marked. The risk of hip fracture incidence in women aged over 75 was not influenced by weather. They felt the disparities were related to the mechanism and location of the falls. Compared to hip fractures, distal radius fractures more often occur due to a trip or slip outdoors in mobile patients with good neuromuscular control and who try to stop their fall using their hand(118).

1.9.2 Management of distal radius fractures

There are many management options available for fractures of the distal radius. When deciding on the optimum intervention many factors should be considered such as fracture pattern, patient factors, and cost.

Initial assessment of the fracture should involve whether any reduction is required at all. Following that the form of stabilisation, if required, must be decided upon.

Variation regarding optimum treatment for distal radius fractures is high amongst surgeons. A study in which 33 experienced trauma surgeons were shown radiographs and clinical details of 5 cases and then asked about their favoured treatment options demonstrated a lack of agreement. The most popular method of management was volar locking plate (44%), followed by K-wire (28%); manipulation with cast immobilisation (15%); non-locking plate (7%) and; external fixation (5%)(119).

1.9.2.1 Non – operative management

If the position of the fracture is satisfactory it will not require any reduction and could be stabilised in a cast or other form of splintage.

Evidence suggests that patients aged over 65 have similar functional outcomes whether treated surgically or non-operatively. This has been assessed at 12-months with the patient-rated wrist evaluation (PRWE) and disability of the arm, shoulder and hand (DASH) scores. There is some evidence that casts may not even be required or can be removed after a short period of immobilisation. A randomised trial of patients with minimally displaced fractures over the age of 55 showed that early mobilisation of the wrist without a cast ensured rapid recovery without any of the complications which may be seen with casting(120). Deformity and pain did not increase with early mobilisation.

Closed manipulation and cast application is frequently performed for displaced fractures. Fractures can be manipulated using traction into an improved position under haematoma block or with intravenous anaesthesia in the emergency department or fracture clinic, a moulded cast is then applied. This technique can also be performed under general anaesthetic.

1.9.2.2 Percutaneous Pinning (Kirschner wires)

The National Institute for Clinical Excellence (NICE) recommends the use of Kirschner wires (K-wires) for percutaneous pinning of dorsally displaced distal radius fractures if the fracture does not involve the articular surface of the radiocarpal joint or if displacement of the radiocarpal joint can be realigned by closed manipulation(121). Smooth pins are passed through the skin into the fracture fragments to stabilise them. The wires are usually removed in clinic about 4 weeks later. The technique has been used successfully for many years and is quick, easy to perform and low cost. This treatment usually requires a longer plaster cast immobilisation than with plate

fixation. Early patient rated outcome scores are generally poorer than those treated with volar locking plates but several high-quality trials have not shown a consistent significant difference.

1.9.2.3 Open Reduction and Internal Fixation

Open reduction and internal fixation(ORIF) is often utilised for fractures of the distal radius which involve the articular surface or are thought to be unstable. This technique has risen in popularity since the advent of volar locking plate (VLP) technology. Locking plates have screws which lock directly into the plate creating a rigid construct attached firmly to the bone. Non-locking plate screws rely on friction and are biomechanically inferior(122). Plating requires an incision on the volar aspect of the wrist and dissection down to the bone and fracture fragments. The fracture is then reduced and the plate applied directly to the distal radius. Fixation with a volar locking plate should provide an immediate stable construct allowing early mobilisation of the wrist. However, the increased surgical exposure and plate fixation increases the risk of infection, tendon damage, neurovascular injury and metalwork problems.

1.9.2.4 External Fixation

External fixation is used less frequently but is popular in some trauma units. Metal pins are inserted into the bone and then into an external cage or cast. This method is usually used to decrease the load at the articular surface of the radius to allow bone healing in a high energy fracture, often with significant soft tissue injuries(123). The radiocarpal joint can be spanned or left free(124). There is insufficient evidence to state which method of external fixation technique is superior(125). Complications include a high rate of pin-tract infection (14%) and stiffness.

1.9.3 DRAFFT (Distal Radius Acute Fracture Fixation Trial)

The DRAFFT (Distal Radius Acute Fracture Fixation Trial) was a UK multi-centre, National Institute for Health Research (NIHR) funded study which compared the effectiveness of K-wire fixation and volar locking plate fixation for fractures of the distal radius(126). The trial has had a measurable impact on the surgical treatment of distal radius fractures in the UK and led to a resurgence in the use of K-wires(127). The analysis is high quality and detailed so frequently referenced in this thesis. 416 patients were enrolled in the study. Patients who required surgery were randomly allocated to either treatment group and assessed using PRWE. Exclusion criteria was: fracture >3cm from the radiocarpal joint; open fracture; articular surface could not be reduced indirectly; contraindications to anaesthetic; evidence that the patient would be unable to adhere to trial procedures. 73.5% of patients in the study had an extraarticular distal radius fracture.

The outcome of DRAFFT said that volar locking plates "offered no clinically relevant advantage" over K-wire fixation which is "cheaper and quicker to perform". Similar findings have been reported by other authors. Goehre et al. found no difference in DASH scores of 40 patients treated with either volar locking plates or K-wires in the Fritz configuration(128).

Further detailed cost-analysis of DRAFFT was performed by Tubeuf et al(129). They found that the total NHS resource cost was significantly higher in the volar-locking plate group than the K-wire group by £903 (260.43, p<0.001).

0.008 quality-adjusted life-years (QALYs) were gained by volar locking plate fixation over Kirschner wires, which is the equivalent of 3 days' good health per year. These were calculated using EQ5D scores with an incremental cost-effectiveness ratio (ICER) of £89,322 per QALY. This is much higher than what is considered acceptable by NICE (£20,000 - £30,000). They concluded that volar locking plates were not cost-effective in the treatment of patients under the age of 50.

1.10 Distal radius fracture radiographic parameters

Plain radiographs of the wrist are mandatory in any patient with a possible distal radius fracture. They are used for diagnosis, prognosis, planning of appropriate treatment and subsequent radiographs can assess healing and further position of the fracture. The fact that almost every patient with a distal radius fracture should have a plain radiograph means that radiographic records can be used to capture almost all patients who sustain a fracture. A large number of radiographic parameters may be measured during the evaluation of distal radius fractures. These include radial height, ulnar variance, dorsal/ palmar tilt, carpal alignment, and intra- articular gaps and steps. These radiographs also include the carpal bones and metacarpals so other information can be gained from them. Some methods exist to estimate bone health from these plain radiographs.

There is minimal evidence to demonstrate the reproducibility and validity of radiographic measurements of distal radius fractures. Most studies were carried out before the modern digital software now commonly used to measure radiographic parameters. Kreder et al. developed a standardised method of measuring eight anatomic parameters at the distal radius to test inter-observer variation(130). Sixteen observers measured these parameters on six radiographs of healed distal radius fractures. The results showed high rater agreement in measurements of palmar/dorsal tilt, ulnar variance and radial shift, but poor inter-observer agreement of intra-articular step and gap. Johnson and Szabo reviewed inter-observer variability in assessment of radial angle and palmar tilt in a cadaver study(131). Mean standard deviation was 3.2 degrees for radial angle, 3.6 degrees for conventional lateral palmar tilt, and 2.1 degrees for 15 degrees lateral palmar tilt.

1.10.1 Radiographic parameters and outcome

A large amount of literature investigating the relationship between radiological parameters and functional outcome exists but the findings are inconsistent. The quality of evidence is low. Most studies are case series and often retrospective. Some expert groups have issued recommendations regarding how much displacement can be accepted before fixation is required but they also acknowledged that the evidence supporting this is weak(132, 133).

Poor functional outcome after malunion with radial shortening or increased ulnar variance has been reported by several authors. Restoration of length has been advocated as the most important factor. In 143 consecutive patients prospectively reviewed by Brogren et al DASH outcome score was significantly higher in those with positive ulnar variance of 1mm or greater(134). In a randomised control trial comparing different treatment methods for distal radius fracture McQueen et al noted that positive ulnar variance of greater than 3 mm, compared with the unaffected side, was associated with reduced grip strength(135). In 78 patients retrospectively assessed after 22 months by Wilcke et al radial shortening ≥2 mm, dorsal angulation >15°, and radial angulation >10° were each significantly associated with poorer DASH scores(136).

In a large study reviewing 260 patients managed non-operatively Finsen et al reviewed 260 patients treated 6 years previously(137). Bivariate regression analysis disclosed a significant relationship between displacement and adverse outcome. The variables studied only explained 23% of the variability of the clinical outcome with radiographic parameters such as dorsal angulation, ulnar variance, and radial inclination only accounting for 11%. The authors concluded that alignment following distal radius fracture only has a minimal influence on clinical outcome. Chung et al found only age and income were significantly associated with outcomes 1 year after surgery in a prospective study of 66 patients treated with volar plating(138). Plant et al recently compared PROMs and physical measures of function with the radiographic measures of palmar tilt and ulnar variance in 50 patients who were treated in DRAFFT(139). The radiographic parameters correlated poorly with physical function measures and patient rated outcome scores at every point assessed (3, 6 and 12 months). The authors felt the results raised concerns about the use of radiological

parameters to determine management and that restoration of 'normal' radiographic parameters may not be necessary to achieve a satisfactory functional outcome.

Age has also been demonstrated to influence the association between malunion and outcome. Grewal et al found that malalignment of the distal radius is associated with increased reported pain and disability in patients aged under 65 years of age(140). In patients aged 65 years and over no radiographic parameter influenced PRWE or DASH scores. There was a decreasing trend with increasing age of a poorer outcome with malalignment, with a significant reduction after 65 years. Other studies have reported similar findings in older groups of patients

A persistent intra-articular step predisposes to the development of radiocarpal degenerative change. Knirk and Jupiter showed a step of 2mm or more led to a 100% incidence of radiological arthritis in 40 young adults at mean follow up of 6.7 years(141). Catalano et al. found 76% of patients with residual intra-articular displacement 7 years after internal fixation of a distal radius fracture had evidence of arthritis(142). This had progressed when reassessed after 15 years by radiographs and CT scans(143). Cadaveric experiments have shown that an articular step affects the biomechanics of the joint with a step of 1mm causing a significant increase in contact stresses(144).

Despite the evidence that articular incongruity leads to radiologically proven arthritis the correlation with symptoms and poor outcome is debatable. Studies by Trumble et al and Chung et al showed that residual articular displacement was associated with a poorer outcome(138, 145). 93% of the patients in Knirk and Jupiter's study were symptomatic. However 61% reported a good or excellent outcome and only one patient who had bilateral fractures had to stop work due to their injury (141). The only functional limitation seen in the 15 year review of Catalano's original study was an insignificant reduction in wrist flexion(143). Kreder et al. found a 9.9-fold increase in the risk of radiological osteoarthritis with persistent articular incongruity after plate fixation but there was no functional difference at one year follow-up(146). Forward et al. retrospectively reviewed 106 young adults with intra-articular distal radius fractures at a mean of 38 years and found DASH (Disabilities of Arm, Shoulder and Hand) scores were not different to population norms and functional impairment was

less than 10% when assessed by the Patient Evaluation Measure(147). Kopylov et al. looked at patients who had sustained a wrist fracture 30 years previously(148). Radiographic osteoarthritis was related to articular incongruity but complaints were limited; 87% of patients reported no difference between their injured and uninjured sides.

1.11 Variation in surgical treatment

Optimum management of distal radius fractures, particularly which fractures require surgical fixation, can be controversial. As discussed above some studies have suggested that only a small amount of displacement can cause a poor outcome whereas others have reported that even with significant malunion long term functional outcome is satisfactory(141-143, 147, 148). Deciding when intervention is required is therefore difficult and high quality guidelines do not exist. Some clinical groups have provided guidance but the evidence to base this on is varied and often low quality(132, 133)

Variation in surgical treatment rate is known to occur in many different interventions including orthopaedic procedures. Studies from the US and the Netherlands have reported variation in distal radius fracture fixation rate(149-151). Variation can be classified as warranted or unwarranted. Some variation is inevitable and random. Warranted variation may describe high quality clinical care. Rates of treatment interventions vary due to individual needs of specific patient populations(152). Innovation and improvement may cause initial variation. This will decrease after the spread of knowledge and effective innovative methods across trusts.

Unwarranted variation suggests low quality healthcare with activities of low value and waste of resources. There are two main problems – under use of effective interventions and over use of low value interventions. Certain groups such as those from deprived backgrounds may not utilise effective interventions. Over use of low value interventions potentially exposes patients to risk from non-essential treatment,

wastes resources and diverts funding from other patients and better treatments. Understanding the causes of variation allows these problems to be identified and action taken. In this thesis investigation is carried out to see if variation is present in the surgical treatment rate of distal radius fracture in England and to attempt to identify causes for this.

1.12 Quality of care

Measuring quality of care for surgical procedures is challenging. Donabedian et al described three domains: structure, process, and outcomes(153).

Structure relates to the systems and settings in which the care takes place such as ratio of staff to patients and utilisation of electronic records. In surgery the most common structural measure is volume of procedures performed, which is known to be linked to outcome(154). Process measures relate to care that patients receive and may be related to clinical guidelines such as the proportion of diabetics who receive retinal screening or uptake of vaccinations in high risk groups. Outcome measures are regularly used for surgical procedures. The most commonly used is operative mortality but there are many other measurable outcomes such as length of stay, complication rate, readmission rate and patient rated outcome measures.

In this thesis measures of care quality such as operative rate by hospital trust will be reviewed along with direct outcome measures such as length of stay and wait for surgery and the relationship with deprivation investigated.

1.13 Aims of thesis

The overall aim of the work described in this thesis is to improve current understanding of the association between social deprivation and distal radius fractures. Deprivation is known to be associated with many types of fracture but the

reasons for this are not clearly understood. I will attempt to establish if deprivation is a risk factor for distal radius fracture and what are the important individual factors related to this. A distal radius fracture is associated with increased risk of hip fracture. This increased risk will be quantified. Further work will investigate whether deprivation influences treatment of patients with distal radius fracture.

The key research questions addressed are:

Is distal radius fracture associated with social deprivation? Does social deprivation influence distal radius fracture treatment? What is the risk of hip fracture after distal radius fracture?

1.14 Overview of thesis

Chapter 2 comprises the general method for the whole thesis. This is referred to in the subsequent individual results chapters and some further specific method detail is found in these chapters. Chapter 3 then describes the reliability and reproducibility of the radiographic measurements used in the thesis. This is important information for clinicians and researchers who frequently carry out these measurements and it is crucial to ensure the radiographic measures used to draw conclusions from in the thesis can be relied upon. Chapter 4 provides a 10 year review of the epidemiology of distal radius fracture in Leicestershire to establish the incidence and burden of this injury. Chapter 5 then investigates the association between deprivation and distal radius fracture in Leicestershire and England. Chapter 6 comprises analysis of variation and care quality for surgical intervention for distal radius fracture including the effect

of deprivation. Chapter 7 reports the results of a prospective survey investigating the relationship between deprivation and mechanism of injury, falls risk and osteoporosis risk in patients aged 50 years and over. Chapter 8 contains a systematic review and meta-analysis of the literature investigating the rate of hip fracture after distal radius fracture. This further establishes the importance of distal radius fracture and its relationship with subsequent fractures. Chapter 9 includes an overarching discussion whilst a final summary and plans for future directions are included in chapters 10 and 11.

Chapter 2 Method

This thesis uses a wide variety of techniques to investigate the research questions. In this chapter the general methods used in the thesis overall and subsequent results chapters are described. Some further specific detail is then provided in the individual results chapters. The results chapters include links back to the relevant methods in this chapter so they can be easily located.

2.1 Settings and population

University Hospitals of Leicester (UHL) NHS Trust provides care to a population of approximately 1 million people living in the city of Leicester, and the surrounding counties of Leicestershire and Rutland. The three areas are grouped together for healthcare provision. In this thesis when I refer to Leicestershire this will describe all three areas unless stated otherwise. The city of Leicester itself had a population of 329,839 in the 2011 Census(155). The city is diverse with 50.6% of the population being white, 37.1% Asian (28% Indian) and 6.3% black. The surrounding County of Leicestershire has a population of 650,500 people. 89% of people are White British, the largest ethnic minority group is Indian who make up 4% of the region. Rutland is the smallest county in the United Kingdom with a population of just 37,400 with only 2.7% of people from ethnic minority backgrounds.

Almost all patients sustaining an orthopaedic injury within these areas are referred for review and follow up at UHL, allowing comprehensive capture of all such presentations within the resident population. There are three hospitals in the city but only one, Leicester Royal infirmary, has an emergency department. This is one of the biggest and busiest emergency departments in the United Kingdom. A small number of minor injuries units may see and treat distal radius fractures in the region but these numbers are very low. Most of these patients would be referred for further management at UHL and almost certainly all displaced fractures which may require

surgery would be referred. Therefore, capture of distal radius fracture data for the region is likely to be very high.

The city of Leicester is ranked as the 21st most deprived area in England in the most recent IMD 2015 report, this is within the 10% most deprived local authorities in England(156). The county of Leicestershire has low levels of deprivation with Leicestershire and Rutland being the two least deprived local authorities in the East Midlands. Comparison with other rural counties again shows Leicestershire to have low levels of deprivation.

2.2 Epidemiology of distal radius fracture in Leicestershire

2.2.1 Patient identification

Details of all patients presenting to UHL with a radiographically confirmed fracture of the distal radius between January 1st 2007 and December 31st 2014 were identified from the Emergency department database. Data is prospectively recorded for all patients that attend the Emergency department. Data was requested and obtained for all patients recorded as having an open or closed wrist fracture, distal radius fracture, Colles fracture, fracture lower end of radius or distal forearm fracture. Data on age, gender, laterality of injury, and the location and date of the injury were collected. A computer generated random sample of 100 patient's radiographs were assessed to check the accuracy of the diagnosis using radiographs and this revealed the accuracy of the ED diagnosis was 97%.

The population at risk was identified by obtaining the adult (18 years and older) population data for Leicester, Leicestershire and Rutland for 2007-16 inclusive from the United Kingdom Office of National Statistics (ONS)(157). The total population at risk was calculated by the addition of the numbers for the three regions. This was calculated for each year of study to ensure accurate incidence rates for that year.

Data were categorised primarily by gender, and age of under and over 50 years of age. Age groups were split at 50 years because bone mineral density is known to reduce after 50 years of age(72). This therefore divides the patients into those who have suffered osteoporotic fragility fractures and younger patients who generally have good bone quality and have suffered injuries from higher energy trauma. This method has been utilized in recent high-quality trials and allows comparison with those results(126). Groups of 10-year age bands were made for some analyses. Season and day of the week of fracture was determined from the date of injury. Location of fracture, when recorded, was reported as occurring at home, in a bar/pub, in the street, in a public place, in the workplace, during sporting activity, following road traffic collision (RTC), or other location.

Date of presentation to ED was used to determine the day of week and season at presentation. Poisson regression modelling was used to investigate trends in fracture incidence, day and season of injury over the study period.

Projections for the future number of distal radius fracture in the UK were made using the calculated fracture incidence rates for Leicestershire and population predictions from the ONS.

2.3 Effect of deprivation on distal radius fracture rate in Leicestershire

From the data collected above a four year sample from 2007 to 2010 was identified for further investigation of the relationship between deprivation and distal radius fracture rate. 4,355 adult patients who attended Leicester Royal Infirmary with a distal radius fracture from January 2007 to December 2010. 77 (1.8%) patients from outside Leicestershire were excluded from the study group as comparison with local data would be inaccurate leaving 4278 (98.2%) for analysis. 3481 patients were white and 797 of other ethnicities (South Asian 629, Black 51, Mixed race 30, Other 87). Patient demographics including sex, age, ethnicity, location of injury, date of presentation and postcode were recorded. Location data was available for 4054 patients (95%).

2.3.1 Index of Multiple Deprivation

The Index of Multiple Deprivation 2010 (IMD 2010) was chosen as the most appropriate measure of deprivation(158). This is the official UK government measure of deprivation and is now frequently used in most deprivation related research. The 2010 version is the most appropriate to the time period studied in this thesis. Quintiles and ranking are used for statistical analysis. Postcodes were used to link each patient to their respective LSOA with its assigned IMD 2010 score and quintile. Accurate post code data was recorded for all patients so IMD score and quintile was available for every patient in the study group.

2.3.2 Incidence rate and deprivation quintile

Population data is available for each LSOA in the studied region by gender, ethnicity, deprivation scores in the following age bands: 16-34, 36-49, 50-64, 65+(155). This data was used to calculate the total denominator of people in each deprivation quintile by gender, ethnicity and age groups.

As I wished to investigate adult fractures in those aged 18 and over estimates were calculated for the age band of 18-49 years to remove those aged 16 and 17 years. Total LSOA population by each year of age and gender is available but this does not include ethnicity or deprivation scores. The relevant LSOA total populations of men and women aged 16 and 17 was obtained. The proportion of each gender and ethnicity per quintile in the 16-34 age group was calculated. These proportions were then applied to the total number of men and women aged 16 and 17 to estimate how many people were in each group. These gender, ethnicity and quintile specific estimates were subtracted from the appropriate group. Groups were then added together to find the denominator by gender in the age groups 18-49 years and 50 years and over.

Incidence rates by quintile, ethnicity and gender were calculated by dividing the number of fractures per year by the population at risk in that quintile.

2.3.3 Surgical treatment

Identification of which patients were treated with surgery was gained using hospital theatre books. These records are kept within the operating theatre and members of staff record the exact procedure, operating surgeon, anaesthetist, patient details and the time and date of the procedure. These generally provide accurate data about surgical procedures. For this detailed four year analysis theatre books were reviewed from Leicester Royal infirmary and Glenfield General Hospital. The vast majority of trauma surgery is carried out at the Leicester Royal infirmary. However, in some very busy periods, such as during cold weather, with a large increase in patients sustaining a distal radius fracture then some additional surgical procedures were carried out at the Glenfield General Hospital. Detail of the dates of the procedure, the type of the procedure performed and patient demographics were obtained from the theatre books.

2.4 Radiographic measurements and deprivation

Radiographs of all patients with a distal radius fracture who underwent surgical intervention in Leicester(n=618) were reviewed along with a 10% computer generated random sample of those treated non-operatively (n=367). Fractures of those who had surgical treatment were classified as intra or extra-articular. Bone density was calculated from the second metacarpal cortical width using the Metacarpal Index (MCI). MCI has been demonstrated in several studies to be a successful assessment of bone mass, bone quality and a predictor of osteoporotic fracture. Dorsal tilt was chosen as a measure of initial fracture displacement. Dorsal tilt measures the angle of the joint surface on a lateral radiograph and is regarded by clinicians as one of the

most important radiographic parameters to measure. Displacement and change in dorsal tilt is most often seen with high-energy trauma and in patients who sustain a fracture with underlying bone fragility. Normal tilt is approximately 11° of volar angulation. Tilt most often become more dorsally angulated following a fracture but it occasionally displaces further volarly depending on the direction of the force causing the injury. To measure displacement the degree of tilt from the normal position of 11° was recorded for all radiographs reviewed. Displacement was measured on the first radiograph available following injury and before any intervention. 219 patients sustained intra-articular fractures. Intra-articular step was measured for these fractures using the longitudinal method. Increasing intra-articular step displacement is also more likely to be seen in fractures caused by high energy trauma or in patients with underlying fragility. Radiographs were evaluated by the primary author and two co-authors following a standardised teaching session.

The association between measures of displacement and MCI with deprivation rank and quintile was then investigated. This was repeated for gender, age and ethnicity specific sub groups.

For any meaningful observations to be taken from the radiographic analysis in this thesis it is essential that the method of measurement was reliable and reproducible. Considerable work was undertaken to ensure this and results are described in chapter 3.

2.4.1 Reliability and reproducibility of radiographic measurements

2.4.1.1 Extra articular radiographic measurements

Two core surgical trainees, both with an interest in pursuing careers in orthopaedics, but with limited clinical experience of the specialty (4 months at the point of training) were trained to perform radiographic measures using a standardised method in a single calibration session. The four most frequently used radiological parameters in

clinical practice were chosen for investigation. Two studies have previously given a detailed description of these measurements, and the methodology used to calculate them as follows(159, 160).



Figure 1 Method of measurement for the extra-articular radiographic parameters of dorsal tilt, radial inclination and radial height

Dorsal tilt (Figure 1) is measured on the true lateral view. It is the angle created between a line drawn between the most distal points of the dorsal and volar lips of the distal radius, and a second drawn perpendicular to the long axis of the radius. Radial inclination is the angle between a line from the tip of the radial styloid to the medial edge of the articular corner of the radius and a line perpendicular to the long axis of the radius. It is measured on the PA view. Radial height is measured on the postero-anterior (PA) view and refers to the distance between a line drawn tangential to the tip of the radial styloid and another tangential to the most distal part of the ulnar head.

All measurements were made using the hospital trust online Picture Archiving and Communication System (PACS) (Agfa HealthCare IMPAX 6.5.2.657). Training involved an individual calibration session, lasting approximately 1 hour, with a senior orthopaedic registrar and a Professor of Hand and Orthopaedic Surgery. A series of ten trial radiographs were measured, with results compared with those of the tutors to check understanding and accuracy of measurements.

A random 10% sample (n= 367) of patients treated non-operatively was computer generated by an independent observer from the Leicester database. No formal power analysis was performed to determine sample size, which was guided by previous investigation of inter and intra-observer reliability of assessment of distal radius fractures(161). All radiographs were reviewed independently by the two trainees with record of each parameter made to 0.1 millimetre and 0.1 degree as appropriate. Repeat measurements of radial height, radial inclination and dorsal/palmar tilt of a further computer generated random 10% sample (n= 37) were made four weeks later to check intra-observer correlation. This was not performed for intraarticular step/gap due to a small sample size. Correlation was assessed using the intraclass correlation coefficient (ICC).

2.4.1.2 Intra articular radiographic measurement

The original sample of 367 patients used for extra-articular measurements only contained 20 patients with an intra-articular fracture. Following some initial exploratory results, it was felt important to investigate the reliability and reproducibility of the intra-articular measures of displacement further.

Two different junior orthopaedic trainees, at a similar stage of their training to the previous two observers were trained to measure intra-articular displacement in a single calibration session as previously described.

Articular step and gap (Figure 2) were measured on the true lateral view using the longitudinal axis method as described by Cole et al. as follows(160). Two points are marked at the subchondral fracture margin of the two most displaced fracture fragments (points A and B). A line (line 1) is drawn parallel with the long axis of the radius through the more central of the two marked points (point A). A second line (line 2) is drawn perpendicular to this, passing through point B. The intersection of line 1 and 2 is marked point C. Step displacement is represented by the distance between points A and C, with gap displacement the distance between points B and C.

Figure 2 Lateral radiograph showing measurement of intra-articular step-off (AC) and gap (BC) using the longitudinal axis method



A power calculation was performed to determine the sample size required. For ICC of 0.2 at 80% power with a significance level of p=0.05, 192 radiographs were required for review but slightly larger samples were used as it was expected that a small number of radiographs may not be suitable for measurement. Radiographs from 200 consecutive adult (age >18) patients presenting with an intra-articular distal radius fracture to UHL Emergency Department in 2015 were prospectively recorded. All radiographs were reviewed independently by the two trainees with record of each parameter made to 0.1 millimetre. Repeat measurements of a computer generated random 10% sample (n= 23) were made four weeks later to check intra-observer correlation. Correlation was assessed using the intraclass correlation coefficient (ICC). The ICC was calculated for all fractures, and then repeated for fractures with displacement ≥ 2mm. The value of 2mm was chosen as it is the amount of intra-articular displacement which would trigger surgical intervention for most surgeons.

2.4.2 Metacarpal Cortical Index

The Metacarpal Cortical Index (MCI) was originally developed in the 1960s for the assessment of Marfan's syndrome and other skeletal conditions associated with arachnodactyly(162). MCI is a measure of the combined width of the cortical bone in the metacarpals expressed as a ratio to the width of the shaft of the metacarpal. MCI can be thought of as the fraction of the metacarpal shaft in cross section occupied by cortical bone. It was first used by Barnett as a diagnostic tool for osteoporosis(163). Although not widely used as an assessment of BMD, MCI has been shown to be significantly associated with known risk factors for osteoporosis (high age, low BMI and smoking) and to be a strong predictor of wrist, hip and vertebral body fracture(164, 165).

In order to be diagnostically useful as a cheap and readily available measure of BMD, MCI needs to be reliable and reproducible, correlate to DXA scan results and be associated with useful clinical outcomes. This analysis aims to determine the interobserver and intra-observer reliability of MCI, to investigate its use as a measure of BMD compared to central DXA scanning and to study its relationship to patient age and fracture displacement in distal radius fractures.

2.4.2.1 MCI measurement

The original randomly generated 10% sample of 367 patients was again used. 32 patients were excluded as they did not have an AP and lateral radiograph of the distal radius including the 2nd metacarpal available on the UHL digital image viewing software leaving 91% of the original sample (n=335).

The radiographic images for each patient were reviewed independently by two Core Surgical Trainees with no previous experience of measuring MCI. Both reviewers underwent a further standardised training session on how to measure these variables as outlined below.

Measurements were taken of the Inner Cortical Diameter (ICD) and Outer Cortical Diameter (OCD) of the narrowest part of the 2nd metacarpal shaft on an AP radiograph as demonstrated in Figure 3. Measurements were taken to 0.1 millimetre. MCI was then calculated by using the following formula and results rounded to 2 decimal places:

MCI = (OCD-ICD)/OCD



Figure 3 Plain film AP radiograph of a distal radius and hand demonstrating how ICD and OCD measurements are taken from the 2nd Metacarpal

A further random sample of 10% (n=37) of patients from the initial sample underwent repeat measurements of the MCI by the same reviewers 4 weeks after the initial measurements to assess intra-observer correlation. Inter-observer and intra-observer correlation was again assessed using the Intraclass Correlation Coefficient (ICC).

2.4.2.2 MCI and bone fragility

A search was conducted of the UHL Picture Archiving and Communication System (PACS) imaging database to identify all patients who had undergone DXA scanning of the spine and plain film imaging which included an AP radiograph of the 2nd metacarpal between June 2013 and June 2015. 104 consecutive patients who had these two investigations within 90 days of each other were identified. The BMD and the T-score of the spine from the DXA report was recorded and the MCI was calculated by both reviewers for each patient in the same way described above. The Tscore was included in the analysis in addition to the BMD as T-score is used by the WHO to define osteoporosis and to guide treatment in many algorithms(166). Further models were constructed to investigate the relationship between MCI with BMD and T-score.

Radiographs were performed according to a standardised hospital protocol, which remained constant throughout the study period for all parameters. The PA view was performed with the wrist and elbow at shoulder height, aligning the joints in the transverse plane, with the hand placed on the cassette, palmar surface down. In this position the radius and ulna are parallel. The lateral view was performed with shoulder, elbow and wrist joints aligned in the sagittal plane, with the ulnar border of the wrist on the cassette.

2.4.3 Intraclass Correlation Coefficient (ICC)

Reliability describes the amount to which measurements can be replicated. This reflects correlation and agreement between measurements. Several tests are commonly used but many only measure agreement (paired t test, Bland-Altman plot) or correlation (Pearson correlation coefficient). Intraclass Correlation Coefficient (ICC) is a useful measure as it incorporates both. The Consensus-based Standards for the selection of the health status measurement instruments (COSMIN) check list was developed to assess methodological quality of studies based on measurement attributes(167). This recommends the ICC as a measurement of inter-rater reliability.

ICC is calculated by mean squares through analysis of variance. 10 forms have been described based on the model (1-way random effects, 2-way random effects, or 2-way fixed effects), type (single rater or mean of several raters) and the definition considered (consistency or absolute agreement). ICC is popular and easy to interpret. A value is obtained between zero and one. One represents perfect reliability whereas zero indicates no reliability. Problems with ICC can include errors selecting and reporting the correct form. ICC is influenced by the variance of the trait in the sample assessed. Therefore, ICC measurements from different samples may not be comparable.

In this thesis the ICC(2,1) form is used to measure reliability. Each radiographic parameter is measured by each rater, and raters are considered to represent a larger population of similar raters. The reliability is calculated from each single measurement. This method is easy to interpret, widely used and appropriate for the evaluation of radiological/clinical assessment methods that are used in routine clinical practise by other clinicians with similar expertise. Sample size was calculated based on a method by Zou which treats interval width and lower limit as random variables and includes a preferred assurance probability(168) (Appendix 7).

2.5 Risk of hip fracture after distal radius fracture

2.5.1 Study group

To identify the risk of hip fracture after distal radius fracture a systematic review and meta-analysis was performed. A multi-disciplinary advisory group was created including a senior clinical librarian, hand and wrist surgeons and a statistical expert. The protocol was reviewed by the group and after each stage of the review and analysis process a meeting of the advisory group took place at which point any issues were resolved by consensus.

2.5.2 Search strategy and data sources

The senior clinical librarian(PD) searched Medline, Embase, Cochrane CENTRAL database and CINAHL from their inception to July 2015, using the following search terms and subject headings: distal radius or radial, or wrist or colles or smith and fracture*; recurrence or refracture or subsequent or fragility or osteoporo* or predict or future; hip* or femur* or femoral or trochant* or pertrochant* or intertrochant or subtrochant* or subcapital* or extracapsular*. These search terms were selected from previous studies and through discussion of the topic in meetings of the study advisory group.

The abstracts of articles were then screened and filtered according to the study inclusion and exclusion criteria. The authors also retrieved potentially relevant articles and reviewed their reference lists for additional articles. There were no language restrictions to the search.

2.5.3 Study selection

To be eligible studies had to meet the following inclusion criteria defined in our protocol(Appendix 1): Men or women suffering a radiographically confirmed wrist fracture and a subsequent radiographically or surgically confirmed hip fracture. The relative risk(RR), odds ratio or hazard ratio(HR) (with p value or 95% confidence intervals) must be reported or the actual number of outcome events provided. Study types for inclusion were prospective longitudinal studies or high quality retrospective studies.

2.5.4 Data extraction and quality assessment

After initial screening, full papers were obtained of those selected and assessed against the inclusion criteria by two independent reviewers (NJ, ES). Difference in assessments were resolved by consensus with the study advisory group.

Data extraction and quality assessment was performed for the studies selected following the second stage of screening. Data extraction was carried out using a modified Cochrane data collection form (Appendix 2). Data items collected included study type, source of funding, setting, start/end date, length of follow up, number of patients with distal radius fracture, method of diagnosis, number of patients with subsequent neck of femur fracture, time to hip fracture and RR, HR or odds ratio.

Quality assessment was carried out using a modified Coleman score (Appendix 3) and the Newcastle Ottawa scale for cohort studies. An assessment of bias was performed for each individual study using a modified Cochrane Risk of Bias tool (Appendix 4). Findings were reviewed at a meeting of the advisory group and studies of poor quality or high risk of bias were excluded.

2.5.5 Statistical analysis

A pooled RR with 95% confidence intervals was estimated from the RR/HRs or odds ratios and confidence intervals reported in the studies. This was calculated from the average of the logarithms of the individual relative risks weighted by the inverse of its variance. HRs and odds ratios were considered to correspond to RRs.

The I-squared and tau-squared statistics were used to evaluate the statistical heterogeneity. Random effects and fixed effects models were both calculated. With low heterogeneity a fixed effects model is appropriate. The random effects model was produced using the Der Simonian and Laird method. Analysis was performed using the meta package (version 4.1-0) in R.

2.6 National data

2.6.1 Hospital Episode Statistics data

Hospital Episode Statistics (HES) data is described as a data warehouse. It is data collected whenever a patient attends an NHS hospital in England and contains details of emergency department visits, outpatient appointments and admissions. The data is reported annually from April to March in the following year. It includes private patients treated in NHS hospitals and those resident outside of England. The data is used to pay hospitals correctly for delivering care. Each record for each period of care is stored separately. A huge amount of information is available. This includes the age of the patient, gender, ethnicity, diagnosis, comorbidities, procedures performed and the individual IMD deprivation score for that patient. Other information includes the day a patient was listed for surgery, if day case surgery was planned and if it was carried out as a day case, and length of stay in hospital. Wait for surgery can also be calculated using the operation date and date of presentation or date listed for surgery. HES data was obtained from April 2009 to March 2013. This was used to answer several research questions regarding deprivation and surgical treatment.

2.6.2 Diagnosis

HES data lists diagnoses and surgical interventions using ICD-10 and OPCS codes. The International statistical classification of diseases and related health problems (ICD-10) is a comprehensive list of codes for diseases, injuries, signs, symptoms and related social problems(169). It is created and maintained by the world health organization. The 10th revision was completed in 1992. The OPCS classification of interventions and

procedures is the national health service classification list covering procedures and interventions performed in UK NHS hospitals(170).

A search algorithm was created to filter the whole HES dataset and identify those who had an acute distal radius fracture from ICD-10 codes (International Classification of Diseases: ICD-108 codes S525 - fracture of lower end of radius - and S526 - fracture of both lower end of radius and ulna). OPCS codes were used to identify all methods of primary surgical intervention including plating, Kirschner wires and external fixation.

59,802 surgical procedures were performed in 130 hospital trusts during the four year period. This data was used to analyse rate of variation per hospital trust. Full data including ethnicity, deprivation and date of surgery was known for 59,315 procedures. This data was used for modelling the effects of deprivation and ethnicity on treatment rates and measurements of quality of care.

2.6.3 Deprivation, ethnicity and rate of fixation

To investigate the effects of deprivation and ethnicity on national fixation rate the background populations must be known. Deprivation data for England by decile is available for ethnicity and gender in age bands of 0-24, 24-49, 50-64 and 65+. Population data by IMD decile is available for 0-15 year olds, this is split by gender but not ethnicity. Number of people aged 16 and 17 years old by gender and ethnicity is available for England. A similar method as previously described in section 2.3.2 was used to create gender, ethnicity, deprivation group estimates for those aged under 18 and subtracted from the 0-24 group.

The proportion of each ethnic group by decile and gender was calculated for the 0-24 age group and these proportions applied to the 0-15 years population data and estimates for each deprivation and gender group calculated. These proportions were then applied to the population data for 16 and 17 year olds. These ethnicity, age and gender specific estimates were subtracted from the 0-24 group to give a 18-24 group. This was then used to create ethnicity gender specific quintiles for 18-49 year olds and
50 years and over by combining deciles and the age groups. Number of procedures performed in each subgroup was gained from the HES dataset. 59,315 patients were included in this part of the analysis. 95% (56,372) of patients were white, 2.5% (1,482) Asian, 1.7% (1,017) other named ethnic group and 0.7% (444) black.

2.6.4 Analysis of variation

The NHS atlas of variation series provides reports on national variation of important health treatments and investigations in the UK with the aim to reduce unwarranted variation. The first report was published in November 2013 and demonstrated clear geographical variation in health care access and service provision levels. The methodology used involves excluding the seven trusts with the highest and lowest rates to remove outliers(152). The magnitude of the difference between the remaining highest and lowest rates is then reported. The method is used in this thesis to investigate variation in distal radius fracture surgical intervention. A negative binomial model is also produced to assess the influence of deprivation, hospital trust, hand specialist unit and trust population characteristics on surgical intervention rate.

2.6.5 Temperature

Weather is known to affect the incidence of injuries and low temperatures increase the risk of falls and distal radius fracture. Minimum daily temperature was gained from the Centre for Environmental Data Analysis for each day of the study period(171). This was then linked to each patient's date of presentation with a distal radius fracture and used as a variable for the statistical modelling.

2.6.6 Hospital trust data

Summary data for each hospital trust was obtained(172). This provided a mean patient IMD score for the trust and the total population of the trust split by gender and age groups. Number of women and number of patients aged over 65 years are included in some models as fragility fracture is seen more frequently in these groups.

2.6.7 Comorbidities and Charlson index

ICD codes were also used to identify medical co-morbidities. Diabetes was included as a separate variable in some models as it was felt it may directly influence type of treatment, wait for surgery and whether the patient is suitable for day case surgery. The Charlson co-morbidity index predicts 10 year mortality based on medical comorbidities(173). Scores are given for certain significant co-morbidities and increasing age. A higher score indicates increased mortality rate. An algorithm was used to calculate the Charlson score for each patient from the HES data.

2.6.8 Semi-elective trauma surgery

2.6.8.1 NICE guidance

The National Institute for clinical excellence is a non-departmental body of the Department of Health which aims to improve health and social care through evidencebased guidance. High-quality reviews of evidence are produced from expert multidisciplinary groups to provide recommendations and clinical guidelines. Adherence to the guidelines may influence funding which a hospital trust receives.

Recent NICE guidance has recommended that intra-articular fractures undergo surgery within 72 hours and extra-articular fractures undergo surgery within one week(121). The rationale for this is that within 72 hours organised haematoma may form which will prevent accurate reduction of intra-articular fragments. A significant portion of distal radius fractures requiring surgery will have intra-articular displacement. These complex fractures will often require treatment by an experienced trauma or hand surgeon. To ensure these guidelines are met hospitals must therefore have experienced specialist surgeons available at almost all times.

Day case surgery is safe, cost efficient and popular with patients. With new anaesthetic techniques, the use of day case surgery for fracture fixation is becoming increasingly popular. Fractures of the upper limb can often be safely treated in this manner. This frees up hospital beds and saves resources. Some concern does exist that if patients are sent home to wait for this "semi-elective" trauma surgery that they may end up having their surgery delayed if patients are admitted with more serious injuries or injuries which necessitate hospital admission such as a fractured neck of femur. As they are not occupying a bed then they are more likely to be cancelled or delayed in times of many acute admissions.

2.6.9 Hospital Episode Statistics analysis

Using the data described above a series of regression models were designed to investigate the relationship between several measures of the quality of treatment with IMD score along with a large number of independent variables in patients who required surgical fixation for a displaced distal radius fracture (Kirschner wires, plating, external fixation).

Independent variables included:

Patient details: age, gender, ethnicity, comorbidities, Charlson index, patient IMD score, procedure

Hospital trust details: total population, population aged 65 or over, population of women, cases performed per year, cases performed that month

Date/seasonal details: year, day of week, month, cases performed per year per trust, cases performed that month per trust, minimum temperature

Treatment factors investigated were:

Rate of variation of surgical fixation

Pre-operative wait for surgery Rate of daycase surgery Post-operative hospital stay Wait for semi-elective surgery

2.7 Prospective investigation of fracture risk

The primary aim of this part of the study was to identify if deprivation was associated with falls risk, mechanism of injury or osteoporosis in patients with a distal radius fracture.

2.7.1 Survey method

Using the same emergency department database all patients with a distal radius fracture were prospectively identified following attendance at the Emergency Department. This component of the study was carried out for a 12 month period from April 2015 to March 2016. All radiographs were analysed to ensure only patients with a confirmed distal radius fracture were included in the study. Patients from outside the region or of no fixed abode were excluded as deprivation data would not be available. Death checks were carried out and any patients who died were removed from the study.

Following identification of suitable patients, a questionnaire was posted to them with a stamped addressed envelope for return. Non-responders were sent up to 2 further questionnaires.

2.7.2 Survey design

A survey was designed specifically for the study. It included questions regarding place and mechanism of injury, a falls risk assessment tool, FRAX assessment of bone health and fracture risk, EQ5D, comorbidity assessment and patient rated wrist evaluation (PRWE) (Appendix 5).

2.7.2.1 Patient-Reported Wrist Evaluation (PRWE)

PRWE is a patient rated outcome measure for wrist problems(174). It is a questionnaire which contains two scales of pain and function. There are five pain questions in which patients are asked to provide scores between 0 and 10 (0 = no pain, 10 = worst pain ever). There are 10 functional questions which contain four questions about everyday activities and six questions regarding specific activities. They are scored by the patient between 0 and 10 (0 = no difficulty, 10 = unable to do). An overall score out of 100 is calculated with a higher score indicating poorer functional outcome. The optimum PROM for distal radius outcome evaluation has yet to be established but the PRWE is relatively quick and easy to use and has been favoured in recent multi-centre trials investigating outcome after distal radius fracture. In this study, it is collected to provide a baseline for future follow up studies.

2.7.2.2 EuroQol EQ-5D-5L

The EuroQol EQ-5D-5L health questionnaire is a generic patient-reported outcome measurement of well-being(175). Responders are asked to select one of five options (eg; no problems, slight problems, moderate problems, severe problems, unable to) for five different domains (mobility, self-care, usual activities, pain/discomfort, anxiety/depression. They are also asked to rate their overall health on a scale of 0 (worst imaginable health) to 100 (best imaginable health). The EQ-5D-5L is not specific for acute distal radius fracture but can be a useful tool in long-term patient management and is frequently used in high quality studies.

2.7.2.3 FRAX®

FRAX[®] is a freely available online tool which calculates the 10-year probability of hip fracture or a major osteoporotic fracture(176). A major osteoporotic fracture is defined as a clinical spine, hip, forearm or humerus fracture. It was developed by the University of Sheffield and launched in 2008. It is used for around 2.8 million fracture calculations each year and has been shown to be robust, effective and is approved by NICE. It can be used with or without femoral neck bone mineral density measurements.

Age, gender and height and weight measurements are required for each patient. The following risk factors are assessed with a simple yes or no response:

Previous fracture

Parental hip fracture

Current smoker

Glucocorticoid use (for more than 3 months)

Rheumatoid arthritis

Secondary osteoporosis

Alcohol (3 or more units per day)

After inputting the data, the tool calculates percentage scores for 10 year risk of a hip fracture and 10 year risk of other major osteoporotic fracture. Both FRAX scores were calculated for all patients in the study.

2.7.2.4 Falls risk assessment

NICE recommend a falls risk assessment for people who have had one or more falls or are considered to be at risk of a fall(177). They also acknowledge that there is a lack of

evidence about which assessment tool is most predictive and useful. Most assessment scales were developed for the elderly in hospital settings or nursing homes(178). Rubenstein et al developed and validated a falls risk self assessment questionnaire following rigorous qualitative analysis (179). The work was commissioned by the US Centre for Disease Control and Prevention and required to meet the American and British Geriatric Society Guidelines. It was designed to be used in a falls risk selfassessment brochure that could be used as a screening tool and for public health education. It is now widely used in falls prevention initiatives in the US such as the national STEADI (Stopping elderly accidents, deaths and injuries) and the Ohio U Steady falls prevention programs (180, 181). It was chosen to be used in this study as it is appropriate to be used by community dwelling individuals and is designed to be a quick and simple checklist which is suitable for a postal survey (Appendix 6). Respondents circle yes or no regarding twelve statements about falls risk factors. Two of the statements ("I have fallen in the past year", "I have been advised to use a cane or walker to get around safely") score 2 points while all others score 1 point. A total of 4 points or more suggests the respondent is at risk of falling.

2.7.2.5 Analysis

Comparison of responders with non-responders is performed to ensure the results are applicable to the underlying population. Categorical results of quintiles 1 and 2 are compared with quintiles 3 to 5 using the Chi squared test. Linear regression analysis is performed to investigate the association between FRAX scores and falls risk with IMD rank. This is then repeated for each individual risk factor.

2.8 Statistical analysis

All statistical analysis was performed using R. A list detailing which packages were used is included as appendix 7.

2.8.1 Modelling

Many regression models are used throughout this thesis. Each model was chosen independently as the most appropriate model for the specific research question. In this section an overall review is provided of the models used, their interpretation and the reasons why they were chosen. In the subsequent chapters the results are linked back to the appropriate text in this section.

2.8.1.1 Linear regression

Linear regression is used to describe the relationship between a dependent variable(Y) and an independent or explanatory variable(X). An equation is given for the straight line that describes the change in the Y variable with an increase in the X variable. This is the equation of the regression line. It includes an intercept and the slope of the line. These values are calculated to minimize the sum of the squared vertical distances of the points from the line. This is known as a least-squares fit. The slope can also be known as the regression coefficient. A significant positive coefficient means that the Y variable increases with an increase in X. Multivariate regression investigates the relationship with multiple explanatory variables. This dependent variable must be continuous. Multivariate regression is used to analyse the relationship between osteoporosis and falls risk factors with deprivation, age and gender.

2.8.1.2 Generalised linear models

Generalised linear models are an extension of linear regression. They can be used with continuous dependent variables and the distribution does not have to be normal but should be a member of the exponential family.

Covariates are included which act through a linear predictor along with a link function. The linear predictors influence the mean of the dependent variable through the link function. This enables the dependent variables to have a non-linear relationship with the independent variables without forcing a linear relationship. GLMs were popularized in the 1980s by McCullagh and Nelder assisted by the development of new computer software(182).

2.8.1.3 Poisson regression

Poisson distribution occurs with events that can only be counted in whole numbers and cannot be negative, they occur independently and the average frequency of occurrence is known for a given time period. It is often used for relatively rare events such as an injury like a fracture. A poisson distribution curve will always show skew to the right but this will decrease with an increasing number of events. It was famously described in 1898 for the chance of a soldier in the Prussian army being killed by a horse kick(183).

Poisson regression is therefore appropriate for modelling count variables. It is used extensively in this thesis as it is suitable for modelling the relatively rare event of a distal radius fracture occurring during a given time period. It is assumed that the response variable demonstrates a poisson distribution. The link function is the log which means all parameter estimates must be exponentiated to aid interpretation.

2.8.1.4 Robust standard errors

Robust standard errors are calculated to control for inaccuracies due to assuming that variance equals the mean(184). These were calculated using the sandwich method.

This calculates standard errors for the coefficients without assuming that the variance of the residual errors is constant. Confidence intervals were calculated from these robust standard errors.

2.8.1.5 Offset

Poisson regression involves the modelling of counts of data. In clinical practice, it is more likely that we are wanting to compare rates of a disease or condition. We therefore need to know the population at risk and include it in the model. This is dealt with by adding an offset parameter to the model. In this thesis, this is frequently the specific group at risk by age, gender, ethnicity and deprivation quintile.

A simplified example of the code for the poisson regression deprivation models is provided below;

Number of fractures~ethnicity+deprivation quintile+age group+gender+offset (log (population at risk))

2.8.1.6 Logistic regression

Logistic regression is another generalised linear model. This is used for data with a binary response outcome. This is used in this thesis for investigating the factors associated with day case surgery. The dependent variables are binary ie surgery as a day case or surgery not as a day case.

2.8.1.7 Negative binomial regression

Negative binomial regression is a variation of poisson regression. It is based on a poisson gamma distribution mixture. In contrast to poisson regression the variance is larger than its mean. It can be used for over dispersed count data. In these cases, the

variance exceeds the mean. It contains an extra parameter to model the over dispersion. This type of model is used for the analysis of variation in surgical treatment rate between hospitals. Over dispersion was present when other models were initially tested and this could not be reduced with forward selection.

2.8.1.8 Generalised linear models interpretation

GLM output can be interpreted in several ways. Coefficients for poisson, negative binomial and logistic regression are given on the log scale. Therefore, the expected log count of the rate decreases or increases by the coefficient value for each one-unit change in coefficient. They can also be thought of as the percentage change in the dependent variable with a one unit change in the covariate ie if modelling rate against population gives a significant coefficient of -0.2 then rate will decrease by 20% with each one unit increase in population. When categorical factors are included in the model the results will show coefficents in relation to one of the factors, the reference group, ie if gender is used in the model the output would show a result for one gender only such as male = 0.2. If this is significant it means that men have a significantly higher effect on the dependent variable and an increase in the dependent variable of one unit would see a 20% higher increase for men than women. Exponentiating these results gives an incidence rate ratio of 1.22 meaning the effect of men is 1.22 times that of women. Incidence rate ratios are frequently used for the deprivation guintile regression results to compare incidence rate of patients in the most deprived quintile with those in the other quintiles.

2.8.1.9 Random effects

Data is frequently structured in a hierarchical fashion. This may occur due to many reasons but in medicine may be due to treatment by different clinicians or at different hospital trusts. Assumptions of independence may therefore not be accurate. An

effect is being caused by a factor who's specific value may not be of interest. This is the case when investigating the relationship between measures of treatment, such as wait to surgery, and other patient related factors. Initial testing revealed large differences dependent on hospital trust. Whilst this was interesting information I wish to investigate the other specific factors such as deprivation and ethnicity whilst also acknowledging the effect of the hospital trust in the model. A solution to this is to include the hospital trust in the model as a random effect term. The random effects are treated as variables with a mean of zero and an unknown variance.

A generalised linear mixed model is a further extension to a generalised linear model which also includes a random effects term. These were created for a logistic regression model and several linear regression models investigating the relationship between measures of quality of treatment with deprivation and other patient factors by adding the hospital trust as a random effect.

P values were estimated using the Kenward Rogers method. P values are not part of the initial output of the random effects mixed model as it is unclear about the degrees of freedom of the t statistic. The Kenward Rogers approximation is reported to obtain the most accurate significant tests from a multi-level model(185). The method involves adjusting both the F statistic and its degrees of freedom to calculate an approximate p value.

2.8.1.10 Overdispersion

Over dispersion occurs when the variance is larger than the mean. It means that there is more variation seen within the data than one would expect. Reasons for this may include that certain important explanatory variables have not been included in the model. Over dispersion is common as data from populations is often heterogeneous whereas common models make simple assumptions. It does indicate that the model is not appropriate. Over dispersion was initially seen in some of the deprivation models but after identifying and adding interactions to the model this was corrected. Over dispersion remained present with regression analysis of variation in surgical treatment

between hospital trusts despite addition of interactions. That data was being compared between hospital trusts and this was an important factor which needed to be included in the model. In this case a different method was chosen for analysis. Options include a negative binomial model or quasi likelihood estimation.

2.8.1.11 Under dispersion

Under dispersion is rare and occurs when less variation is seen between data than expected. This was not seen in any of the models tested or used in this thesis.

2.8.1.12 Generalised additive models

Generalised additive models are another extension of the GLMs. They were invented by Trevor Hastie and Robert Tibshirani in the 1980s(186). They are a technique where the effect of predictive variables is captured using smooth functions. These may be non-linear and can be used for non-parametric data. The relationships between the dependent variable and the predictors follow smooth patterns. They can demonstrate a non-linear pattern that a classic linear model would not identify. The smoothness is determined by smoothing functions which are estimated from the data. They are used to investigate the relationship between IMD rank and continuous variables of radiographic measurements.

The model was chosen using the Bayesian Information criterion model selection tool. This attempts to reduce the risk of error using a penalty term. This increases with the number of parameters. It therefore penalizes complicated models which have too many parameters for accurate estimates of that sized dataset. A smaller score indicates a better model.

2.8.1.13 Model selection

Models were selected using forward selection. An initial model was created using only one variable then other variables added one at a time. The deviance is checked and those that add the largest significant drop in deviance are added to the model. This process is repeated until the model cannot be improved or no variables can be added. Interactions were tested and added if they would reduce the deviance. In some cases, variables were noted to significantly interact with all other variables so separate models were created.

2.8.2 Multiple imputation

Multiple imputation is a strategy for dealing with missing values in datasets. A set of plausible values are given for each missing value. The datasets created can then be analysed with standard methods and the results combined. A classification and regression tree method of imputation was used in this thesis. Missing height and weight survey data was imputed using the already known data of gender, age, deprivation rank, number of comorbidities, falls risk factors and other osteoporosis risk factors

2.8.3 Prediction intervals

The relationship between hospital trust population and surgical treatment rate is presented as a scatter plot with 95% prediction intervals. These differ from confidence intervals which describe how well the mean is determined in 95% confidence intervals should contain the true value of the mean 95% of the time. The prediction interval states where the next observation, not the mean, is likely to fall. It describes the distribution of values rather than the uncertainty in establishing the mean.

2.8.4 Other tests

Deprivation quintiles were analysed in two ways. Deprivation population data was non-parametric. With large samples the kruskal-wallis test was used to identify any difference between the five groups. When sample size was smaller quintiles 1 and 2 were combined, and compared with the combined quintiles of 3 to 5 to produce a simpler deprived versus non-deprived analysis. Mann whitney U test was then used for non-parametric and t-test for parametric data. Chi squared test was used to compare categorical data such as mechanism or place of injury.

2.9 Permissions

Full permission was granted to access and use HES data for the study purposes from the Health & Social Care Information Centre. Local audit approval was granted to review patient demographics, radiographs and contact patients by survey. Permission was given to access census data and historical climate data for academic purposes by UK data service, census support and the Centre for Environmental Data Analysis respectively.

Chapter 3 Reliability and reproducibility of radiographic measurements

To allow significant findings to be gained from radiographs the methods used to measure the parameters must be reliable and reproducible. Results are presented investigating this for the commonly measured radiographic parameters of distal radius fracture displacement along with metacarpal cortical index and its relationship with bone mineral density. Intraclass correlation coefficient was used to measure inter-observer and intra-observer correlation between measurements (section 2.4.1.1 & 2.4.1.2). A generalised additive model was used to investigate the relationship between MCI and BMD, T-score, age and fracture displacement (section 2.4.2 & 2.8.1.12).

3.1 Extra-articular radiographic measurements

A random sample of 367 patients was computer generated by an independent observer. 341 patients had postero-anterior radiographs suitable for measurement of radial height and radial inclination. Measurement of dorsal tilt was possible on lateral radiographs from 338 patients.

Inter-observer correlation was excellent (> 0.8) for radial height, radial inclination and dorsal tilt between the two observers (Table 1). Intra-observer correlation was also excellent when repeat measurements of 10% of the sample were repeated four weeks after the original measurements.

Table 1 Measurement of inter-observer correlation between the two observers with95% confidence intervals for common extra-articular radiographic parameters

	ICC	Cl - high	CI - low	N
Radial height	0.9	0.88	0.92	341
Radial inclination	0.86	0.83	0.88	341
Dorsal Tilt	0.93	0.92	0.95	338

Table 2 Intra-observer correlation between the two observers with a 10% retest sample 4 weeks after initial measurements for common extra-articular radiographic parameters

	ICC	CI - high	CI - low	N
Observer 1				
Radial height	0.95	0.97	0.91	37
Radial inclination	0.87	0.93	0.77	37
Dorsal Tilt	0.99	0.99	0.98	37
Observer 2				
Radial height	0.98	0.99	0.96	37
Radial inclination	0.88	0.94	0.78	37
Dorsal Tilt	0.99	0.99	0.97	37

3.2 Intra-articular radiographic measurements

200 patients with intra-articular fractures were identified, 193 had satisfactory lateral radiographs suitable for measurement of intra-articular step and gap. Radiographs were deemed unsuitable for inclusion if incorrect positioning of the wrist had resulted in rotational distortion preventing appropriate visualisation of the longitudinal axis and true volar and dorsal lips of the distal radius required for accurate measurement.

Inter-observer correlation was poor (< 0.4) for articular step and gap measurements of the whole sample (Table 3).

Measurements were then repeated for fractures with displacement ≥ 2 mm. This value was chosen as this displacement is likely to cause functional impairment and would be most surgeons threshold for surgical intervention. Correlation was also poor in this group. (Table 4).

Table 3 Measurement of inter-observer correlation between the two observers with95% confidence intervals (whole sample) for intra-articular radiographic parameters

	ICC	CI - high	CI - low	Ν
Intra-articular step	0.030	-0.111	0.17	193
Intra-articular gap	0.37	0.24	0.48	193

Table 4 Inter-observer correlation between the two observers with 95% confidence intervals (fractures with displacement \geq 2mm) for intra-articular radiographic parameters

	ICC	CI - high	CI - low	N
Intra-articular step	-0.4174	-0.595	-0.20	68
Intra-articular gap	-0.049	-0.314	0.22	52

Intra-observer correlation (Table 5) was poor for both parameters assessed by observer 2. Correlation was good (0.6 - 0.74) for assessment of gap but poor for step measurement by observer 1.

Table 5 Intra-observer correlation between the two observers with a 10% re-test sample four weeks after initial measurements for intra-articular radiographic parameters

	ICC	CI - high	CI - low	N
Observer 1				
Intra-articular gap	0.69	0.4	0.85	23
Intra-articular step	0.18	-0.23	0.54	23
Observer 2				
Intra-articular gap	0.04	-0.36	0.44	23
Intra-articular step	0.12	-0.29	0.50	23

3.3 Metacarpal Cortical Index (MCI)

3.3.1 MCI measurement

From the same sample used for extra-articular measurements 335 AP radiographs demonstrated the 2nd metacarpal allowing measurement of MCI. Inter-observer correlation for MCI measurements was high between the two observers with an ICC value of 0.85 (95% confidence interval 0.82-0.88).

In the 37 patients that were retested by both reviewers 4 weeks after the initial data collection to determine intra-observer reliability, ICC values also remained high at 0.93 and 0.91 for each reviewer respectively (See Table 6).

Table 6: Intra-observer Correlation with 95% Confidence Intervals (CI) for MCI measurement

	ICC	95% CI high	95% CI low	N
MCI (Reviewer 1 retest)	0.93	0.96	0.86	37
MCI (Reviewer 2 retest)	0.91	0.95	0.84	37

3.3.2 MCI and age and fracture displacement

Data was non-parametric and non-linear so a generalised additive model (GAM) was used to investigate the relationship between MCI and age, gender and dorsal displacement (section 2.8.1.12). This model was chosen after comparison with other non-linear regression techniques using cross validation and Bayesian information criterion.

MCI inversely correlated with patient age (p<0.0001, r^2 =0.56)(Figure 4). The decline in MCI was most noticeable in patients over 50 years old and steeper in women (women; p<0.0001, r^2 =0.62, men; p<0.0001, r^2 =0.32) (Figure 5). In patients over 50 MCI decreased in a steep linear fashion with increasing age (p<0.0001, r^2 =0.53) (Figure 6).





Metacarpal Index per age of patient

Figure 5: Relationship between MCI and age by gender in patients with distal radius fracture



Metacarpal Index per age for men and women

Figure 6: Association between age and decreasing MCI in patients over 50 years of age with distal radius fracture



Metacarpal Index per age of patient - patients aged over 50

No significant relationship was seen between MCI and the degree of dorsal angulation of distal radius fractures on the lateral radiograph (p=0.17, r²=0.02). Only dorsally angulated fractures were included in this analysis (n=175). When the model was repeated for patients aged over 50 years only (n=124) who are likely to have suffered fractures due to fragility no significant relationship was again found (p=0.06, r=0.07)(Figure 7).

Figure 7: MCI and dorsal angulation of distal radius fractures in patients aged 50 years and over



Relationship between MCI and volar tilt in patients over 50

3.3.3 MCI and bone fragility

From a separate sample of consecutive patients who had undergone DXA scanning and imaging of the 2nd metacarpal within 90 days of each other (n=104) GAM modelling revealed a significant correlation between MCI and BMD (p<0.0001, r^2 =0.22) (Figure 8) and MCI and T-score (p<0.0001, r^2 =0.22) (Figure 9). A sharp decline in BMD and T-score was seen in patients with an MCI of <0.5.

Figure 8: Relationship between MCI and BMD in consecutive patients who had undergone radiographs demonstrating the 2nd metacarpal and a DXA scan



Relationship between Metacarpal Index and Bone Mineral Density

Figure 9: Relationship between MCI and T-score in consecutive patients who had undergone radiographs demonstrating the 2nd metacarpal and a DXA scan



Relationship between Metacarpal Index and T score

3.4 Summary of radiographic measurement results

These results confirm that the extra-articular measurements of distal radius fracture displacement are reliable and reproducible (section 9.2.1). Dorsal tilt is used frequently in the thesis for further radiographic analysis. Dorsal tilt was chosen as it is believed to be the most important parameter influencing final functional outcome and

is also likely to significantly change with poor bone health and osteoporosis. Intraarticular step is used in only a small number of tests due to poor intra and interobserver correlation (section 9.2.2). Step is chosen over gap as it is felt to be more important for functional outcome by clinicians.

MCI is shown to be reliable and reproducible and to correlate well with bone health (section 9.3). MCI provides a rapid, low cost assessment of bone frailty which can be conducted in most patients who present with a distal radius fracture. It is used throughout the thesis as a measure of bone fragility.

Chapter 4 Epidemiology of distal radius fracture in Leicestershire

A 10 year review was carried out of all patients who attended the UHL Emergency Department with a distal radius fracture (section 2.1). Poisson regression analysis was carried out to investigate gender and age specific trends (section 2.8.1.3). Prediction of future fracture prevalence is made based on projected population profile changes. This will assist with healthcare planning and delivery by allowing estimation of expected number of injuries. Understanding the total expected number of distal radius fractures along with the influence of deprivation on fracture rate allows accurate predictions to be made for at risk deprived groups.

4.1 Population changes

From 2007 to 2016 there was a 9.8% increase in total population in the region. This has primarily resulted from an 18.2% increase in the over 50-age group, whilst only a 3.5% increase was seen in the 18-50 population. The percentage of the total adult population made up of the over 50 age group has also increased, confirming an aging population in this geographical area. Review of the UK population profile for the 10 year period studied in this section revealed close correlation with that of our geographical region (Table 7). The gender make-up of our study population and that of the UK as a whole remained stable at 51% female for all years studied.

Table 7 Comparative adult population profiles of Leicestershire and the whole United Kingdom

	Leicestershire		υκ	
Year	18-50 (%)	>50 (%)	18-50 (%)	>50 (%)
2007	57.7	42.3	56.8	43.2
2016	54.5	45.5	53.5	46.5

4.2 Trends in distal radius fracture incidence

4.2.1 Overall Fracture Incidence

Between 2007 and 2016, 10,259 adult patients presented to the Emergency Department with fractures of the distal radius. The mean annual number of fractures during the 10 year period was 1,026 (range 899 – 1258). Mean incidence was calculated from the total number of fractures and total population at risk during the study period. Poisson regression modelling revealed no overall change in the number of fractures or fracture incidence in any of the subgroups studied when compared to 2007 as the reference year (Table 8). There were significant increases in fracture incidence in specific years (Figure 10). In 2009 and 2010, fracture incidence increased overall (p<0.0001 both years), for men (p=0.0141 and p=0.0046) and women aged 50 years and over(p= 0.0055 and p<0.0001). There were significant reductions in incidence for women under 50 in 2011 (p=0.0116), 2012 (p=0.0118), 2015 (p=0.0011) and 2016 (p=0.0026), women 50 and over in 2016 (p=0.0153), men under 50 in 2016 (p=0.0014) and overall in 2016 (p<0.0001).

	All Patient	S	Men		Women	
	No. of	Incidence	No. of	Incidence	No. of	Incidence
Vear	Fractures	(/100,000 person	Fractures	(/100,000 person	Fractures	(/100,000 person
i cui		years)		years)		years)
2007	973	126.8	306	81.6	667	170.1
2008	944	121.8	307	81.0	637	160.9
2009	1180	150.9	371	97.0	809	202.6
2010	1258	159.2	370	95.6	888	220.1
2011	1010	126.5	327	83.7	683	167.5
2012	1032	128.3	337	85.6	695	169.2
2013	979	120.7	319	80.2	660	159.7
2014	1005	122.5	342	84.9	663	158.8
2015	979	117.9	338	82.8	641	151.8
2016	899	106.8	282	67.9	617	144.5

Table 8: Total number of distal radius fractures and incidence (expressed per 100,000 person/years) in Leicestershire from 2007 to 2016

Figure 10: Annual distal radius fracture incidence in Leicestershire from 2007 to 2016 by gender and age group



Distal radius fracture incidence per year

The overall male: female ratio of fractures was 32%: 68% with no significant variation over the study period. Mean age at fracture was 43.8 (range 18-98) in male patients and 62.9 (range 18-105) in female patients. Within the young (<50 years) population, fractures were more common in men, but older (50 years and over) females were by far the commonest group overall to present with a fracture (Table 9). Figure 11 demonstrates the age and gender distribution.

Table 9 Mean (10-year) distal radius fracture incidence rates in Leicestershire by

	Male	Female	All
<50 age group	94.6	72.0	83.3
>50 age group	69.5	289.6	184.7
All	83.9	170.2	127.9

gender and age group (expressed per 100,000 person/years)

Figure 11 Total number of distal radius fracture in Leicestershire stratified cumulatively by gender and age from 2007 to 2016



Histogram of total distal radius fractures by age and gender

4.2.2 Age and Gender Specific Incidence

Age and gender specific incidence rates of distal radius fracture were calculated and are presented in Figure 12. In women the incidence of fracture increased steadily from the fifth decade to reach a peak of 515/100,000 person years in the 80-89 age group before declining. In male patients, incidence remained low until the eighth decade from which point it rose to a peak of 124/100,000 person years in the over 90 age group.

Figure 12 Gender and age specific distal radius fracture incidence rates in Leicestershire from 2007 to 2016



Distal radius fracture incidence per age group

4.2.3 Seasonal Variation of Fracture Incidence

Poisson regression analysis of the cumulative data demonstrated a statistically significant increase (p<0.0001) in fracture incidence in winter for all groups except for young males (Figure 13), where a statistically significant fall (p<0.0001) was recorded. Across the ten-year period, 28% of fractures occurred in winter compared with 23%, 25% and 24% in spring, summer and autumn respectively (section 2.8.1.3).

Figure 13 Seasonal variation of distal radius fracture incidence by gender and age group in Leicestershire from 2007 to 2016



Distal radius fracture incidence per season

4.2.4 Weekly Variation in Fracture Incidence

The data reveals a notable pattern of rate of distal radius fracture in association with day of the week. In the under 50 population there is a higher incidence of fracture on or immediately after the weekend, with 52% of total fractures occurring on Saturdays, Sundays and Mondays. This was statistically significant for both men (p<0.0001 for all 3 days) and women (p=0.0075 on Mondays, and p<0.0001 for Saturday and Sunday) aged under 50 years when compared to Tuesdays using poisson regression (Figure 14). Tuesday was chosen for comparison as it is in the working week and unlikely to be influenced by weekend activities. There was no significant difference in incidence amongst the over 50 population across the week, but the magnitude of change in the young population was sufficient to affect a significantly increased incidence of injury at the weekend overall, accounting for 45% of total fractures (p=0.0022 on Mondays, p=0.0129 on Saturdays and p<0.0001 and Sundays).

The increased incidence of fracture amongst the young population over the weekend affects a reduction in the mean age of patients presenting with distal radius fracture at this period of a week (p<0.0001) (Figure 15).

Figure 14 Weekly variation of distal radius fracture incidence by gender and age group in Leicestershire from 2007 to 2016



Distal radius fracture incidence per day of the week
Figure 15 Mean age of distal radius fracture patients each weekday in Leicestershire from 2007 to 2016



Mean age of distal radius fracture patients per day of the week

4.3 Fracture Projection Analysis

Figure 16 shows the expected trend for the projected number of distal radius fracture in the next 20 years in the UK. Our data suggest there will be 11.2% and 23.1% increases in total fractures in the next ten and twenty years respectively. This results primarily from more fractures in the over 50 male and female age groups as a consequence of the predicted changing population structure. Figure 16 Projected number of distal radius fracture per year in the United Kingdom by gender and age group from 2016 to 2036



Projected number of distal radius fractures per year

4.4 Summary of the epidemiology of distal radius fracture in Leicestershire

Despite an aging population, the incidence of distal radius fracture does not appear to be changing (section 9.5). Weekly and seasonal trends are apparent which may assist with resource planning and allocation. A significant rise in the number of distal radius fracture is expected in the UK in the next two decades which will require suitable planning of resource allocation to ensure appropriate treatment of patients. The population profile in the Leicestershire study region closely matches that of the UK as a whole so these results should be generalisable to the rest of the UK. Further understanding of how deprivation affects fracture rates allows identification of at risk groups and enables resources to be targeted appropriately.

Chapter 5 Effect of deprivation on distal radius fracture rate in Leicestershire

The aim of this chapter was to establish if distal radius fracture rate is associated with deprivation in Leicestershire and nationally. Age specific rates are calculated by deprivation quintile and compared with expected rates. Poisson regression modelling is performed investigating the effect of age, gender, ethnicity and deprivation on fracture rate (section 2.6.3 & 2.8.1.3). Analysis of the relationship between deprivation and radiographic measures of fracture displacement and bone fragility is performed using GAM models to try and establish reasons for differences in fracture rate (section 2.4 & 2.8.1.12).

5.1 Total numbers of fractures

4,278 consecutive patients with a Leicestershire postcode attended the emergency department with a distal radius fracture during the four year period and were included in the analysis.

There was a clear trend of distal radius fracture rate increasing with higher deprivation quintile for male and female patients (Table 10).

-		<u>Men</u>		<u>Women</u>		<u>.II</u>
Deprivation	<u>#/yr</u>	<u>Rate</u>	<u>#/yr</u>	<u>Rate</u>	<u>#/yr</u>	<u>Rate</u>
Q1 (most deprived)	73.5	143.57	128.75	242.38	202.25	193.88
Q2	73.25	106.15	135.75	190.67	209	149.07
Q3	49	76.88	121.75	184.36	170.75	131.58
Q4	56.75	59.79	161	161.74	217.75	111.98
Q5 (least deprived)	66.25	59.49	205.75	174.77	272	118.73

Table 10 Mean number and rate of distal radius fracture in Leicestershire (expressed per 100,000 person/years) by gender and deprivation quintile from 2007 to 2010

5.1.1 Observed and expected fracture rate

Higher than expected numbers of fractures were seen in the most deprived two quintiles and less than expected in the two least deprived quintiles. Similar numbers of observed and expected fractures were seen in quintile three, the median quintile. Table 11Observed number of distal radius fracture seen in Leicestershire andexpected number of distal radius fracture in the region for each deprivation quintilefrom 2007 to 2010

		% of	Fractures	Fractures
Quintile	Population	population	expected	observed
1	104315	13.1	140	202
2	140203	17.6	188	209
3	129772	16.3	174	171
4	194460	24.4	261	218
5	229094	28.7	307	272

Expected number of fractures is calculated from the proportion of the population in each quintile

Figure 17 Comparison of the expected and observed number of distal radius fracture seen in Leicestershire in each deprivation quintile from 2007 to 2010



Expected and observed distal radius fractures per year by quintile

5.1.2 Deprivation and age of distal radius fracture patients

Figure 18 Mean age of distal radius fracture patients in Leicestershire in each quintile with 95% confidence intervals from 2007 to 2010



Relationship between age and deprivation quintile

Mean age of patients in the most deprived two quintiles was 54.4 years. This was significantly lower than the mean age of 60.1 years in the least deprived three quintiles (p<0.001).

5.2 Relationship between deprivation and radiographic parameters

Analysis of the relationship between deprivation, measured by IMD rank, and dorsal tilt, intra-articular step and metacarpal cortical index was performed using GAM models (section 2.8.1.12).

5.2.1 Dorsal tilt

Figure 19 Relationship between dorsal tilt and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model



Relationship between tilt and IMD rank

865 patients of the group who underwent radiographic measurement had lateral radiographs which allowed measurement of dorsal tilt. No relationship was demonstrated between tilt and deprivation measured by IMD rank (p=0.741).

5.2.2 Intra-articular step

219 patients had sustained a fracture with a displaced intra-articular step. There was no association with IMD rank (p=0.102).

Figure 20 Relationship between intra-articular step and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model



Relationship between intra-articular step and IMD rank

5.2.3 Metacarpal Cortical Index

There was no significant relationship seen between MCI and IMD rank in 888 patients (p=0.097). Further analysis by quintile revealed that MCI was higher in patients from

quintiles 1 and 2 (mean = 0.50) than those in the more deprived quintiles (mean =0.49) but this did not reach significance (p=0.07)

Figure 21 Relationship between MCI and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model



Relationship between MCI and IMD rank

Figure 22 Mean MCI of distal radius fracture patients in Leicestershire from 2007 to 2010 in each deprivation quintile with 95% confidence intervals



Relationship between metacarpal cortical index and deprivation quintile

5.3 Patients with distal radius fracture aged 50 years and over

2,803 patients from Leicestershire aged 50 years and over sustained distal radius fractures during the four year period studied. As with the overall results a clear trend was seen of the distal radius fracture rate increasing with higher deprivation quintile. More than expected numbers of fractures were seen in the most deprived two quintiles and less than expected in the two least deprived quintiles.

Table 12 Mean number of fractures per year and rate per deprivation quintile (expressed per 100,000 person/years) in patients in Leicestershire aged 50 years and over from 2007 to 2010

Deprivation	<u>#/yr</u>	<u>Rate</u>
Q1	115.5	335.06
Q2	124	236.28
Q3	117	206.27
Q4	148	160.83
Q5	197.75	178.25

Table 13 Observed number of distal radius fracture seen in Leicestershire and expected number of distal radius fracture in the region for each deprivation quintile from 2007 to 2010 in patients aged 50 years and over

Quintilo	Population	% of	Fractures	Fractures
Quintile	ropulation	population	expected	observed
1	34471	9.9	69.84	115.5
2	52480	15.1	106.32	124
3	56723	16.4	114.92	117
4	92021	26.5	186.43	148
5	110938	32.0	224.75	197.75

Figure 23 Comparison of the expected and observed number of fractures seen in Leicestershire in each deprivation quintile from 2007 to 2010 in patients aged 50 years and over



Expected and observed distal radius fractures per year by quintile in patients aged 50 years and over

Figure 24 Mean age of distal radius fracture patients in Leicestershire from 2007 to 2010 aged 50 years and over in each deprivation quintile with 95% confidence intervals



Relationship between age and deprivation quintile in patients aged 50 and above

A significant difference in mean age was seen across the deprivation quintiles (p=0.001). Mean age of patients in the most deprived to quintiles was 69.6. This was significantly lower than the mean age of 71.2 in the least deprived three quintiles (p<0.001).

5.3.1 Relationship between deprivation and radiographic parameters in patients aged 50 years and over

5.3.1.1 Dorsal tilt

Figure 25 Relationship between dorsal tilt and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model in patients aged 50 years and over



Relationship between tilt and IMD rank in patients aged 50 years and over

537 patients in this age group had measurement of dorsal tilt performed. No relationship was seen between dorsal tilt and IMD rank (p=0.919).

5.3.1.2 Intra-articular step

Figure 26 Relationship between intra-articular step and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model in patients aged 50 years and over



Relationship between intra-articular step and IMD rank in patients aged 50 years and over

133 patients had sustained a fracture with a displaced intra-articular step. No association was seen with IMD rank (p=0.96).

5.3.1.3 Metacarpal Cortical Index

Figure 27 Relationship between MCI and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model aged 50 years and over



Relationship between MCI and IMD rank in patients aged 50 years and over

No relationship was seen between MCI and IMD rank in the 551 patients in which it was measured (p=0.248).

5.4 Patients with distal radius fracture aged between 18 and 49 years

1,475 Leicestershire residents suffered a distal radius fracture in the four-year study period who were aged between 18 and 49 years of age inclusive. Again, a clear trend was seen of an increased fracture rate in more deprived quintiles. More fractures than expected were seen in quintiles one and two and less fractures than expected were seen in quintiles four and five.

Table 14 Mean number of fractures per year and rate per deprivation quintile (expressed per 100,000 person/years) in adult patients in Leicestershire aged under 50 from 2007 to 2010

Deprivation	<u>#/yr</u>	<u>Rate</u>
Q1	115.5	335.06
Q2	124	236.28
Q3	117	206.27
Q4	148	160.83
Q5	197.75	178.25

Figure 28 Comparison of the expected and observed number of fractures seen in Leicestershire in each deprivation quintile from 2007 to 2010 in adult patients aged under 50 years



Expected and observed distal radius fractures per year by quintile in adult patients aged under 50 years

Table 15 Observed number of distal radius fracture seen in Leicestershire and expected number of distal radius fracture in the region for each deprivation quintile from 2007 to 2010 in adult patients aged under 50 years

Quintilo	Deputation	% of	Fractures	Fractures
Quintile	Population	population	expected	observed
1	69844	15.5	57.20	86.75
2	87723	19.4	71.84	85
3	73048	16.2	59.82	53.75
4	102439	22.7	83.89	69.75
5	118156	26.2	96.76	74.25

Figure 29 Mean age of distal radius fracture patients in Leicestershire from 2007 to 2010 in adult patients aged under 50 years in each deprivation quintile with 95% confidence intervals



Relationship between age and deprivation quintile in patients aged below 50 years

No significant relationship was seen between age and deprivation quintile across the groups (0.784) or when quintiles 1 and 2 were compared with quintiles 3-5 (0.274). Mean age of patients in the two most deprived quintiles was 33.3 years compared to 33.9 in the least three deprived quintiles.

5.4.1 Relationship between deprivation and radiographic parameters in adult patients aged between 18 and 49 years

5.4.1.1 Dorsal tilt

Figure 30 Relationship between dorsal tilt and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model in adult patients aged below 50 years



Relationship between tilt and IMD rank in adult patients aged under 50 years

No relationship was seen between dorsal tilt and IMD rank in the 392 patients studied (p=0.862).

5.4.1.2 Intra-articular step

Figure 31 Relationship between intra-articular step and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model in adult patients aged below 50 years



Relationship between intra-articular step and IMD rank in adult patients aged under 50 years

86 patients in this age group had measurement of a displaced intra-articular step performed. A relationship between increasing step size and increased deprivation came close to statistical significance (p=0.052).

5.4.1.3 Metacarpal Cortical Index

Figure 32 Relationship between MCI and IMD rank of distal radius fracture patients in Leicestershire from 2007 to 2010 investigated using a generalised additive model in adult patients aged below 50 years



Relationship between MCI and IMD rank in adult patients aged under 50 years

No significant relationship was seen between MCI and IMD rank in 337 patients (p=0.353).

5.5 Regression model results

Further analysis was performed investigating the effect of age, gender, ethnicity and deprivation on fracture rate. Poisson regression modelling showed a significant difference between ethnicity and all other factors (age<0.001, gender<0.001, deprivation<0.001) indicating important differences between the ethnic groups.

Further regression was therefore carried out separately for the ethnic groups (section 2.8.1.3).

Regression modelling of white patients only (n=3481) showed fracture rate was associated with increased deprivation, age 50 years and over (p<0.001) and male gender (p=0.010) (Table 16). Model testing revealed a significant interaction between deprivation quintile and gender (p<0.001) This interaction was therefore added to the final model and separate results produced for men and women per quintile. Incidence rate ratio of the least deprived quintile compared to the most deprived was 0.33 (95% CI : 0.30 - 0.37) for men and 0.47 (95% CI : 0.44 - 0.49) for women.

Table 16 Regression results for white patients demonstrating the relationship between fracture rate and deprivation quintile, gender and age

Factor	IRR	95% Cl lower limit	95% Cl upper limit	P value
Quintile 2 - women	0.63	0.59	0.68	p<0.001
Quintile 3 - women	0.52	0.48	0.55	p<0.001
Quintile 4 - women	0.43	0.41	0.46	p<0.001
Quintile 5 - women	0.47	0.44	0.49	p<0.001
Quintile 2 - men	0.60	0.53	0.68	0.716
Quintile 3 - men	0.40	0.35	0.45	0.045
Quintile 4 - men	0.32	0.28	0.36	0.01
Quintile 5 - men	0.33	0.30	0.37	0.004
Age 50+	4.18	3.98	4.39	p<0.001
Male	1.28	1.16	1.40	0.01

IRR of fracture rate for deprivation quintiles are compared with quintile 1 (most deprived), Age 50 years and over is compared with patients below 50 years and male gender with female gender Figure 33 distal radius fracture incidence rate ratio of white patients in each deprivation quintile compared to the most deprived quintile in Leicestershire from 2007 to 2010



The regression model demonstrated no association with fracture and deprivation for patients of other ethnicities (n=797). There was no interaction between gender and deprivation (p=0.743) suggesting similar results for men and women. Fracture risk was higher in males (p<0.001) and patients aged 50 years and over (p<0.001) (Table 17).

Table 17 Regression results for patients of other ethnicity demonstrating therelationship between fracture rate and deprivation quintile, gender and age

IRR of fracture rate for deprivation quintiles are compared with quintile 1 (most deprived), Age 50 years and over is compared with patients below 50 years and male gender with female gender

Factor	IRR	95% Cl lower limit	95% Cl upper limit	P value
Quintile 2	1.16	1.05	1.28	0.125
Quintile 3	1.24	1.11	1.39	0.056
Quintile 4	1.22	1.08	1.38	0.097
Quintile 5	1.24	1.09	1.40	0.088
Age 50+	5.09	4.54	5.72	p<0.001
Male	2.62	2.36	2.91	p<0.001

Figure 34 distal radius fracture incidence rate ratio of patients of other ethnicity in each deprivation quintile compared to the most deprived quintile in Leicestershire from 2007 to 2010



5.6 Ethnicity, age and radiographic findings

Following the significant findings regarding the importance of age group, gender and ethnicity in the regression analysis further exploratory sub analysis was performed on the radiographic measurement data by gender, age and ethnicity using GAM. Some sub analysis groups, particularly other ethnicity, were small. This leads to the irregular plots seen. Further analysis was therefore carried out comparing those in the two most deprived quintiles with the three least deprived quintiles.

5.6.1 Patients aged 50 years and over

5.6.1.1 Dorsal tilt

No relationship between dorsal tilt and IMD rank was seen in any of the gender/ethnicity subgroups in older patients.

Figure 35 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and ethnicity in distal radius fracture patients in Leicestershire from 2007 to 2010 aged 50 years and over



Relationship between tilt and IMD rank

Table 18 P values obtained for each gender ethnicity subgroup from the generalizedadditive model investigating relationship between IMD rank and dorsal tilt

Group	Ν	P value
Female other	50	0.543
Female white	375	0.402
Male other	25	0.732
Male white	94	0.916

Table 19 Results of the comparison of mean dorsal tilt between the two most deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender ethnicity subgroup

Tilt	Q1-2	Ν	Q3-5	Ν	Р
	mean		mean		
female other	24.36	31	20.52	19	0.377
female white	24.10	115	24.65	260	0.747
male other	22.09	16	21.36	9	0.916
male white	22.98	26	23.17	68	0.962

5.6.1.2 MCI

Significant trends were seen in females of other ethnicity and white males. Females of other ethnicity had higher MCI in more deprived areas. In deprived areas, their MCI was higher than white females.

White males had lower MCI in more deprived areas. In the most deprived two quintiles MCI of white males was significantly less than the MCI of males of other ethnicity.

Figure 36 Generalised additive model analysis comparing MCI and IMD rank by gender and ethnicity in distal radius fracture patients in Leicestershire from 2007 to 2010 aged 50 years and over



Relationship between MCI and IMD rank

Table 20 P values obtained for each gender ethnicity subgroup from the generalized additive model investigating relationship between IMD rank and MCI

Group	Ν	P value
Female other	50	0.004
Female white	382	0.951
Male other	25	0.839
Male white	94	0.034

Table 21 Results of the comparison of mean MCI between the two most deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender ethnicity subgroup

MCI	Q1-2	Ν	Q3-5	Ν	Р
	mean		mean		
female other	0.52	31	0.46	19	0.008
female white	0.44	117	0.44	265	0.734
male other	0.55	16	0.53	9	0.753
male white	0.44	26	0.50	68	0.001

Table 22 Results of the comparison of mean MCI between ethnicity groups of the same gender in the two most deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5)

MCI	Q1-2	N	Q3-5	N
	mean		mean	
female other	0.52	31	0.46	19
female white	0.44	117	0.44	265
Р	0.0001	Р	0.399	
male other	0.55	16	0.53	9
male white	0.44	26	0.50	68
Р	0.0001	Р	0.292	

5.6.2 Patients aged between 18 and 49 years

5.6.2.1 Dorsal tilt

In younger patients, no relationship was seen between dorsal tilt and IMD rank in any of the subgroups. Numbers of patients from other ethnicities was small.

Figure 37 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and ethnicity in adult distal radius fracture patients in Leicestershire from 2007 to 2010 aged under 50 years



Relationship between tilt and IMD rank

Table 23 P values obtained for each gender ethnicity subgroup from the generalized additive model investigating relationship between IMD rank and dorsal tilt in adult patients aged under 50 years

Group	Ν	P value
Female other	33	0.771
Female white	119	0.204
Male other	45	0.381
Male white	136	0.262

Table 24 Results of the comparison of mean dorsal tilt between the two most deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender ethnicity subgroup in adult patients aged under 50 years

Tilt	Q1-2	Ν	Q3-5	N	Р
	mean		mean		
female other	19.75	21	21.27	12	0.800
female white	23.21	45	25.39	74	0.524
male other	21.35	21	21.89	24	0.920
male white	21.76	57	24.66	79	0.336

5.6.2.2 MCI

No significant relationship was identified between MCI and IMD rank for any of the subgroups of younger patients.
Figure 38 Generalised additive model analysis comparing MCI and IMD rank by gender and ethnicity in adult distal radius fracture patients in Leicestershire from 2007 to 2010 aged under 50 years

Relationship between MCI and IMD rank

Table 25 P values obtained for each gender ethnicity subgroup from the generalized additive model investigating relationship between IMD rank and MCI in adult patients aged under 50 years

Group	Ν	P value
Female other	33	0.688
Female white	121	0.453
Male other	47	0.572
Male white	136	0.081

Table 26 Results of the comparison of mean MCI between the two most deprived quintiles (Q1-2) and the three least deprived quintiles (Q3-5) for each gender ethnicity subgroup in adult patients aged under 50 years

MCI	Q1-2		Q3-5	Ν	Р
	mean		mean		
female other	0.57	21	0.59	12	0.469
female white	0.57	45	0.57	76	0.986
male other	0.54	22	0.53	25	0.936
male white	0.54	57	0.54	79	0.951

5.6.3 Comparison between ethnic groups

Plots of the generalised additive models are provided to allow visual comparison of the differences between the subgroups of patients of the same ethnicity.

5.6.4.1 Dorsal tilt

Figure 39 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and age group in white distal radius fracture patients in Leicestershire from 2007 to 2010



Relationship between tilt and IMD rank

5.6.4.2 MCI

Figure 40 Generalised additive model analysis comparing MCI and IMD rank by gender and age group in white distal radius fracture patients in Leicestershire from 2007 to 2010



Relationship between MCI and IMD rank

5.6.5 Patients of other ethnicity

5.6.5.1 Dorsal tilt

Figure 41 Generalised additive model analysis comparing dorsal tilt and IMD rank by gender and age group in distal radius fracture patients of other ethnicity in Leicestershire from 2007 to 2010



Relationship between tilt and IMD rank

Figure 42 Generalised additive model analysis comparing MCI and IMD rank by gender and age group in distal radius fracture patients of other ethnicity in Leicestershire from 2007 to 2010



Relationship between MCI and IMD rank

5.7 Effect of deprivation and ethnicity on national distal radius fracture surgery rate

Full demographic data was available on 40,386 patients aged 50 years and over and 18,929 patients aged 18 to 49 years who underwent surgery for distal radius fracture over 4 years. Poisson regression models were constructed using total population at risk, calculated accurately for age, ethnicity, gender and quintile, as an offset.

Initial overdispersion of the models was corrected using forward selection and by creating separate models for under and over 50 years of age. This was chosen after age was seen to significantly interact with all other factors (section 2.8.1.13).

5.7.1 Patients aged under 50 years

Ethnicity was seen to interact with gender so this interaction was added to the model. Fixation rate significantly fell in the less deprived quintiles (IRR 0.75, CI 0.73-0.77, comparing the least deprived quintile to the most deprived). Rate was lower for all other ethnicities compared to white patients (Asian IRR 0.18, CI 0.17-0.20, Black IRR 0.10, CI 0.09-0.12, Other IRR 0.45, CI 0.41-0.49). Males of all ethnicity were more likely to undergo fixation (Appendix 8).

5.7.2 Patients aged 50 years and over

Significant interaction was seen between quintile and ethnicity, quintile and gender and ethnicity and gender. These interaction terms were added to the model.

Fixation rate significantly fell in the less deprived quintiles only in white patients. The influence of deprivation was greater in men (white women IRR 0.90, CI 0.88-0.92, white men IRR 0.72, CI 0.68-0.76, comparing the least deprived quintile to the most

deprived for each gender). White patients were more likely to undergo surgery. Fixation rate was significantly lower in white men compared to women (IRR 0.23, CI 0.23-0.24)

Figure 43 Incidence rate ratio of white patients who underwent surgery for distal radius fracture in each deprivation quintile compared to the most deprived quintile in the UK from 2009 to 2013



5.8 Summary of the effect of deprivation on distal radius fracture rate in Leicestershire

Deprivation was strongly associated with increased distal radius fracture rate. More fractures were seen than expected in the two most deprived quintiles for both age groups studied (section 9.6.1). Mean age of deprived distal radius fracture patients was significantly lower. When deprivation, ethnicity and age are considered being male is a risk factor for distal radius fracture. Modelling showed important differences between ethnic groups. Deprivation was an independent risk factor for distal radius fracture only in white patients. Women of other ethnicity aged 50 years and over in the most deprived quintiles had a higher MCI than those in the less deprived quintiles and when also compared to white women from the same quintiles (section 9.6.2). The reverse was true for white men in the most deprived quintiles who had a lower MCI than white men in the less deprived quintiles along with a lower MCI than men of other ethnicity in the same quintiles. These findings may help explain why deprivation increases fracture risk in white men yet has less effect on patients of other ethnicity.

Chapter 6 Influence of deprivation on treatment

This chapter aims to investigate if deprivation has an influence on the treatment of patients with a distal radius fracture. Characteristics of patients undergoing surgery for distal radius fracture in Leicestershire are compared with those treated conservatively using poisson regression modelling to identify if deprivation effects surgical treatment rate (section 2.8.1.3). Modelling of national patient data is used to investigate if deprivation influences treatment measures such as wait for surgery, daycase surgery, semi-elective surgery and post operative hospital stay (section 2.6.9, 2.8.1.6 & 2.8.1.9). Hospital trust data is used to explore national variation in surgical treatment with the effect of deprivation also studied (section 2.6.4, 2.6.6 & 2.8.1.7). This information can help identify and reduce inequalities in healthcare and provides useful information for hospitals to improve patient experience and pathways of care.

6.1 Surgical fixation rate in Leicestershire

Patients who underwent surgical fixation were compared with the total group who sustained fractures in Leicestershire over the four year period studied. A poisson regression model demonstrated no relationship with deprivation quintiles suggesting no difference in deprivation related to treatment. Patients aged under 50 years were less likely to undergo surgery (IRR 0.57, Cl 0.48 to 0.68, p=0.001). There was no association with gender (p=0.391). White patients were more likely to undergo surger surgical intervention (IRR 1.74, Cl 1.40 to 2.15, p=0.010) (Appendix 9).

Patients with intra-articular fractures (p=0.057) and those who sustained injuries outdoors (p=0.075) showed a tendency towards surgical fixation but this did not reach statistical significance.

Male gender was associated with intra-articular fracture (p=0.024). Injuries outdoors were strongly associated with the younger age group (p<0.001). There was no relationship between intra-articular fracture and any deprivation quintile.

6.2 National data Models

A series of regression models is reported to investigate the relationship between several measures of the quality of treatment and deprivation (IMD rank) along with a large number of independent variables in patients who required surgical fixation for a displaced distal radius fracture (Kirschner wires, plating, external fixation).

Independent variables included:

Patient details: age, gender, ethnicity, comorbidities, Charlson index, patient IMD score, procedure

Hospital trust details: total population, population aged 65 or over, population of women, cases performed per year, cases performed that month

Date/ seasonal details: year, day of week, month, cases performed per year, cases performed that month, minimum temperature

Hospital trust was included as a random effect for each model.

Significant values are included in the tables for each model along with results for deprivation as it is the main variable of interest. In the models day of the week was compared to Tuesday as this is a working day which is unlikely to be influenced by increased fracture rate at the weekend or different weekend hospital staffing and processes. Monthly results were compared to April as this month is unlikely to be affected by extremes of temperature and weather conditions seen in winter or summer which may influence fracture rate.

6.2.1 IMD rank

A linear regression model was created with IMD rank as the dependent variable and all the other variables mentioned above as independent variables (section 2.8.1.1). Low IMD rank indicates high deprivation so coefficients with a negative value are associated with more deprivation.

Men and black patients were more likely to be from more deprived areas. Higher numbers of comorbidities as measured by actual number of comorbidities and the Charlson score were both seen with deprivation. Trusts that carried out higher number of cases a year were associated with deprivation.

Increasing age and trust with a higher proportion of older patients were both associated with less deprivation.

Table 27 Patient and hospital trust factors which interact significantly with deprivation (measured by IMD rank) in patients who required surgical treatment for an acute distal radius fracture

Variable	Co-efficient	95% Cl lower limit	95% Cl upper limit	P value
Male	-1.069	-0.900	-1.238	<0.001
Black	-1.39	-0.127	-2.653	0.031
Comorbidity	-0.446	-0.382	-0.510	<0.001
Charlson score	-0.285	-0.135	-0.435	<0.001
Cases per year	-1.878	-0.125	-3.631	0.038
Older population	7.501	0.862	14.140	0.029
Increasing age	0.012	0.010	0.014	<0.001

6.2.2 Wait for semi elective trauma surgery

Linear regression modelling with hospital as a random effect demonstrated that patients who are listed for surgery and sent home to return on a planned date (n=9,318) wait longer if they are listed towards the end of the week (section 2.8.1.9). Mean time to surgery was 3.04 days (range 1 to 30 days, sd 3.14). Those listed from Thursday onwards wait on average more than three days. This is longer than the NICE guidance for fixation of intra-articular distal radius fractures. 6,538 patients underwent surgery within 3 days (70.2%) and 8,747 within 7 days (93.9%). Those with a higher number of comorbidities also had a significantly longer wait for surgery.

Table 28 Patient and hospital trust factors which interact significantly with wait for semi elective trauma surgery in patients who required surgical treatment for an acute distal radius fracture

Variable	Co-efficient	95% Cl lower limit	95% CI upper limit	P value
Comorbidity	0.066	0.010	0.122	0.02
Wednesday	0.413	0.224	0.602	<0.001
Thursday	0.594	0.404	0.784	<0.001
Friday	1.032	0.829	1.235	<0.001
Saturday	0.728	0.430	1.026	<0.001
Sunday	0.879	0.585	1.173	<0.001
Deprivation	-0.002	-0.003	-0.001	0.537

Day of the week is the day the patient was listed for surgery

Figure 44 Mean wait for semi elective trauma surgery in patients who required surgical treatment for an acute distal radius fracture dependent on the day they were listed for surgery with 95% confidence intervals



Mean wait for surgery by day listed

6.2.3 Daycase

A logistic regression model was constructed to assess whether any of the factors were related to the likelihood of the patient undergoing surgery as a day case (section 2.8.1.6).

Day case surgery was less likely for men, older patients and those with more comorbidities. Those who present on a Friday, Saturday or Sunday were also less likely to have surgery as a day case.

Patients who sustained their injuries in the winter months, times of low temperature or in months with more cases performed were more likely to have day case surgery. Fixation with K-wires was more likely to be carried out as a day case compared to plating or external fixation. Table 29 Patient and hospital trust factors which interact significantly with the likelihood of undergoing daycase surgery in patients who required surgical treatment for an acute distal radius fracture

All months from October to March inclusive were associated with more day case surgery. Day of the week is the day the patient was listed for surgery.

Variable	Co-efficient	95% Cl lower limit	95% Cl upper limit	P value	
Male	-0.228	-0.156	-0.300	<0.001	
Larger female population	-0.794	-0.024	-1.564	0.043	
Increasing age	-0.009	-0.007	-0.011	<0.001	
Comorbidity	-0.248	-0.219	-0.277	<0.001	
Charlson score	-0.089	-0.016	-0.162	0.015	
Friday	-0.221	-0.123	-0.319	<0.001	
Saturday	-1.645	-1.486	-1.804	<0.001	
Sunday	-1.887	-1.706	-2.068	<0.001	
Months – Oct-March - min	2.349	2.182	2.516	0.002	
Months – Oct-March - max	3.61	3.438	3.782	<0.001	
K-wires	0.071	0.003	0.139	0.035	
White	0.263	0.036	0.490	0.023	
Cases per month	0.189	0.117	0.261	<0.001	
Lower temperature	0.013	0.004	0.022	0.004	
Deprivation	0.0031	-0.0004	0.0066	0.079	

6.2.4 Pre operative wait for surgery

A further linear regression model investigating the wait to surgery showed that wait was longer for patients in September or those having fixation with K-wires. Patients who underwent external fixation or had surgery as a planned day case waited for less time. Table 30 Patient and hospital trust factors which interact significantly with increasing wait for surgery in patients who required surgical treatment for an acute distal radius fracture

Variable	Co officient	95% CI	95% CI	P value	
Variable	Co-enicient	lower limit	upper limit		
September	0.088	0.008	0.168	0.03	
K-wires	0.140	0.107	0.173	<0.001	
External fixation	-0.513	-0.408	-0.618	<0.001	
Daycase surgery	-0.415	-0.367	-0.463	<0.001	
Deprivation	0.0003	-0.0015	0.0020	0.748	

6.2.5 Post operative stay

Linear regression demonstrated that men, older patients, patients with more comorbidities and those who underwent k-wire or external fixation stayed in hospital longer after surgery. Wait was also longer for those who presented on a Saturday or in January.

Length of stay was less in those who underwent surgery as a day case, at times of more monthly procedures and in patients who had type I diabetes.

Table 31 Patient and hospital trust factors which interact significantly with post operative length of stay in patients who required surgical treatment for an acute distal radius fracture

Variable	Co-efficient	95% CI lower limit	95% Cl upper limit	P value	
Male	0.379	0.364	0.394	<0.001	
Larger trust population	0.158	0.015	0.301	0.033	
January	0.508	0.125	0.891	0.009	
Age	0.056	0.037	0.075	<0.001	
K-wires	0.554	0.405	0.703	<0.001	
External fixation	3.604	3.133	4.075	<0.001	
Comorbidity	1.284	1.227	1.341	<0.001	
Charlson score	0.318	0.181	0.455	<0.001	
Saturday	0.294	0.033	0.555	0.027	
Daycase surgery	-1.79	-1.578	-2.002	<0.001	
Diabetes type1	-1.333	-0.978	-1.688	<0.001	
Cases per month	-0.419	-0.248	-0.591	<0.001	
Deprivation	-0.0012	-0.0089	0.0065	0.762	

6.3 National Variation

59,802 surgical procedures for acute distal radius fracture for people aged 18 years and over were performed over the study period.

In England, the rate of surgical fixation of distal radius fracture ranged from 9.8 to 89.2 per 100,000 population (9.1 fold variation) (Figure 45). When the seven hospital trusts with the highest rates and the seven hospital trusts with the lowest rates are excluded the range is 16.5 to 58.9 per 100,000 population and the variation is 3.6 fold.

Figure 45 Acute distal radius fracture fixation rate (expressed per 100,000 person/years) by hospital trust in England from 2009 to 2013



Fixation rate per hospital trust

Hospital trust

Increasing size of hospital trust population was associated with reduced rate of fixation. (Figure 46). Trusts with a higher population of people aged over 65 years demonstrated a higher rate of fixation. Mean deprivation score of hospital trust, number of women and the presence of a hand unit did not influence rate of fixation (Table 32)(section 2.8.1.7).

Table 32: Negative binomial regression analysis of the relationship between hospital trust factors and surgical fixation rate in patients who required surgical treatment for an acute distal radius fracture

Variable	Co officient	95% CI lower	95% CI	P value	
Variable	Co-encient	limit	upper limit		
Population (100,000)	-0.202	-0.269	-0.135	<0.001	
Women (100,000)	-0.064	-0.221	0.093	0.4286	
Patients 65 and over (100,000)	0.827	0.339	1.315	<0.001	
Hand unit	0.089	-0.105	0.283	0.907	
Deprivation	-0.0003	-0.0019	0.0013	0.733	

Figure 46 Plot of acute distal radius fracture fixation rate (expressed per 100,000 person/years) by hospital trust in England population with 95% prediction intervals



Fixation rate and hospital trust size

6.4 Summary of the influence of deprivation on treatment

Deprivation did not independently influence any of the indicators of patient treatment measured (section 9.7). Men, older patients and those with multiple comorbidities had a less positive healthcare experience in terms of the factors reviewed. Considerable variation of rate of surgical fixation of distal radius fractures exists between hospital trusts across England (section 9.8). Patients who present to hospital on a Friday, Saturday or Sunday are less likely to have day case surgery and wait longer for semi elective surgery after being sent home. Further work should focus on improving pathways of care for older people, those with poor health and preventing patients from suffering inferior care due to sustaining an injury at the weekend.

Chapter 7 Prospective investigation of fracture risk

Social deprivation is associated with many fractures but the reasons why are unclear. The aim of this chapter was to identify if deprivation was associated with falls risk, mechanism of injury or osteoporosis in patients with a fragility fracture of the distal radius. Details of patients aged 50 years and over with a distal radius fracture were prospectively recorded over a one year period (section 2.7.1). Patients were sent a questionnaire including questions regarding mechanism of injury, comorbidity assessment, falls risk assessment tool and FRAX assessment of bone health (section 2.7.2). Location and height of fall was compared between the most and least deprived patients. Regression analysis was used to investigate the relationship of deprivation with falls risk and FRAX score (section 2.7.2.5). Understanding the reasons why deprived patients are more likely to sustain fractures can allow preventative measures to be implemented.

7.1 Participants and response rate

Data from all patients who had sustained a distal radius fracture during the one year study period was screened. After removing patients from outside the region, those with no address or who had passed away 754 patients were identified of which 521 were aged 50 years and over. Surveys were sent to these 521 patients and were completed by 333 patients (279 female; 54 male). Response rate was 64%.

Comparison between characteristics of responders and the four year Leicestershire fracture group was performed. There was no difference in the proportions seen in the most and least deprived quintiles overall (p=0.58) and when examined by gender separately (men p=0.95, women p=0.55) (Table 33). This suggests that the survey responders are representative of the total distal radius fracture population.

Table 33 Comparison of survey responders and total population by gender and most (Q1-2) and least (Q3-5) deprived quintiles

	Survey respon	ders	Total populati		
	Q1-2	Q3-5	Q1-2	Q3-5	P value
All	108 225		951 1852		0.58
Women	85	5 194		1566	0.55
Men	23	31	192	286	0.95

Data regarding mechanism of fall, falls risk and osteoporosis risk factors was completed by 96% of participants. Responders were also asked their height and weight. This was completed by 72% of patients. Multiple imputation was therefore performed using gender, age, deprivation rank, number of comorbidities, falls risk factors and other osteoporosis risk factors to generate the missing height and weight values (section 2.8.2). These values were then used to calculate FRAX scores for all patients.

7.2 Mechanism of fall

Comparison was performed between place and height of fall between the two most deprived quintiles and the three least deprived quintiles. A significantly higher proportion of falls occurred in the home in less deprived patients (Q1/2: 35%: Q3-5: 48%, p=0.037) and more falls happened outdoors in the road or street in deprived patients (Q1/2: 39%: Q3-5: 24%, p=0.001). There was no difference in the height at which falls took place with most being from standing height (Q1/2: 81%: Q3-5: 86%, p=0.336).

7.3 Osteoporosis risk

There was no difference in FRAX scores for major 10 year fracture risk and 10 year hip fracture risk between the two most deprived quintiles and the three least deprived quintiles (Table 34). Linear regression analysis was carried out to investigate the relationship between IMD rank and the two FRAX scores but no association was found (major fracture risk: p=0.274, hip fracture risk: p=0.283). FRAX score does incorporate gender and age but due to the likely large effect of these variables further separate linear regression models were constructed for each gender which included age as a separate independent variable. This again demonstrated no association between IMD rank and FRAX scores. Increasing age was significantly associated with higher FRAX scores for both genders.

Further regression analysis of individual osteoporosis risk factors revealed that smoking was the only risk factor with a significant relationship with deprivation (Table 35). A lower rate of smoking and higher rate of rheumatoid arthritis was seen with increasing age. Men were more likely to smoke and less likely to have rheumatoid arthritis. Table 34 Mean values for each osteoporosis and falls risk score for patients in the two most deprived quintiles(Q1-2) compared to those in the three least deprived quintiles(Q3-5)

	Q1-2 mean score	Q3-5 mean score	P value
FRAX 10yr major	15.75	17.43	0.227
fracture	sd 11.1	sd 12.2	
FRAX 10yr hip	6.50	7.75	0.257
fracture	sd 8.1	sd 10.0	
Falls risk factors	3.62	2.79	0.028
	sd 3.2	sd 3.1	

Table 35 P values from individual regression models investigating the relationship between osteoporosis risk factors and deprivation (as measured by IMD rank), age and gender

	<u>Depriva</u>	ation			Age	Age				Gender (male)		
Osteoporosis risk	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value
Parental hip fracture	-0.005	-0.040	0.030	0.762	-0.021	-0.053	0.011	0.189	-0.987	-2.216	0.242	0.116
Smoker	-0.052	-0.095	-0.009	0.019	-0.048	-0.089	-0.008	0.020 *	1.079	0.190	1.968	0.017
Steroids	-0.003	-0.032	0.025	0.812	-0.018	-0.044	0.008	0.183	-0.773	-1.681	0.136	0.1
Secondary osteoporosis	-0.005	-0.040	0.030	0.771	0.028	-0.004	0.059	0.083	-0.821	-2.046	0.405	0.189
Alcohol	-0.025	-0.072	0.023	0.307	-0.031	-0.072	0.011	0.15	-0.683	-1.692	0.326	0.184
Rheumatoid Arthritis	-0.024	-0.057	0.009	0.15	-0.044	-0.074	-0.013	0.005	-1.484	-2.947	-0.021	0.047 **

*less likely to be a smoker with increasing age

**less likely to be male

7.4 Falls risk assessment

Number of falls risk factors was significantly higher in those in the two most deprived quintiles compared to the three least deprived (Q1/2: 3.62: Q3-5: 2.79, p=0.028). Linear regression analysis also demonstrated a significant relationship between lower IMD rank and increased number of falls risk factors (p=0.002). Increasing age was also a significant factor (p<0.001).

A significantly higher proportion of people with a total falls risk score of four or greater (indicating risk of falling) was seen in those from the two most deprived quintiles (Q1/2: 39%: Q3-5: 28%, p=0.041). Mean IMD rank was lower in those with a score of four or greater indicating more deprivation (score \geq 4: mean IMD rank 16874, score <4: mean IMD rank 19094, p=0.042).

Separate regression analysis of the individual risk factors was performed (Table 36). Being worried about falling, having trouble stepping up onto a curb, often needing to rush to the toilet, taking medication which causes light headedness or tiredness and feeling sad or depressed were associated with deprivation. Out of these factors only two were not also associated with increasing age: taking medication which causes light headedness or tiredness and feeling sad or depressed. Gender was not an influence.

Table 36 P values from individual regression models investigating the relationship

between falls risk factors and deprivation (as measured by IMD rank), age and gender

	<u>Depriva</u>	ation			<u>Age</u>	Age				Gender (male)			
<u>Falls risk</u>	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value	Co- efficient	95% Cl lower limit	95% Cl upper limit	P value	
fallen in the past year	-0.017	-0.040	0.006	0.138	0.015	-0.005	0.036	0.149	-0.114	-0.732	0.504	0.718	
advised use cane or walker	-0.001	-0.004	0.002	0.643	0.103	0.072	0.135	<0.001	-0.554	-1.475	0.367	0.238	
feel unsteady	-0.011	-0.036	0.014	0.373	0.068	0.045	0.092	<0.001	0.021	-0.650	0.691	0.952	
hold onto furniture	-0.025	-0.054	0.005	0.101	0.067	0.039	0.095	<0.001	0.142	-0.651	0.935	0.725	
worried about falling	-0.038	-0.062	-0.014	0.002	0.051	0.029	0.073	<0.001	-0.608	-1.295	0.079	0.082	
push with hands from a chair	-0.016	-0.041	0.008	0.193	0.062	0.040	0.085	<0.001	0.035	-0.621	0.692	0.916	
trouble stepping onto curb	-0.065	-0.095	-0.034	0.024	0.077	0.047	0.106	<0.001	-0.358	-1.259	0.542	0.436	
rush to the toilet	-0.038	-0.064	-0.013	0.003	0.051	0.028	0.075	<0.001	-0.105	-0.800	0.589	0.766	
lost feeling in feet	-0.010	-0.048	0.029	0.63	0.054	0.018	0.090	0.003	0.435	-0.539	1.409	0.381	
medicine makes light- headed or tired	-0.047	-0.077	-0.018	0.002	0.025	-0.001	0.052	0.064	-0.242	-1.076	0.592	0.570	
medicine for mood/sleep	-0.022	-0.053	0.009	0.17	-0.003	-0.031	0.025	0.841	0.453	-0.306	1.212	0.242	
feel sad or depressed	-0.040	-0.067	-0.012	0.004	0.019	-0.005	0.044	0.118	0.312	-0.383	1.006	0.379	

7.5 Summary of prospective investigation of fracture risk

This study has identified some key findings. Distal radius fractures in more deprived patients are more likely to occur outdoors. Osteoporosis and risk of fracture measured using the FRAX score was not associated with deprivation (section 9.9.1). An important relationship between deprivation and higher falls risk leading to distal radius fracture has been identified (section 9.9.2). Further work should concentrate on understanding the mechanism for this and focus on specific falls prevention strategies.

Chapter 8 Risk of distal radius fracture after hip fracture

Distal radius fracture is common and the prevalence is increasing. Distal radius fracture is known to be a risk factor for a subsequent hip fracture. The aim of this chapter was to quantify the risk of hip fracture risk in men and women who have suffered a distal radius fracture compared to those who have not. A systematic review and meta-analysis of previous studies, including many large, recent, high quality studies was performed (section 2.5). The findings may assist with health care resource planning, assessment and management of patients presenting with distal radius fracture, and guide appropriate application of interventional strategies.

8.1 Systematic review and meta-analysis

A systematic review and meta-analysis was performed. The initial literature search identified a total of 1,203 articles. 10 further records were found by hand reference searching. After duplicate screening there were 1,087 records which was reduced to 99 for consideration after initial screening by the review team. (Figure 47)

Figure 47 Flow chart of study identification, screening and selection.



Secondary screening excluded 43 records leaving 56 for full text review(86-94, 187-232). After assessment of quality and bias 12 studies were deemed suitable to be included.

Five studies were retrospective and seven prospective. All 12 studies were published in English. 6 studies were from the United States and 2 from the United Kingdom. The other 4 were conducted in Canada, Denmark, Sweden and Taiwan.

Five studies reported results for women only and 1 for men only. 6 described combined results for men and women together (Table 37). Out of these 6 studies 3

also reported their results separately for men and women. This meant 4 studies provided individual results for men and 8 for women (Table 38 & 39).

Table 37 Description of Studies Included in the Meta-Analysis - Combined results for men and women

Study	Study Location	Study Period	Number (Total)	Number (wrist fractures)	Number with subsequent NOF	HR/SIR/RR	Control Population	Follow-up (person years)
	United					6.4 (2.6 to		
Owen et al. 1982	States	1945-1959	394	394	54	13.1)	Clinical/Societal	6,145
						2.27 (1.15 to		
Mallmin et al. 1993	Sweden	1968-1972	1,338	1,338	223	4.5)	Epidemiological	6,611
	United					2.66 (0.98 to		
Cuddihy et al. 1999	States	1975-1994	1,288	1,288	78	5.79)	Clinical/Societal	9,664
	United					3.22 92.81-		Mean FU
Robinson et al. 2002	Kingdom	1988-1999	1,075	1,075	-	3.66)	Clinical/Societal	3.7 yrs
	United							
Van Staa et al. 2002	Kingdom	1988-1998	13,581	13,581	-	2.0 (1.8-2.1)	Clinical/Societal	665,000
						3.45 (2.59-		
Chen et al. 2013	Taiwan	2000-2007	44,450	3,762	19	4.61)	Epidemiological	3,748

Epidemiological means a comparison of individuals with or without a prior fracture, and clinical/societal means the relative risk of a hip fracture in an individual or a group of individuals compared with the average risk of a hip fracture in the population.

Study	Study Location	Study Period	Number (Total)	Number (wrist fractures)	Number with subsequent NOF	HR/SIR/RR	Control Population	Follow-up (person years)
	United					1.3 (0.9 to		
Owen et. al 1982	States	1945-1959	350	350	47	1.7)	Clinical/Societal	6,145
						1.8 (1.3 to		
Lauritzen et al. 1993	Denmark	1976-1984	1,162	1,162	49	2.2)	Clinical/Societal	4,588
						1.54 (1.24 to		
Mallmin et al. 1993	Sweden	1968-1972	1,126	1,126	197	1.93)	Epidemiological	34,221
	United					1.44 (1.12 to		
Cuddihy et al. 1999	States	1975-1994	1,045	1,045	72	1.86)	Clinical/Societal	9,664
Schousboe et. al	United					1.43 (1.17 to		Mean FU
2005	States	1989-2002	7,880	1,224	-	1.74)	Epidemiological	10.1 yrs
						1.29 (0.88 to		
Hodsman et al. 2008	Canada	1990-2004	21,432	1,225	-	1.89)	Epidemiological	359,737
	10					1.04 (0.71 to		
Gehlbach et al. 2011	countries*	2006-2008	51,762	4,411	-	1.51)	Epidemiological	2 yrs
	United					1.5 (1.32 to		Mean FU
Crandall et al. 2015	States	1993-2010	160,930	8,792	431	1.71)	Epidemiological	11.8 yrs

Table 38 Description of Studies Included in the Meta-Analysis – Women only

*Australia, Belgium, Canada, France, Germany,

Italy, Netherlands, Spain, United Kingdom and

United States

Epidemiological means a comparison of individuals with or without a prior fracture, and clinical/societal means the relative risk of a hip fracture in an individual or a group of individuals compared with the average risk of a hip fracture in the population.

Study	Study Location	Study Period	Number (Total)	Number (wrist fractures)	Number with subsequent NOF	HR/SIR/RR	Control Population	Follow-up (person years)
	United	1945-				6.4 (2.6 to	Clinical/Societ	
Owen et al. 1982	States	1959	44	44	7	13.1)	al	6,145
		1968-				2.27 (1.15	Epidemiologica	
Mallmin et al. 1993	Sweden	1972	212	212	26	to 4.5)	I	6,611
	United	1975-				2.66 (0.98	Clinical/Societ	
Cuddihy et al. 1999	States	1994	243	243	6	to 5.79)	al	9,664
Barrett-Connor et al.	United	2000-				1.9 (1.1 to	Epidemiologica	Mean FU
2012	States	2002	5,878	-	-	3.2)	I	8.9 yrs

Table 39 Description of Studies Included in the Meta-Analysis – Men only

Epidemiological means a comparison of individuals with or without a prior fracture, and clinical/societal means the relative risk of a hip fracture in an individual or a group of individuals compared with the average risk of a hip fracture in the population.

Meta-analysis of women only studies revealed low heterogeneity. There was a significantly increased log relative risk of suffering a hip fracture after a distal radius fracture of 0.36 (95% CI 0.24 – 0.47) compared to those with no previous distal radius fracture (Figure 48). For men only studies heterogeneity was again low. Log relative risk was 0.75 but not significant with 95 % confidence intervals of -0.07 to 1.58 (Figure 49). Combined studies of pooled results for men and women had high heterogeneity. Using a random effects model log relative risk was 0.60 (CI 0.34 – 0.86) (Figure 50).

Figure 48 Meta analysis of studies reporting results for risk of hip fracture after distal radius fracture for women only



Increased risk of hip fracture after wrist fracture - Women

Figure 49 Meta analysis of studies reporting results for risk of hip fracture after distal

radius for men only

Study	TE seTE		95%-CI	W(fixed) W(random)
Barrett-Connor 2012 Cuddihy 1999 Mallmin 1993 Owen 1982	0.64 0.5357 0.98 1.2270 0.82 0.8546 1.86 2.6786 -	 	0.64 [-0.41; 1.69] 0.98 [-1.43; 3.38] 0.82 [-0.86; 2.49] - 1.86 [-3.39; 7.11]	61.6%61.6%11.7%11.7%24.2%24.2%2.5%2.5%
Fixed effect model Random effects mod Heterogeneity: I-squared=I	el 0%, tau-squared=0, p= <u>0.9687</u> -6 -4	-2 0 2 4 6	0.75 [-0.07; 1.58] 0.75 [-0.07; 1.58]	100% 100%

Increased risk of hip fracture after wrist fracture - Men

Figure 50 Meta analysis of studies reporting results for risk of hip fracture after distal radius for men and women combined



Increased risk of hip fracture after wrist fracture - Men and Women

8.2 Summary of risk of distal radius fracture after hip fracture

This meta-analysis confirms an increased relative risk of hip fracture after distal radius fracture in women (section 9.4.1). Previous distal radius fracture in women was associated with a relative risk of hip fracture of 1.43 (CI 1.27 to 1.60). Further studies are required in men but the findings suggest men with a distal radius fracture are also at higher risk of a hip fracture. Identification of those at highest risk of further fractures allows resources to be targeted efficiently. Effective interventions are available to prevent subsequent fractures such as a hip fracture (section 9.13). When a patient presents with a distal radius fracture these treatments can be commenced.

Chapter 9 Discussion

Distal radius fractures are the most frequently treated fracture worldwide and a huge burden on healthcare resources. Around 70,000 individuals are affected each year in the UK(98). Distal radius fracture is largely an osteoporotic fragility fracture. With an ageing population the number of distal radius fractures continues to rise(233). Estimates for the direct medical cost alone for treatment of fractures of the forearm and wrist have been predicted to total over £35 million by 2020 in the UK(99).

Previous fragility fracture is widely understood to be a risk factor for subsequent hip fracture. Several studies report increased fracture rate with deprivation but the reasons for increased fracture risk have not been established(27, 36, 52). Understanding reasons for inequalities in healthcare and which groups are at higher risk of injury can allow preventative mechanisms and prophylactic treatments to be put in place. Even relatively modest changes in distal radius fracture rate will have a large impact on health services.

The unique make-up of the Leicestershire population provides an excellent opportunity for studying the effects of deprivation and ethnicity on fracture rate. Overall Leicestershire has less deprivation than average for England. However, there are significant pockets of deprivation within the city. A large ethnic population enables the influence of ethnicity to be studied which is not possible in most parts of the country. There is only one emergency department in the whole county. This serves a large population of approximately one million people. A few minor injury units may manage distal radius fractures but the numbers are small and any significant injuries would be referred and captured in the local data. Utilising this data, along with national HES data, has allowed meaningful findings to be made in this thesis.

179

9.1 Accuracy of radiographic measurement

Almost every patient who sustains a distal radius fracture will undergo a plain radiograph. These investigations are cheap, quick and simple to perform. They are used by clinicians for diagnosis, planning treatment and monitoring healing. A huge wealth of information can be gained from each radiograph. In this thesis I wished to use radiographs to measure displacement and bone quality. For useful information to be obtained it is important that the measurement of these parameters is accurate, reliable and reproducible. Tests were therefore undertaken to investigate this. This information can be used by clinicians and other researchers.

9.2 Measurement of displacement

9.2.1 Extra-articular radiographic parameters

There is currently little available evidence to demonstrate the validity and reproducibility of radiographic measurements of distal radius fractures, and the studies that have been identified predate the introduction of modern hospital imaging systems such as PACS(131, 234).

The results in this thesis demonstrate strong correlation of both inter and intraobserver measurement of three of the four parameters measured: radial height, radial inclination and dorsal/palmar tilt.

The results of correlation of these parameters suggest that, following appropriate training, the degree of displacement of distal radius fractures can easily, and reliably be calculated using digital imaging software. It can therefore be used as a quick and cheap adjunct to assist management planning. Analysis of fractures using digital software is beneficial, as visual estimation alone has been shown to correlate poorly with actual measurement, and result in treatment decisions that may conflict with
those made based on such methodology when clinically relevant thresholds for acceptable displacement are applied(235).

There may be scope to apply the described methodology to the development of protocols for the management of distal radius fractures. Abramo et al. evaluated a standardised treatment algorithm which based management on the radiographic appearance of the fracture (i.e. the degree of fracture displacement), reducibility and stability, and the demands of the patient(236). They concluded that a protocol ensured similar and good subjective end results were achieved regardless of the severity of the fracture. Confirmation that measurements of fracture displacement are reliable could therefore support implementation of such a protocol into emergency departments to assist clinicians in management decisions.

9.2.2 Intra-articular radiographic parameters

An initial analysis of a small sample of intra-articular fractures revealed poor correlation. To investigate this in more detail, a larger, adequately powered study to determine if measurements of articular congruity of distal radius fractures are consistent between independent reviewers, was performed using the prospective study data.

Despite a larger sample size, the results still demonstrate poor inter and intraobserver correlation for measurement of both parameters of articular congruity, suggesting measurement of these parameters is intrinsically more difficult than the extra-articular components previously reviewed. Notably, when correlation for radiographs with \ge 2mm displacement was calculated independently, correlation was even poorer. This corresponds with other aforementioned studies but is significant as this is the generally accepted level beyond which intervention is advisable(234, 237). The observers stated that they found measurements difficult due to comminution and the multi plane nature of fractures causing difficulty clearly identifying the fracture line itself and therefore accurate placement of the digital measuring tools.

A review of the radiographs of the 20 fractures with the highest and lowest respective agreement was performed. Those with the highest agreement were minimally displaced, less comminuted, frequently had two large fragments and the fracture involved a small area of the articular surface of the distal radius. Those with poor agreement were significantly displaced, comminuted, involved large amounts of articular surface and were often impacted. Methods of measuring intra-articular step and gap generally involve the placement of several points from which to measure between and rely on large identifiable fracture fragments. Clearly with complex, broad, comminuted fractures this will be challenging and difficult to reproduce.

This data is of importance as it suggests that the use of computer assisted measurements of intra-articular fracture displacement using radiographs is unreliable, and therefore may not be useful to assist with management decisions, audit or research. Computed Tomography (CT) scanning has been advocated in the pre-operative assessment of complex intra-articular fractures for this reason. Evidence exists that CT evaluation improves the sensitivity of articular surface gapping and may alter treatment plans compared with decision making based on radiographs alone(160, 238). However, O'Malley et al. demonstrated that concordance of management decisions between observers based on CT may be poor(235). Newer methods of measuring step and gap on plain radiographs should be developed which do not rely on fractures having large, identifiable fragments.

This work has limitations. Measurements were performed by junior trainees, with limited experience in the review of radiographs of distal radius fractures, making erroneous measurement possibly more likely. However, the strong correlation of most parameters measured suggests that assessment was generally accurate, and junior trainees can therefore utilise the methodology described reliably. It may in fact be of greater relevance to junior staff, who may be less familiar with the management planning considerations for these injuries than senior colleagues, and therefore gain greater benefit from exact measurement when making decisions in the clinical environment. Measurement was made to 0.1mm and 0.1 degree, and this may reflect excessive precision to be practical in the clinical environment, particularly if computer

monitors are of poor resolution. However, having demonstrated strong correlation to this level of accuracy one would expect similar concordance at lower precision levels.

Measurement of radial height, radial inclination and dorsal/palmar tilt is a useful and reproducible means of assessing the degree of displacement of distal radius fractures by individuals trained in correct methodology. Inexperienced trainees can quickly be trained to perform the measurements to a high standard. They may therefore be used as a reliable adjunct to management planning, including protocols and guidelines based on radiographic parameters, and for audit and research purposes. Measurement of intra-articular distal radius fracture displacement is challenging and not reproducible using radiographs alone, despite the use of digital techniques. Results used for research purposes must be interpreted with caution.

In clinical practice the overall assessment of the injury in the context of the individual patient, their functional demand and wishes following appropriate counselling of relative risks and benefits of different management strategies, is of greater importance. However, accurate measurement of radiographic parameters can aid communication and help clinicians recognise fractures which may require intervention or specialist referral.

9.3 Measurement of bone health

The results show a high level of inter-observer (ICC 0.85, 95% CI:0.82-0.88) and intraobserver (ICC 0.93, 95%CI:0.86-0.96; ICC 0.91 95%CI:0.84-0.95) agreement between measurements of MCI. These were achieved following another single short, standardised training session in two doctors with no previous experience in MCI measurement. This would suggest that MCI is a reliable and reproducible measurement that can be quickly and accurately measured with the aid of digital imaging software. These results differ to those from older studies carried out prior to the widespread use of digital imaging software which criticised the reproducibility of MCI measurements. Naor et al quoted an intra-observer variability equivalent to 10% of the normal MCI in their 1972 paper and in 1976 Dequeker et al published figures stating the inter and intra-observer error in MCI measurements of the 2nd metacarpal equate to 1.5% and 1.2% for the outer cortical diameter and 6.4% and 4.8% for the inner cortical diameter respectively(239, 240).

It has been suggested that peripheral measurements of BMD have poor correlation with each other and with central DXA but MCI has been shown to correlate significantly with both forearm BMD and central BMD measurements indicating that MCI can be successfully used for assessment of bone mass and quality, and for predicting osteoporotic fractures(165, 166, 241, 242). The results in this thesis show a relationship between MCI and BMD of the spine and T-score which again suggests that MCI reflects BMD.

BMD declines with age, and following the menopause in women(243). Huachou et al looked at age and menopause-related changes in MCI in a sample of 383 women and found an annual decrease of 1.11% per year with acceleration in the rate of decrease in the 50-59 year old age group and in the early post-menopausal period(244). Similarly, these results show a sharp linear decrease in MCI after age 50. They found that the changes in MCI were more dependent on menopausal status than on ageing.

R² values in this work were similar to other studies looking at the relationship between patient factors and measures of fragility. Seo et al analysed the interaction between seven variables including age and BMI with BMD in over 2,000 Korean patients using multivariate regression and obtained an R² value of 0.21(245). In a similar analysis Kroger et al found that anthropometric and lifestyle factors explained only 18.7-25.4% of the variability of BMD in 14,220 peri-menopausal women(246).

This study also looked at the relationship of MCI to dorsal displacement in distal radius fractures. It has previously been demonstrated that patients with lower T-scores on central DXA scanning have more severely displaced and unstable distal radius

fractures(247). The results obtained in this work did not show a significant trend for increasing displacement with lower MCI.

Low BMD plays a greater role in the pathogenesis of fractures in older patients. Results in this thesis show that MCI correlates with BMD spine and that this correlation is stronger when MCI is less than 0.5. Most patients with an MCI <0.5 are older. This would suggest that MCI is more useful as a measure of BMD in this age group and is of more use in the analysis of distal radius fractures that have been sustained because of bone fragility rather than in all fractures of the distal radius fracture.

A wrist fracture can provide health practitioners with the opportunity to assess an individual's future fracture risk and allow initiation of preventative strategies and treatment. DXA scanning is the current gold standard for BMD assessment although other modalities such as quantitative CT and ultrasound are available. These incur a financial cost, a wait for an appointment and additional radiology staffing and reporting. The current cost of a DXA scan payable by commissioners is £69(248). The majority of PA radiographs taken for a distal radius fracture include the 2nd metacarpal. Calculation of the MCI from the 2nd metacarpal can allow a quick, simple and cheap assessment of a patient's bone mineral density.

91% of the patients included in this study with a distal radius fracture had plain film radiographs taken at presentation which enabled calculation of MCI from the second metacarpal. This suggests that most patients presenting with a distal radius fracture can have a quick and reliable assessment of BMD performed without the need for further investigations saving money, time and reducing radiation dose. This measurement can aid clinicians in determining which patients are at risk of further osteoporotic fractures and which should be referred on for further investigation and secondary fracture prevention therapies.

Table 40 demonstrates the relationship of MCI to T-score. The World Health Organisation defines a normal T-score as -1 or higher, osteopenia as a T-score between -1 and -2.5 and osteoporosis as a T-score of less than -2.5(166). The results suggest that these correspond to the MCI values seen in Table 40. Examples of

radiographs with MCI measurements that would fall into the osteopenia and osteoporosis categories are shown in figure 51 and figure 52 respectively with figure 53 showing a patient with normal BMD as calculated by MCI.

Table 40 MCI values corresponding to T-score values used to diagnose osteopenia and osteoporosis

	T-score	MCI
Normal	≥-1	≥0.50
Osteopenia	<-1 and >-2.5	<0.50 and >0.44
Osteoporosis	≤2.5	≤0.44

Figure 51 Example radiograph of an osteopenic patient (MCI of 0.50, equivalent to a T-score of -1)



Figure 52 Example radiograph of an osteoporotic patient (MCI of 0.44, equivalent to a T-score of -2.5)



Figure 53 Example radiograph of a patient with normal BMD (MCI of 0.53)



This data suggests that if the patients combined cortical width at the narrowest part of the 2nd metacarpal shaft is less than half the whole width of the metacarpal (i.e. an MCI of <0.50) then the patient is likely to be osteopenic. If the cortex occupies approximately 40% of the metacarpal width or less, then the patient is likely to be osteoporotic.

Limitations are noted. Measurements were again performed by junior trainees, with no experience of measuring. However, the strong correlations again demonstrated suggest that assessment was generally accurate, and junior trainees can therefore utilise the methodology described reliably. The sample of patients with a DXA scan and radiograph of the 2nd metacarpal was randomly generated. This meant patients undergoing the investigations for any indication were included. Data was not collected on whether patients were taking bone protective treatment at the time of either scan or the duration and nature of this treatment. Maximum anti-resorptive effects of bisphosphonates for example are seen within 3 months of commencing oral therapy and sooner for those receiving intravenous bisphosphonates(249).

The radiographs were from both left and right hands and no record was made from which side measurements were taken or which was the patient's dominant side. However, Kimura et al showed that MCI is nearly identical in both right and lefts hands as are other measures of bone mineral density and cortical thickness(250).

Furthermore, it was not possible to retrospectively standardise the angle at which AP and lateral radiographs of the 2nd metacarpal and distal radius were taken which could have affected the measurements. The radiology department have a standard protocol for radiographs of the hand and wrist which remained constant throughout the study period. The AP views from which MCI was calculated were obtained by placing the palm of the hand flat on a horizontal surface and the X-ray beam angled directly above. This approach is more representative of how MCI can be calculated and used in clinical practice.

In summary, this work demonstrates that MCI is a reliable and reproducible measurement which correlates with DXA measurements of BMD and T-score and that it declines with age. No association with dorsal angulation of distal radius fractures was seen. MCI provides a rapid, low cost assessment of bone frailty which can be

conducted in most patients who present with a distal radius fracture without the need for further investigations.

The aim of this thesis was not an in-depth analysis of radiographic measures but important results have been identified which are useful for clinicians and those carrying out research using radiographs.

9.4 Is distal radius fracture associated with a higher risk of subsequent hip fracture?

9.4.1 Meta analysis

Approximately 300,000 fragility fractures are seen in the United Kingdom (UK) each year(76). The greatest economic burden is due to hip fractures with nearly 85,000 UK hospital admissions per year accounting for over 1.8 million hospital bed days and an estimated cost of hospital care alone of over £1.9 billion(80). Incidence is rising and projections suggest the number of hip fracture hospital admissions could rise to 140,000 per year by 2036 if effective preventative strategies are not introduced(81). The cost of treating other fragility fractures is lower but still significant at over £200 million per year(251).

The purpose of this systematic review and meta-analysis was to investigate and quantify the relative risk of hip fracture for men and women who have suffered a distal radius fracture. Previous distal radius fracture in women was associated with an overall relative risk of hip fracture of 1.43 (Cl 1.27 to 1.60) compared to those without. A similar association was demonstrated in studies looking at pooled results for men and women together (RR: 1.82 (Cl 1.40 – 2.36)). Results of the four trials providing results for men only did not demonstrate a statistically significant increased relative risk (RR: 2.11 (0.93 - 4.85)).

Identification of high risk patients, and understanding of the reasons why they are at risk, allows effective primary prevention to be initiated. Successful primary prevention will prevent an initial distal radius fracture and reduce the risk of hip fracture.

Following a distal radius fracture secondary preventative measures can be commenced. Osteoporosis is very common but difficult to detect. Unlike an osteoporotic spine fracture which may run a more indolent course a patient with a distal radius fracture is likely to seek medical attention acutely. This then provides the opportunity to initiate secondary prevention. In the current economic climate resources are scarce and must be used prudently so targeting high risk groups is an efficient strategy.

Previous evidence has suggested men have a higher risk of hip fracture following a distal radius fracture. Robinson et al carried out a large prospective study of patients over 45 with a further fracture after an initial low energy index fracture(87). They found men were at higher risk of a second fracture following all the index fractures they investigated. Van Staa et al looked at over 200,000 patients in the UK General Practice Research database and found relative risk of further fractures was higher in men(230). In a study using life tables and then a subsequent meta-analysis from 2003 Haentjens et al found men had a relative risk of 3.22 (Cl 2.81 to 3.66) (92, 208). They suggested distal radius fracture is an early and sensitive marker of skeletal fragility in men. In their meta-analysis three studies with data for men were included. Sample size was small in these studies and confidence intervals large. This search only identified one additional study which despite showing a significantly increased relative risk (RR 1.9 (1.1 to 3.2)) when meta-analysis was carried out the overall relative risk was not significant. It is likely that this is a type 2 error and RR would be significant with larger studies.

Heterogeneity was very low in the meta-analysis of separate gender studies (men: $I^2=0\%$, P=0.97, women: $I^2=0\%$, P=0.75) but high when results for men and women were combined ($I^2=72.8\%$, P=0.0025). This again suggests important differences between men and women and is the most likely reason for the heterogeneity.

The main strength of this meta-analysis is the robust results obtained for relative risk of hip fracture after distal radius fracture for women. Overall sample size is large and heterogeneity low. Several important, large studies have been published since the previous meta-analysis by Haentjens et al(92). Whilst the results for men are not as reliable an important research question has been identified for investigation. This would suggest a prospective multi-centre longitudinal trial should be performed. A study design such as this would be required to enrol enough participants to answer the research question. Throughout the study stringent guidelines were followed and well recognised methods used. A multi-disciplinary advisory group was set up

including a senior clinical librarian, hand and wrist surgeons and statistical expertise. Regular meetings of the advisory group took place and any issues were resolved by consensus.

Limitations of this meta-analysis include the small number of studies reporting data for men only and the small sample size of these studies. The studies included report results for patients of different ages. All include the elderly but several also include younger adults. However, the number of distal radius fractures in the younger patients is likely to be small and not affect the results. Publication bias is a concern with studies showing significant positive findings potentially being more likely to be published.

This meta-analysis confirms and quantifies the increased relative risk of hip fracture after distal radius fracture in women. Further studies are required in men but one would expect that men with a distal radius fracture are at higher risk of a subsequent hip fracture. Identification of those at highest risk of further fractures allows resources to be targeted to those who need them most. Effective strategies are available to prevent further fractures including hip fracture. Presentation with a distal radius fracture provides an opportunity to implement these interventions.

9.5 Epidemiology of distal radius fracture in Leicestershire – a 10 year review

The aging population profile of the UK and the inherent associated challenges is well described. The profile in the Leicestershire study region closely matches that of the UK as whole, with change occurring at a similar rate. Ten years of data was reviewed for patients presenting with distal radius fracture to a single hospital trust, to define current incidence rates of distal radius fracture, and to determine if this is changing over time as the population ages.

This local Leicestershire data correlates with previous studies in demonstrating a bimodal distribution of distal radius fracture(101, 102). Post-menopausal women are most commonly affected, reflecting the significant impact of underlying osteoporosis in this population. Young men are injured more frequently than young women due to high-energy trauma. A recent UK multi-centre trial reported incidence rates of 9/10,000 and 36.8/10,000 person-years for men and women respectively, compared with our calculations of 83.9/100,000 person-years and 170.2/100,000 person years(98). The variation in rates for women is likely a consequence of their study including patients aged over 35 only. Fracture incidence in young women is low, resulting in lower overall female incidence rate. Reporting on data collected in the same study region in the 1980s, Donaldson et al. reported rates of 21.3/10,000, 18.2/10,000 and 24.3/10,000 for all patients, males and females(101). Although this study included paediatric patients, the disparity with my calculations suggests there may have been a real reduction in fracture incidence over time.

Early studies describe a steady rise in the rate of distal radius fracture in women from the age of about 40, before a plateauing of fracture incidence in the mid-sixties to early seventies(114, 252, 253), a trend which has been reproduced in a more recent study from Norway(104). However, other studies show no such plateau, with a continued rise with increasing age(105, 107, 108). In this thesis the data correlates most closely with that described by Lofthus et al.(110); a sharp increase in incidence of distal radius fracture in the peri-menopausal period climbing to a peak mean

incidence of 515/100,000 person-years in the 80s before a decline in the over 90 age group. Most other studies stop calculating incidence at 80 or 85, which may explain this seemingly new trend. In older men, the findings are consistent with those previously described(105, 107, 108); a steady but more gradual rise in distal radius fracture incidence occurring from the age of about 70, reflecting the delayed onset of osteoporosis in male patients.

Because of the aging population it could be expected to see an increase in the overall incidence of distal radius fracture. Previous evidence has demonstrated a real increase in age-specific incidence in distal radius fracture over time in multiple locations(254, 255). This data shows no statistically significant change in either of these measures. The study duration may be too short to demonstrate these trends. However, other studies have shown no change in incidence of distal radius fracture with time(109, 110). Lofthus et al. reported no increase in fracture incidence in Oslo over a 20 year period, while Jaglal et al. described a decline in the incidence of osteoporotic hip and distal radius fracture over a 13 year period in Ontario, and correlated this with a simultaneous increase in BMD testing and treatment with bone-sparing medications. The Leicestershire results may reflect similar mechanisms. Proactive treatment of patients at risk of fragility fracture is more prominent in the minds of both specialist and general medical practitioners, and fracture liaison services have improved significantly over recent years following publication of specific guidelines(248, 256). The benefits of these services has been clearly demonstrated elsewhere in the UK and may be relevant to our study outcomes(111).

The data demonstrate statistically significant fluctuations in rate of fracture in specific years, most notably in 2009 and 2010. These findings are likely to be the consequence of particularly severe winters. In particular, a significantly increased rate of fracture was noted in the female 40-69 age group, and most markedly in those aged 50-59, during these years. It could be hypothesised this is secondary to occult osteoporosis whilst maintaining high activity levels, therefore increasing risk of fracture in slippery conditions. Similar findings have previously been reported in studies on the effect of weather on distal radius fracture incidence in Rochester, Minnesota(116).

Analysis of seasonal data supports this conclusion. Overall fracture incidence increased in the winter compared with the other seasons, with this finding most marked in the female over 50 population. Melton et al. reports that seasonal variation in fracture incidence persisted after adjustment for weather conditions citing possible causes of reduced light levels with a resultant increased risk of falling in individuals with poor vision, and reduced vitamin D synthesis that can lead to osteoporosis(255). The findings correlate with other studies reporting on seasonality and poor weather conditions conducted in the UK and internationally(104, 115, 116, 257).

Analysis of weekly incidence of fracture also revealed notable trends. I report an increase in the number of fractures sustained during or immediately after a weekend. This primarily resulted from an increase in number of injuries in the under 50 age group, possibly consistent with fractures sustained in sporting or leisure activity. More significant were the findings relating to mean age at fracture across a week. This decreased significantly over the weekend period in the overall population and in the over 50 group in isolation. I postulate that those aged 50-70 are more likely to be active over the weekend compared to those aged over 70, and therefore at greater risk of fracture.

The projections for the future national incidence of distal radius fracture are based on the gender and age specific average incidence of injury over the 10-year period and the ONS UK population projections for the next 20 years(157). A 23.1% overall increase in the number of fractures by 2036 to total over 83,000 fractures is predicted. In the over 50 age group one would expect a 33.2% increase with almost 60,000 fractures in 2036. These figures are lower than those predicted in previous studies. Burge et al. estimated that over 60,000 distal radius fractures would be sustained in the over 50 demographic annually by the year 2005(99). However, these figures were calculated using incidence rates from a study conducted in Cardiff(258) over a single year, which reported an overall incidence of 4.79 fractures/1000 person-years, nearly four times greater than the Leicestershire overall mean incidence rate, explaining the disparity in projections. One would expect the methodology in this thesis to give a more accurate estimate of future fracture numbers because of the age and gender specific population projections, and average incidence rates calculated over an

extended period. However, the variation seen may add weight to the idea that improved detection and preventative methods have reduced the rate of fracture.

There are potential limitations to this study. The incidence rates may underestimate the true figure; although UHL is the only hospital trust in this region, patients living in the peripheries may have attended other hospitals, though one would expect this to be of minor influence and the figures to therefore be reliable. Data collection was based on ED coding but measures were taken to ensure the accuracy of this and it was consistently high. Many databases include only fractures which required intervention but the Leicestershire data also included cases treated non-operatively. Population estimates were collected from national governmental departments and therefore should be precise. The study could be strengthened by more rigorous analysis of data for narrower age bands. This was performed to calculate overall incidence rates, but trends in weekly and seasonal variation were not analysed as the numbers were too small to do so.

This section of the thesis reports one of the largest analyses of distal radius fractures performed in the UK to date. Multiple trends have been identified, some of which have not previously been described. The ability to make estimates for the number of distal radius fractures at specific weekly or seasonal periods are useful for workforce and resource planning in emergency departments, trauma units and subsequent rehabilitation and social care services. Furthermore, given the predictive value of distal radius fracture for fragility fracture at other anatomical sites, the data will assist with forecasting the overall burden of management and prevention of osteoporosis.

Valuable and current information regarding the epidemiology of distal radius fracture based on one of the largest databases recorded is presented, which can be extrapolated to make projections of the burden of this injury to the UK in the future. Incidence of distal radius fracture does not appear to be rising but with an aging population the number of fractures is increasing. Estimates for the direct medical cost alone for treatment of fractures of the forearm and wrist have been predicted to total over £35 million by 2020(99). With healthcare resources under ever-increasing demand, it is vital that we act to implement preventative strategies to reduce the burden of fragility fracture in the future.

9.6 The influence of deprivation on distal radius fracture

9.6.1 Effect of deprivation on distal radius fracture rate

Further in-depth analysis of four years of the Leicestershire data (2007-2010) demonstrated that deprivation was strongly associated with distal radius fracture. Similar results were seen in both age groups studied (below 50, 50 and over) with more fractures seen than expected in the two most deprived quintiles and less seen in the least deprived two quintiles. The number of fractures seen in the median quintile (quintile 3) was as expected. Mean age of the deprived distal radius fracture patients was significantly lower; they sustain their fractures almost 4 years earlier than those in the less deprived quintiles. A similar finding was also seen in those aged 50 and over with patients in the least deprived quintiles suffering a fracture nearly 2 years later.

Incidence of distal radius fracture is higher in women but the poisson regression model has shown that being male is a risk factor for distal radius fracture when deprivation, ethnicity and age are considered.

Curtis et al reported fracture rates by gender and IMD quintile from UK General Practitioner data in their study involving 11.3 million people(27). They also found deprived men had higher risk of fractures. In men, the relative risk of vertebral, wrist and hip fractures increased with deprivation. No association between risk of wrist or hip fractures was seen in deprived women and vertebral fracture risk was reduced. They suggested that the difference between genders with deprivation was due to adverse lifestyle factors such as smoking, alcohol and poor diet which are seen more often in men than women(28-30). Other studies have reported similar findings(32).

The poisson regression analysis of the whole group revealed significant differences related to ethnicity with deprivation not being associated with fracture risk in patients of other ethnicities compared to white patients. Fracture risk was higher for men, patients aged 50 and over, and for white patients. Similar findings were seen

reviewing the national data of patients in England who underwent surgical fixation. In patients aged 50 years and over deprivation was only related to fracture risk in white patients. Trends related to deprivation were not seen in the other ethnic groups.

Curtis et al also reported fracture rate by ethnicity. They found white patients had the highest fracture rate, it was lowest in black patients and intermediate for South Asians. This is consistent with other studies and the incidence rates in this thesis. Differences in height, body composition, bone architecture and bone mineral density between races have been demonstrated between ethnicities with black people having higher BMD. White people are generally taller which increases fracture risk. No analysis to investigate interactions between factors was carried out by Curtis et al and they acknowledged that some data was missing with ethnicity unknown for 44% of patients with a wrist or forearm fracture.

In the whole cohort of patients, who underwent surgical fixation in England over the four year period studied, deprivation was associated with being male, black and having more medical problems. This information is unsurprising and it is well described that deprivation is associated with these factors. A higher number of cases per year was also associated with more deprivation. This could reflect the higher rate of fracture in deprived areas and that large units carrying out many procedures are likely to be in major cities who have high rates of deprivation. Increasing age and hospital trusts with a larger proportion of older patients were associated with less deprivation. Again, this is unsurprising as older patients tend to be more affluent.

9.6.2 Effect of deprivation on radiographic measurements

Radiographic analysis of patients in the two age groups did not demonstrate any significant findings. The GAM relationship between increasing deprivation and increasing intra-articular step size in younger patients came close to significance potentially suggesting more displaced intra-articular fractures occur in younger, deprived patients (p=0.052).

Further sub analysis of radiographic findings split by age, gender and ethnicity revealed interesting results. No differences were seen between the groups in the younger patients. In those aged 50 years and over women in the most deprived quintiles had a higher MCI than those in the less deprived quintiles and when also compared to white women from the same quintiles. This may help explain why an association with fracture and deprivation was not seen in those of other ethnicity. These findings may represent a deprived ethnic population who are still required to be active, and possibly in employment, who therefore maintain bone mineral density. The reverse was true for white men in the most deprived quintiles who had a lower MCI than white men in the less deprived quintiles along with a lower MCI than men of other ethnicity in the same quintiles. These findings may help explain why deprivation increases fracture risk in white men yet has less effect on patients of other ethnicity. However, it must be noted that the number of patients becomes small on subanalysis, particularly men and patients of other ethnicities.

9.7 The influence of deprivation on distal radius fracture treatment

Regression analysis comparing those who underwent surgery in Leicestershire to all of those who sustained a distal radius fracture in the same region allowed independent factors for fixation to be identified. Patients aged 50 years and over were more likely to undergo surgery which is most likely due to the increased rate of displaced fragility fractures. Deprivation did not influence fixation rate which supports the radiographic findings showing no overall association between deprivation and displacement.

White patients were more likely to undergo surgical intervention (IRR 1.737, CI 1.139 to 2.648, p=0.010). This was despite our radiographic analysis, which included all patients who underwent surgery, showing no difference in displacement between the ethnic groups. The reasons for this are likely to be complex and beyond the scope of this thesis but an important research question has been identified. Cultural beliefs, family support systems, functional demands and language may play a part but bias must be considered. Unconscious bias is part of normal cognitive processing where a

person's subconscious prejudicial beliefs or unrecognized stereotypes influence their responses. Surgical decision making is understood to be particularly vulnerable to unconscious bias and is well described in many surgical specialties including orthopaedics(259).

The modelling of the National data against measures of quality of care, including deprivation as an independent variable, produced some useful findings.

9.7.1 Wait for semi elective trauma surgery

These results are reassuring and demonstrate that most hospitals are able to meet the NICE guidelines for timing of surgical fixation. Increasing number of medical comorbidities was associated with increased wait to surgery. Differences existed between hospitals which were eliminated when hospitals were added to the model as a random effect. The decision to operate, when taken on a Friday, Saturday or Sunday, was associated with a longer wait for surgery.

Specialist surgeons will often be required to operate on complex intra-articular fractures. Many hospitals, particularly smaller ones, will only have one or two surgeons available over weekend periods. It is therefore unlikely that they will have necessary sub speciality cover available to operate on complex distal radius fractures over the weekend. Therefore, for a fracture that is first seen by a specialist on a Friday it may be difficult for the surgery to be performed over the weekend by a necessary specialist surgeon. Larger hospitals may be able to innovatively plan their weekend and holiday on-call cover to ensure there are surgeons available to deal with most the commoner complex fractures.

9.7.2 Pre-and post-operative wait to surgery

Preoperative wait for surgery was less in those undergoing day case surgery and treatment with external fixation. External fixation is often used for severe displaced fractures with significant soft tissue injury. These usually require urgent surgery so this is likely to be the reason for the reduced wait for these patients. Having surgery as a planned day case was also associated with less wait. This suggest that units who are able to offer this service can do it in a timely fashion and patients being treated as a day case are not disadvantaged compared to those who are inpatient. This may also be influenced by the fact that patients chosen to be treated as a day case are likely to be medically fit.

It is unclear why the month of September is associated with a longer wait to surgery. This may be an anomaly. However, it could represent the fact that September is likely to be the month when initial winter weather is first seen and therefore could lead to an increase in fracture rate. During winter, hospital trusts are often able to employ methods to increase their capacity for surgery to deal with the increased number of patients. In September, these mechanisms may not be put in place yet and therefore hospitals are not prepared for the increased workload that the start of winter weather may bring. This could suggest the hospitals may be being reactive rather than actively planning. Patients having treatment with K-wires also waited longer. K-wires are often favoured in older patients and those who have more comorbidities and this may explain a longer wait due to surgical and anaesthetic planning reasons.

Increased post- operative length of hospital stay was seen in men, older patients, patients with more comorbidities and those treated with K wires or external fixation. This is likely to reflect the fact that these factors are more likely to be seen in patients with poorer health or more severe fractures. Admission or listing for surgery on a Saturday is also associated with a longer stay. As seen in the other models this may be due to an initial longer wait for surgery. Interestingly diabetes type I was associated with a shorter hospital stay. Reasons for this could be that type I diabetes is seen in younger patients compared to Type 2 diabetes. Another reason could be that these

patients are identified to be at risk of medical and surgical complications so a high level of care is devoted to them leading to a smooth pathway with minimal stay and good outcomes.

9.7.3 Day case surgery

Similarly to the models above factors relating to generally poorer health such as being male, older and more comorbidities reduce the likelihood of day case surgery. Patients admitted or listed for surgery on a Friday, Saturday or Sunday were significantly less likely to have day case surgery. This suggest that day case services stop or are commonly reduced at the weekend. During the winter months (from October to March), during months when more cases are performed and at times of low temperature day case surgery rate is higher. This may again reflect the ability of hospitals to increase their capacity during the busy winter months. More K-wire procedures were performed as a day case compared to plating and external fixation. This technique is often used for simpler fractures and is unlikely to be used for severe injuries requiring hospital admission so may often be more suitable to carry out as a day case.

9.8 National variation in fixation rate

Considerable variation exists across England in the rate of surgical fixation for distal radius fracture. With the seven hospital trusts with the highest and lowest rates of fixation excluded patients still experience a 3.6 fold difference in treatment rate depending on where they live.

Only three other studies have investigated variation in distal radius fracture treatment and none from the United Kingdom. An in-depth analysis was therefore performed in this thesis. Fanuele et al studied Medicare claims in the US and found a significant

variation related to age and religion(149). Chung et al built on this further and reported that women, white patients and those treated by a hand surgeon were more likely to receive internal fixation(150). A recent study from the Netherlands showed no variable influenced operative rate except a small significant increase in fixation with increasing age in one of the years studied(151).

Preference sensitive care describes medical treatment for which the evidence does not clearly support one treatment option over another(260). Many surgical procedures including distal radius fracture fixation are an example of this. Decision making should therefore be shared, with patients making an informed choice from clinically appropriate options after discussion regarding risks, benefits and current evidence with a clinician. Most surgeons can agree that very displaced fractures will require surgical intervention but for those which have a moderate amount of displacement there is only limited evidence and guidance about those which would benefit from surgical intervention. Without high quality evidence, it is difficult to understand the value of the intervention.

This analysis has shown hospital's serving a smaller population have a higher tendency towards fixation. "Roemer's Law," is the notion that an increase in the number of hospital beds per population increases rate of hospital utilisation(261). It was deduced after a positive correlation between the number of hospital beds available and the number of hospital days used per population was identified. It is thought to be due to induced demand, which occurs when clinicians encourage patients to use services that they may not have chosen if fully informed. Essentially it means that if capacity is available then it will be utilised. It could be hypothesized that smaller units are less busy and have more capacity to treat fractures surgically. This may be an example of overuse of an intervention. Conversely it may be that large, busy units with limited theatre capacity treat more patients conservatively and deny patients an effective intervention. Another reason may be that clinicians in the larger university hospitals have considered the inconsistent evidence regarding fixation and have a higher threshold for intervention. However, there was no change in fixation in specialist hand units in which one would expect clinicians to have a thorough understanding of the published literature.

Other reasons for variation may include the initial fracture assessment and management. The Leicester trauma unit is large and busy with a low surgery rate. All displaced fractures are referred to the oncall trauma team from ED and nearly all displaced fractures are manipulated that day in fracture clinic under image intensifier guidance. This prompt reduction and cast stabilisation by a specialist team may often be adequate definitive treatment limiting the need for surgical fixation. If initial reduction and stabilisation is inadequate or referral is delayed, then reduction and surgical stabilisation is more likely to be required.

Increased fixation rate is seen in trusts with higher populations of older people. This is likely to be warranted variation due to a higher rate of fragility fractures which are often displaced and unstable requiring surgical treatment and related to the needs of the population. Interventions such as bone health programs and falls risks prevention may reduce osteoporosis and falls and subsequently the number of displaced fractures in older patients but are long term and costly.

This analysis demonstrates significant variation in surgical treatment for distal radius fracture in England. Variation in the characteristics of the trust population account for some of this. Variation in fixation rate related to the size of the trust appears unwarranted. Improved initial treatment pathways may reduce the need for surgical intervention. Further research is required to identify which patients with distal radius fracture would benefit from surgical fixation and provide evidence stop based guidance to clinicians and reduce unwarranted variation.

Limitations of these models include the use of HES data which only contains details of those admitted to hospital. This does not identify those patients who have attended and had a distal radius fracture treated conservatively or with reduction in the ED or fracture clinic as the definitive treatment. However, in terms of cost and resource usage it is more important to identify trends with those admitted to hospital and treated with more expensive surgical interventions. investigation of trends in type of fixation was not attempted. This has already been reported by Costa et al in response to the DRAFFT study(127). Type of intervention changed but the overall rate of fixation did not change over the same period as studied in this thesis. For this analysis hospital trust size, has been equated with 'busyness' of that unit and access to theatre

for surgical intervention. Anecdotally that seems correct with large trauma units dealing with the highest volume of patients. However, to investigate this further would require a complex study involving staffing levels and theatre availability for all hospital trusts. In this analysis, the background trust population has been used as the denominator to calculate the rate of intervention. Others have used the total number of distal radius fracture in each population. The method used in this thesis allows simple calculation of fixation rate for different geographical areas which is easily obtainable to allow comparison.

It was reassuring that deprivation did not seem to independently influence patient's individual treatment. It was a recurring theme that men, older patients and those with multiple comorbidities have less positive healthcare experience in terms of the factors reviewed. Whilst this is inevitable to some degree further work could focus on improving pathways of care for those with poorer health. It was noticeable how hospitals do have the ability to increase workload and improve services for their patients during busy times. Further investigation of these efficiency measures may be helpful in improving service throughout the year. Weekend services are a controversial subject. It is difficult to maintain appropriate staffing levels and subspecialty expertise throughout the full week. Those who present on a Friday, Saturday or Sunday are less likely to have day case surgery and they wait longer for surgery after being sent home. Trusts should look further into ways of preventing patients from suffering an inferior pathway of care due to suffering an injury at the weekend.

9.9 Investigation of fracture risk

The prospective analysis of fracture risk revealed important findings. Fractures in more deprived patients are more likely to occur outdoors. Osteoporosis and risk of fracture measured using the FRAX score was not associated with deprivation. Falls risk is associated with increased deprivation.

9.9.1 Osteoporosis and deprivation

As discussed in detail in earlier sections the relationship between osteoporosis and deprivation is complex and not well understood. Some studies do show a significant relationship between increasing deprivation and osteoporosis but several others do not. Curtis et al found osteoporotic fractures were associated with deprivation in women but not men(27). Regression analysis in this thesis demonstrated that the effect of deprivation in older patients was more significant in men. This prospective analysis did not reveal a relationship between deprivation and osteoporosis for patients aged 50 years and over. The study population was representative of the background population which is largely white and female. It has been shown that lifestyle factors which are adverse to bone health are more commonly seen in men. This may explain the difference between genders. The radiographic analysis in this thesis would support this with deprived men having lower MCI. It may be that the effect of osteoporosis is mainly seen, in relation to deprivation, in white men from deprived areas who are at high risk of having a lifestyle which contribute to poor bone health. Due to smaller numbers of fragility fractures in men these findings are difficult to identify and in this analysis numbers may have been too small to demonstrate this.

The only component of the FRAX score which was associated with deprivation (and male gender) was smoking. This has been well described previously with smoking rates being shown to be four times higher amongst the most disadvantaged. It is also thought that the disadvantaged are at increased risk of harm due to tobacco. Interventions to prevent smoking are less successful in those who are more deprived. Hiscock et al found that raising the price of tobacco was the intervention most likely to reduce health inequalities due to smoking(262).

Smokers from more deprived areas describe different reasons for smoking and relapsing after quitting involving more nervousness and depressive symptoms. More problems with living conditions including homes and the outside neighbourhood were reported in those from deprived areas. This is thought to contribute to more stressful lives and quitting smoking not being a priority. Smoking is a modifiable risk factor so any effective interventions to reduce smoking rate, especially among more deprived

patients and men, may have some positive effect on bone health along with general health. The relationship between smoking, male gender and fracture risk found may allude to negative lifestyle factors affecting bone health seen in deprived men. A study with a large cohort of men may demonstrate this more clearly.

9.9.2 Falls risk and deprivation

Falls risk was significantly associated with increasing deprivation. Patients with deprivation were significantly worried about falling, had trouble stepping onto a curb and often had to rush to the toilet. These risks were also associated with increasing age. Feeling sad or depressed or taking medicine which makes them lightheaded or tired was independently associated with deprivation alone. Depression and other mental health problems are well known to be associated with deprivation(263). Polypharmacy, drug side effect problems and depression are potentially modifiable through primary care.

Limitations of this prospective analysis include the response rate of 64% despite multiple efforts to increase response rate. This is a common finding with this type of data collection and the comparisons with the population studied show that the data is representative of the overall population. The number of patients of other ethnicity in the study is low and the responders are mainly white women. This does not allow analysis of trends related to ethnicity but again it is representative of the largest group of people to suffer a distal radius fracture, white women, and therefore the most important to target for prevention.

Higher falls risk with deprivation leading to distal radius fracture is a key finding from this thesis. Further work should concentrate on understanding the reasons for this and identify preventative strategies.

9.10 Burden of falls

Every year 30% of people over 65 will fall and the rates increase with increasing age(264). Almost a third of these who fall suffer injuries which cause reduction in independence and mobility and increase the risk of death(265). 5% of falls result in a fracture(266). Repeated falls can lead to fear of further falling, depression and loss of self-confidence leading to social withdrawal. Falls prevention strategies have been shown to be effective. The National Service Framework for Older People and recent NICE guidelines set out guidance for prevention of falls(177, 267). Multifactorial interventions including strength and balance training, vision assessment, home hazard identification and medication review can reduce incidence of falls(268-270).

With an ageing population, the number of falls is expected to rise so further work is required to reduce falls and the subsequent morbidity, mortality and healthcare burden. Falls risk factors can be classified into three groups: intrinsic factors, extrinsic factors and exposure to risk.

Intrinsic factors include previous falls, age, and female gender. Medicines such as the use of benzodiazepines, diuretics, psychotrophic drugs and some anti arrhythmic medications increase risk(271, 272). Polypharmacy is also associated with falls, with a significant increased risk if a patient is taking more than four medications, irrespective of the type of drug(273). White people are also more likely to fall(274). Other medical comorbidities, gait, vision and foot problems(275, 276), sedentary behaviour(277), fear of falling(278) and impaired cognition(279) are also intrinsic risk factors.

Extrinsic risk factors relate to environmental hazards, footwear and inappropriate walking aids. Up to 50% of falls amongst those living in the community has been thought to be due to environmental factors(280). These include poor lighting, uneven surfaces and slippery floors.

The risk exposure is thought to be highest in the most inactive and the most active people. Some studies have suggested walking increases risk of falls whereas others have suggested more physical activity reduces fall rate(281, 282). Some activity may increase risk by exposing people to environmental hazards(283).

It is thought that intrinsic factors are the most important in patients aged 80 and over(280). In younger older people, extrinsic factors are more significant. Falls risk for all patients increases exponentially with increasing number of these risk factors(284).

In deprived patients, it seems likely that risk factors from all three groups are present. This analysis has identified risks related to medication, depression, falls outdoors and concerns about falling. The multifactorial falls prevention methods are largely targeted at frail, elderly people who fall indoors. Distal radius fracture patients are generally fitter and more mobile than hip fracture patients with more injuries occurring outdoors. This is particularly true of the deprived patients who were younger and had a significantly higher proportion of injuries outdoors. Injury rate was seen to rise in the younger women of the older age group during low temperatures. Prevention of outdoor falls in this group has enjoyed comparatively little attention and there are no established guidelines(285-287). Simple measures such as walking aids, appropriate footwear and avoidance of walking outdoors in slippery conditions would seem sensible and acceptable(288).

The World Health Organisation published a report by Todd et al in 2004 entitled: What are the main risk factors for falls amongst older people and what are the most effective interventions to prevent these falls?(289)

The report reviewed evidence and identified gaps in knowledge and conflicting evidence regarding falls. They noted that there was very limited research regarding the relationship between falls risk and socio-economic factors and the effectiveness of interventions in different socio-economic groups. It was also seen that very few studies have investigated the relationship between ethnicity and falls or the effectiveness of fall prevention for men or different ethnic groups.

They suggested a three stage approach to falls and subsequent fracture prevention.

1) Identifying high-risk groups of older people

2) Further assessment of high-risk older people to identify individual risk factors for falling

3) Interventions to reduce risk factors

This thesis provides valuable information to help identify those at risk and recognise the most important risk factors for falling which can then be used to put in place specific preventative interventions.

9.11 Increase of deprivation

The respected Joseph Rowntree foundation has recently published a report entitled UK poverty 2017(290). They define poverty using the main indicator being whether someone lives in a household with an income less than 60% of expected median income. For the past 20 years poverty has been falling in the UK. However, they noted that this trend appears to be reversing. Poverty amongst those aged over 65 has risen from 13% in 2011/12 to 16% in 2017. Poverty is also increasing amongst children and adults of working age with children. These findings were thought to be due to higher costs of housing, less state support through benefits and tax credits and a lack of impact of a continued rise in employment. One in eight workers were found to live in poverty. Housing standards are inferior in the poorest 20% of the population with 23% living in non-decent housing. Non-decent housing is defined using several standards including warmth, facilities, state of repair and safety. Physical health was measured on a score between zero and 100 with a higher score indicating better health. Significant differences are seen related to poverty with lower scores in the poorest fifth of the population. Differences between those on higher and lower incomes is greater amongst pensioners. Mental health was also adversely affected by poverty. More anxiety and depression was seen in those from the poorest fifth of the population. This has increased over the last 20 years.

It must be remembered that the methodology used to define poverty in the Rowntree report is different to that of the IMD. However, they do report many results by fifths of the population. The IMD uses ranking and quintiles primarily. These are based on scores of the severity of deprivation but are relative. The report above has demonstrated that poverty is increasing with a higher proportion of people living in poverty. Therefore, one would expect that those in the most deprived IMD quintiles

will be experiencing a greater severity of deprivation and potentially that those in the middle quintile are becoming more deprived. This would not be reflected using IMD quintiles or rank but is likely to lead to an even higher rate of fracture in the most deprived quintiles.

9.12 Further projections and cost

Leicestershire and Rutland has a population of 1,017,697 people with 132,301 people living in the most deprived quintile. Incidence rate was 194 fractures per 100,000 population per year for this most deprived quintile. In a city the size of London with a population of 8,308,400 people of which 2,176,801 live in the most deprived quintile this would equate to 4,219 distal radius fractures from this group alone per year.

LSOA's contain approximately equal numbers of residents. The total population for England is split into quintiles by dividing the total number of LSOA's into five equal groups. The most recent estimates for the population of England show an expected population of people aged 18 and over of 43.5 million people. This equates to roughly 8.7 million people in each quintile. Using the incidence rates generated from Leicestershire there would be 10,353 fractures in the least deprived quintile compared to 16,878 in the most deprived quintile per year in England.

The UK DRAFFT trial and economic analysis provides relevant and up to date information from which projections can be made for distal radius fracture resource use(126, 129). An estimate of fracture fixation rate can be calculated from the DRAFFT screening data ((fixation: [eligible for trial and wanted surgery - 617; Ineligible but required surgery:fracture required opening - 430; open fracture(>Gustilo 1) – 43, 617+430+43=1090) (total adult fractures within 3cm of radiocarpal joint – 5947) (1090/5947=0.18 : 18% fixation rate)). This gives an estimated fixation rate of 18%.

If this fixation rate is applied to the national quintiles and it is estimated that half of the procedures were K-wire fixation and half volar locking plates the costs and surgical time for the least deprived quintile would be £7.2 million and 1,919 hours (*volar* locking plates:0.18* 5176.5*£4,287.89= £3,995,327.27, 0.18*5176.5*70mins= 65,223.9mins=1,087 hrs, K-wires: 0.18*5176.5*£3384.78=£3,153,836.45, 0.18*5176.5*53.57mins=49,910mins= 831.8 hrs).

For the most deprived it would be £11.7 million and 3,128 hours (volar locking plates:0.18*8439*£4,287.89=£6,513,390.67, 0.18*8439*70mins=106,331.4mins= 1,772 hrs, K-wires:0.18* 8439*£3384.78=£5,141,548.6, 0.18*8439 *53.57mins=81,373.9mins= 1,356 hrs). A difference of £4.5 million is seen between the most and least deprived quintiles alone. Using the expected population trend increases identified this would mean that in 10 years time fractures from the least deprived quintile would cost £8 million and £13 million in the most deprived quintile with an increased difference of £5 million. With increasing poverty it is very likely that these costs will in fact be higher with a greater disparity between quintiles.

9.13 Fracture prevention strategies

NICE provide guidance regarding the identification, assessment and treatment of fragility fracture(291). They recommend consideration of assessment of fracture risk in all women aged 65 and over and men aged 75 and over. For women aged 50 to 64 and men aged 50 to 74 fracture risk assessment should be considered if other risk factors are present such as:

Steroid use

Previous fragility fracture

Family history of hip fracture

History of falls

Secondary osteoporosis

Smoking

Low BMI

High alcohol intake

From this thesis deprivation has been shown to be associated with fragility fracture, falls and smoking. Other studies have described a relationship with high alcohol intake. It therefore seems that deprivation is intrinsically linked with many fracture risk factors and they may act as a surrogate for deprivation. Additional work could investigate this further and it may be that deprivation could be included as a risk factor in place of others such as smoking and alcohol.

For those aged under 50 assessment is not recommended unless major risk factors such as previous fragility fracture, early menopause or steroid use are present.

Assessment should take the form of a bone health calculation tool such as FRAX or Qfracture. A 10 year fracture risk of 10% or greater is the approximate threshold for organising further investigation with a DXA scan. For those aged 50 and over with a fragility fracture it is not necessary to use a bone health tool and DXA scan should be arranged. Risk factors for falls should be identified and Calcium intake and vitamin D exposure evaluated. Following a DXA scan with a T score of -2.5 or less bone sparing medication such as bisphosphonate is commenced with Vitamin D supplementation, along with Calcium if intake is inadequate.

Primary prevention before any fracture occurs would be the ideal intervention but with limited resources identifying and treating such a large population is unlikely to be economically viable. 5 to 6 times as many patients would need to be identified, assessed and treated to achieve a similar fracture incidence reduction as that achieved with secondary prevention(292, 293).

Secondary prevention is effective and efficient. Half of patients with a hip fracture have sustained a previous fracture(294). Identifying and treating these patients would prevent an estimated 25% of hip fractures per year. This equates to 20,000 hip fractures in the UK per annum(295). Reduction of hip fracture rate with alendronate treatment has been shown to be significant with secondary prevention but not for primary prevention. Alendronate use following an initial non-vertebral fracture produces a significant relative risk reduction of further fracture. Along with reducing

fracture risk pharmacological therapy has been shown to improve quality of life and reduce mortality(296, 297)

In a Canadian study Majumdar et al randomly allocated distal radius fracture patients to an intervention involving patient education and the provision of evidence based guidelines to their family doctor(298). Compared to the control group osteoporosis treatment rate was three times higher six months following distal radius fracture. Cost effectiveness analysis conducted one year after initial fracture showed that for every 100 patients in the intervention group three fractures, including one hip fracture, were prevented. \$26,800 was saved over a patient's lifetime and 1.1 quality-adjusted life year gained(299).

Fracture liaison services are successful and cost effective. Further fragility fractures can be reduced by 50%(95, 300). The fracture liaison service in Glasgow reported on their management of over 50,000 patients over a 10 year period(301). During this time Glasgow hip fracture rates fell by 7% whereas rates in England increased by 17%. A cost effectiveness analysis showed that for every 1,000 patients treated 18 fractures were prevented, including 11 hip fractures, and £21,000 was saved(111). If comparable services were available throughout the UK £6.3 million could be saved per year. However similar services are currently only available to 42% of the UK.

Ideally a secondary prevention program should consist of osteoporosis assessment and treatment together with a falls risk assessment, in a 'one-stop shop' setting(256). Organising and providing such services is a considerable challenge. A Fracture Liaison Service delivered by a Nurse Specialist is a cost effective, proven approach to the identification, assessment and treatment of fracture risk(80). The National Hip Fracture Database National report shows only 36% of hospitals currently provide this service(302). Presentation with a distal radius fracture provides an opportunity to implement these interventions. In the current economic climate resources are scarce and must be used prudently. Identification of those at highest risk of further fractures allows resources to be targeted to those who need them most.

Targeted outreach services to increase uptake of health services to specific groups have been shown to be effective. Roberts et al reported on a health check service

directed towards key groups including men, people of South Asian ethnicity and those from deprived areas(303). Percentage of health checks taken up amongst South Asians and those in the lowest deprivation quintiles was significantly higher than checks in primary care. They targeted people at venues such as supermarkets, mosques and bus stations. A similar cardiovascular disease risk assessment and management scheme was also able to reach more deprived patients in many areas of England including Leicestershire(304). This could easily be adapted for osteoporosis and falls prevention services.

Chapter 10 Summary

In this thesis I set out to investigate if distal radius fracture is associated with deprivation and if treatment is influenced by deprivation. A further aim was to establish the risk of hip fracture after distal radius fracture. Analysis of local and national data was performed using a variety of modelling techniques. The United Kingdom government official measure of deprivation (IMD2010) was used to investigate the influence of socioeconomic status. A prospective study was performed to identify factors responsible for the effects of deprivation on fracture incidence.

Deprivation was strongly associated with distal radius fracture in the whole population studied. Deprived patients sustained their injuries at an earlier age. Further regression modelling showed important differences between the ethnic groups which were present in both the local and national data. Increasing deprivation was an independent risk factor for distal radius fracture only in white patients. The influence of deprivation was larger in white men. Incidence rate in the least deprived quintile was a third of that in the most deprived for white men and almost half for white women. Age 50 years and over and male gender was an independent risk factor for distal radius fracture in all ethnicities. Falls risk was associated with increasing deprivation.

Deprivation was not related to any of the parameters of hospital care studied. Significant variation has been identified between hospital trusts treating distal radius fractures with a higher fixation rate seen in trusts with a smaller population. This may be exposing patients to unnecessary procedures and utilising excessive healthcare resources. Further robust research is required to provide guidance on which fractures are best treated with surgical fixation. It was identified that patients with more co-morbidities, men and patients admitted at the weekend waited longer for surgery.

Patients who suffer a distal radius fracture are an at-risk group for a subsequent hip fracture. Systematic review and meta-analysis showed that previous distal radius fracture in women was associated with an overall relative risk of hip fracture of 1.43
(CI 1.27 to 1.60) compared to those without. A similar association was demonstrated in studies looking at pooled results for men and women together (RR: 1.82 (CI 1.40 – 2.36)). Results of the four trials providing results for men only did not demonstrate a statistically significant increased relative risk (RR: 2.11 (0.93 – 4.85)). Deprivation measured by similar methods is also a risk factor for hip fracture. Effective secondary prevention after a distal radius fracture is required to prevent the catastrophic event of a hip fracture.

With the increasing size and age of the population the number of patients who sustain a distal radius fracture and require treatment will continue to increase. Poverty is rising which will cause higher levels of social deprivation and an increase in distal radius fracture rate. Fracture prevention strategies are required to combat this rise. Knowing which patients are at highest risk allows interventions to be efficiently targeted. Resources should be targeted to those at-risk patients from deprived areas and preventative strategies put in place.

Chapter 11 Future directions

In this thesis, several important findings have been demonstrated which provide the opportunity to help reduce the risk of distal radius fractures and subsequently hip and other osteoporotic fractures. Deprivation is clearly associated with distal radius fracture incidence. Probably the key result of this work is the identification of the association between increased deprivation and falls risk. It seems this is therefore the area which would be the most effective to target to reduce fracture rate. Information regarding deprivation, along with gender, age and ethnicity, of UK cities and counties is readily available. Areas of high risk can be easily identified. Other health interventions have been delivered direct to those at risk who traditionally do not access health services including those from deprived areas including Leicestershire. These methods have been shown to be effective. An assessment of bone health and falls risk along with education and falls prevention measures could be delivered directly into the communities of at risk patients. Significant funding would be required but it is likely to be effective and prove to be cost-effective overall.

Additional work should also concentrate on modification of the falls prevention programs specifically for deprived patients. These patients are younger and suffer frequent injuries outdoors compared to hip fracture patients. A shift of focus towards outdoor hazards, social support and reasons for outdoor journeys is likely to be beneficial for this group.

Other important deprivation related factors which increase falls risk include depression, polypharmacy and medications causing side effects such as light headedness. Smoking was also associated with deprivation and is known to have a negative effect on bone health. These factors are all amenable to being addressed in primary care. Screening of deprived patients for depression, regular medication reviews and specific smoking cessation advice may all help reduce fracture risk as well as improving general health and well-being.

Other aspects of interest identified include the radiographic differences of bone quality between those of different ethnicity and gender in the 50 and over age group. In this work a significant difference was demonstrated but the numbers for subanalysis were relatively small. Further work could include larger reviews of radiographs or utilising different methods of bone health measurement such as DXA scans to investigate the ethnic differences in more detail.

Gender is another factor which requires further attention. Deprivation has a larger effect on fracture risk in men. Other studies have suggested this is due to lifestyle factors such as smoking, alcohol and risk taking behaviour. A more detailed larger, prospective assessment of lifestyle factors and mechanism of injury of men from the deprived quintiles compared to the least deprived quintiles should help answer this question.

The radiographic analysis demonstrated significant problems with intra and inter observer error for measure of intra-articular parameters. Further studies should try and develop improved measurement techniques. Relying on two points of maximum displacement is difficult and flawed. Methods measuring distance between the articular surfaces of major fragments are likely to be easier and more clinically relevant for long term functional outcome and management decisions.

Research issues identified from the analysis of the influence of deprivation on treatment include variation in fixation rate between hospital trusts, delay to treatment for men, patients with co-morbidities and with presentation at or just before the weekend. As discussed previously there is a lack of high quality studies investigating at which thresholds of displacement intervention should take place for distal radius fractures. Therefore, strong evidence based guidelines are not available leading to individual variation between units. Production of this evidence and subsequent guidelines would almost certainly reduce variation. Trusts and trauma units should aim to develop better pathways for patients with multiple co-morbidities. Appropriate pre-assessment and early anaesthetic team involvement is likely to prevent unexpected delays on the day of surgery due to medical problems.

Appendix 1 Systematic review protocol

PROTOCOL

Systematic Review – What is the risk of a subsequent hip fracture following a distal radius fracture?

Background

Wrist fractures are common and with an ageing population the incidence is rising. 6% of women in the western world will have suffered a distal radius fracture by the time they are 80. Fractures in older patients are likely to occur at lower energy usually because of osteoporosis. A patient who suffers an osteoporotic distal radius fracture is at a higher risk of sustaining a hip fracture, especially within the first year after injury . A hip fracture is a catastrophic event with an associated significant morbidity, mortality and high cost of treatment.

<u>Review Objective</u> – the purpose of this review is to assess the evidence regarding hip fracture after a distal radius fracture and use this to calculate the risk of a subsequent hip fracture occurring

Study Inclusion Criteria

Subjects - Men and women with a distal radius fracture

Exposure – Radiographically confirmed distal radius fracture

Outcomes –Radiographically or surgically confirmed hip fracture. Relative risk, incidence or observed/expected number of outcome events reported

Types of Study – Prospective longitudinal design

Identifying Evidence

Electronic literature review of studies published in full in Pubmed and Embase databases performed by senior clinical librarian. Further studies identified from reference lists of retrieved articles. Thesis search performed. Hand search any relevant journals which are not fully medline linked.

Study Selection and Data Extraction

Initial Screening -

Stage 1: First decision is made based on abstracts assessed against the predetermined inclusion criteria by single reviewer

Stage 2: For studies that appear to meet the inclusion criteria the full paper will be obtained and assessed against the inclusion criteria by 2 independent reviewers. Difference in assessments resolved by consensus with the advisory group.

QualityAssessment – performed using Newcastle Ottawa scale, Coleman methodology tool and domain specific tool. Risk of bias assessment carried out using Cochrane Risk of Bias tool.

Data Extraction

- Study Type, country, study period
- Case definition
- Patients and population- Total no., age, gender, exclusion criteria applied
- Outcomes No. of hip fractures, age, gender, time to fracture, follow up period, loss to follow up
- Results Relative risk, confidence interval, p value, adjustment for variables, total numbers

<u>Data Synthesis</u> – Assessment of homogeneity and if appropriate followed by fixed effects meta-analysis to calculate pooled relative risks with 95% confidence intervals

Dissemination – Presentation at international and national conferences and publication in peer reviewed journals. Emphasis on dissemination to health professionalstreating patients who have suffered a distal radius fracture to enable preventative strategies to be put in place to prevent future hip fracture.

Appendix 2 Systematic review data collection form



Review title or ID	
--------------------	--

Study ID (surname of first author and year first full report of study was published e.g. Smith 2001)

Report IDs of other reports of this study (e.g. duplicate publications, follow-up studies)

Notes:

1... General Information

1.	Date form completed	
	(dd/mm/yyyy)	
2.	Name/ID of person extracting	
	data	
3.	Report title	
(title	of paper/ abstract/ report that	
data	are extracted from)	
4.	Report ID	
(if th	ere are multiple reports of this	
stud	v)	
5.	Reference details	
6.	Report author contact details	
7.	Publication type	
(e.g.	full report, abstract, letter)	
8.	Study funding source	
(inclu	uding role of funders)	
Poss	ible conflicts of interest	
(for s	study authors)	
9.	Notes:	

Data extraction form 2013 08 12

2... Eligibility

Study	Review Inclusion Criteria		Location in text
Characteristics	(Insert inclusion criteria for each characteristic as		(pg & ¶/fig/table)
	defined in the Protocol)	Yes/ No / Unclear	
10. Type of	Randomised trial		
study		•••	
	Non-randomised trial	-	
	Prospective longitudinal study		
	Retrospective cohort		
	Other design (specify):		
11. Participants			
12. Distal Radius			
Fracture –			
how		•••	
diagnosed?			
13. Types of			
outcome			
measures-			
NOF - how			
diagnosed?			
14. Decision:			
15. Reason for			
exclusion			
16. Notes:			

DO NOT PROCEED IF STUDY EXCLUDED FROM REVIEW

3... Population and setting

		1
	Description	Location in text
	Include comparative information for each aroun (i.e. intervention and	(na & ¶/fia/table)
	apprended if available	(pg a 11/)ig/ cable/
	controls) if available	
17. Population		
description		
(from which study		
participants are drawn)		
18. Setting		
(including location and		
social context)		
19. Inclusion criteria		
recorded		
20. Exclusion criteria		
recorded		
21. Method/s of		
recruitment of		
participants		
22. Note		
s:		

4... Methods

	Descriptions as stated in report/paper	Location in text
		(pg & ¶/fig/table)
23. Aim of study		
24. Unit of allocation		
(by individuals, cluster/		
groups or body parts)		
25. Start date		
26. End date		
27. Duration of		
participation		
(from recruitment to last		
follow-up)		
28. Notes:		

5... Risk of Bias assessment

See <u>Chapter 8</u> of the Cochrane Handbook. Additional domains may be required for non-randomised studies.

Domain	Risk of bias	Support for judgement	Location in text
	Low/ High/Unclear		(pg & ¶/fig/table)
29. Random sequence			
generation			
(selection bias)			
30. Allocation concealment			
(selection bias)			

Domain	Risk of bias	Support for judgement	Location in text
	Low/ High/Unclear		(pg & ¶/fig/table)
31. Blinding of participants and personnel (performance bias)		Outcome group: All/	
(if required)		Outcome group:	
32. Blinding of outcome assessment (detection bias)		Outcome group: All/	
(if required)		Outcome group:	
33. Incomplete outcome data (attrition bias)			
34. Selective outcome reporting? (reporting bias)			
35. Other bias – possible confounding factors reported			
36. Notes:			

6... Participants

Provide overall data and, if ava	ilable, comparative data for each intervention or comparison group.	
	Description as stated in report/paper	Location in text
		(pg & ¶/fig/table)
37. Total number		
(or total pop. at start of study		
for NRCTs)		
38. Clusters		
(if applicable, no., type, no.		
people per cluster)		
39. Baseline imbalances		
40. Withdrawals and		
exclusions		
(if not provided below by		
outcome)		
41. Age		
42. Sex		
43. Race/Ethnicity		
44. Co-morbidities		
45. Other treatment		
received		
(additional to study		
intervention)		
46. Other relevant		
sociodemographics		
47. Subgroups measured		
48. Subgroups reported		
49. Note		
s:		

7... Outcomes

Copy and paste table for each outcome.

Outcome 1

	Description as stated in report/paper	Location in text (pg & ¶/fig/table)
50. Outcome name- NOF		
51. Time points reported		
52. Imputation of missing data		
(e.g. assumptions made for ITT analysis)		
53. Assumed risk estimate (e.g. baseline or population risk noted in Background)		
54. Note	'	
5:		

8... Results

For prospective longitudinal/retrospective cohort studies

	Description as stated in re	port/paper	Location in text
rs. Total pariod manufad			(pg & ¶/fig/table)
55. Total period measured			
56. No. participants with			
DRF			
57. Mean age of			
participants with DRF			
(with variance			
measure)			
58. No. participants with			
subsequent NOF#			
59. Mean age of patients			
at time of NOF#			
(with variance measure)			
60. Mean time from DRF to			
subsequent NOF#			
(with variance measure)			
61. Relative risk/Hazard			
ratio reported			
62. Reanalysis required?			
(specify)	Yes/No/Unclear		
63. Reanalysis possible?			
	Yes/No/Unclear		
64. Reanalysed results			
55 Notes:			
05. NOLES.			

9... Applicability

66. Have important		
populations been	Yes/No/Unclear	
excluded from the		
study?		
(consider disadvantaged		
populations, and possible		
differences in the		
intervention effect)		
67. Is the intervention		
likely to be aimed at	Yes/No/Unclear	
disadvantaged groups?		
(e.g. lower socioe conomic		
groups)		
68. Does the study directly		
address the review	Yes/No/Unclear	
question?		
(any issues of partial or		
indirect applicability)		
69. Notes:		

10. Other information

	Description as stated in report/paper	Location in text
70. Key conclusions of study authors		(pg & 11/)19/ tuble/
71. References to other relevant studies		
72. Correspondence required for further study information (what and from whom)		
73. Further study information requested (from whom, what and when)		
74. Correspondence received (from whom, what and when)		
75. Notes:		

Appendix 3 Modified Coleman score assessment

Modified Coleman Methodology Score

Part A Study size no of patients >60 10 41-60 7 20-40 4 0 <20 Mean follow-up (months) >24 5 12-24 2 0 <12 Type of study RCT 15 Prospective cohort 10 Retrospective cohort 0 Diagnostic criterion clinical only 0 (DRF) + imaging 3 7 validated scoring system imaging + validated scoring 10 0 Assessment of severity clinical only Imaging 2 3 Validated scoring system imaging + validated scoring 5 Description of prevention Adequate (stated technique & details) 5 Fair (stated technique only) 3 Inadequate (not stated) 0 Description of management 10 Well described with >80% compliance Well described with 60-80% compliance 5 0 Inadequate description or <60% compliance Part B

Outcome criteria (NOF)	
Outcome measures clearly defined	2
Timing of outcome stated	2

Appendix 4 Risk of bias tool

Risk of bias tool:

Domain	Risk of bias
	Low/ High/Unclear
1. Random sequence generation	
(selection bias)	
2. Allocation concealment	
(selection bias)	
3. Blinding of participants and personnel	
(performance bias)	
(if required)	
Blinding of outcome assessment	
(detection bias)	
(if required)	
Incomplete outcome data	
(attrition bias)	
Selective outcome reporting?	
(reporting bias)	
Other bias – possible confounding factors	
reported	
8. Notes:	

Appendix 5 Fracture risk assessment questionnaire

■ Wrist	Injury Audit
Prospective Wrist Injury Data Collection	Study Ref:
Date of injury:	
Surname:	
Name:	
Age: Wrist injured:	
Right Left Both	
Which is your dominant hand:	
Right Left	Both
Where did you injure your wrist:	
Home Sports ground/pitch	If sports please specify:
Road/Street Other	If other please specify:
Work place	
Did you fall from:	
Standing height Above standing he	ight
Other If other please specify:	
Are you in paid employment/ Yes No self employed:	What is your Occupation:
Bone Health Assessment	
Weight: kg Height:	cm
Have you had a previous fracture	Yes No
Did any of your parents fracture their bin	
De veu smeke	
bu you smoke	
Do you or have you been treated with steroids	in the past
Are you known to have osteoporosis caused by	another medical condition

NJ

1. I have feller in the								Yes	No	
. I have fallen in the past year . I use or have been advised to use a cane or walker to get around safely										
2. I use or have beer	I use or have been advised to use a cane or walker to get around safely									
3. Sometimes I feel u	Sometimes I feel unsteady when I am walking									
4. I steady myself by	holding o	nto furniture	when w	alking at	home					
5. I am worried abou	ut falling									
6. I need to push wit	h my hanc	ls to stand up	o from a	chair						
7. I have some troub	le steppin	g up onto a c	urb							
8. I often have to rus	sh to the to	oilet								
9. I have lost some f	eeling in m	ıy feet								
10. I take medicine that sometimes makes me feel light-headed or more tired than usual										
11. I take medicine	to help me	sleep or imp	orove my	mood						
12. I often feel sad o	r depresse	d								
G. Previous Wrist Pr	oblems									
Did you have any problems with your wrist Yes No before you injured it?										
Please rate the avera injured it on a scale t means severe pain a	age amoun from 0-10. nd stiffnes	t of pain and A zero (0) m s.	l stiffness eans you	s you had did not	l in your have any	wrist in y pain o	the wee r stiffnes	ek BEFOR s and ten	E you (10)	
None	2	$ \begin{array}{c c} & \square \\ 3 & 4 \end{array} $	5	6	7	8	9	Se 10	vere	

NJ

EuroQol EQ-5D

By placing a cross (\boxtimes) in one box in each group below, please indicate which statement best describes your own health state **TODAY.**

Q2 Self Care

Q1 Mobility

	I have no problem walking about	I have no problem with self care
	I have some problem walking about	I have some problem with self care
	l am confined to bed	I am unable to wash or dress myself
Q3	Usual Activities	Q4 Pain/Discomfort
	I have no problem with my normal	I have no pain or discomfort
	activities	I have moderate pain or discomfort
	I have some problem with my normal activities	I have extreme pain or discomfort
	I am unable to perform my normal activities	

Q5 Anxiety/Depression

NJ

I am not anxious or depressed	
l am moderately anxious or depressed	

I am **extremely** anxious or depressed



Your own health state today	Best
Think about how good or bad your own health is today.	imaginable health state 100
This scale may help. The best health you can imagine is marked 100, and the worst health you can imagine is marked 0.	9 0 0
Please indicated, with a cross on the scale, how good or bad your own health is today.	
	<u><u></u> 0</u>
	Worst imaginable health state

NJ

Page 4

In general, how	would you describe your he	ealth:		
Excellent	Very Good	Good	Fair	Poor
Have you been t (Check all that ap	old by a doctor that you ha	ve any of the follow	ing?	
Heart Disease		i.e. angina,	heart attack or heart failure, p	pacemaker etc
High Blood Pressure	e			
Problems caused by	y a Stroke			
Leg Pain when walk	ing due to poor circulation			
Lung Disease		i.e. asthma	, chronic bronchitis or emphys	sema
Kidney Disease				
Diseases of the Ner	vous System	i.e. epileps	y, Parkinson's disease or multi	ple sclerosis
Liver Disease				
Cancer (within the l	ast 5 years)			
Depression				
Rheumatoid Arthrit	is			
Blood Clots in the lu	ungs (PE) or legs (DVT)			
Hepatitis (A, B or C)	or HIV or AIDS			
Kidney problems ne	eeding Dialysis			
Diabetes		If you have	Diabetes, how is it treated?	Diet Controlled
				Tablets
				Insulin
Do you consider	yourself to have a disability	/?		Yes No
, Do you suffer fro	om any other disease for wh	, nich you are having t	treatment?	Yes No
f ves please write d	etails below:			

NJ

Page 5

Agreement: I would be happy to be contacted again in the futu	ire for further research purposes.
Name (please use block capitals)	Signed

NJ

Page 6

Name:

Date:

PATIENT RATED WRIST EVALUATION

_

The questions below will help us understand how much difficulty you have had with your wrist in the past week. You will be describing your <u>average</u> wrist symptoms <u>over the past week</u> on a scale of 0-10. Please provide an answer for ALL questions. If you did not perform an activity, please ESTIMATE the pain or difficulty you would expect. If you have <u>never</u> performed the activity, you may leave it blank.

1. PAIN

Rate the average amount of pain in your wrist over the past week by circling the number that best describes your pain on a scale from 0-10. A zero (0) means that you did not have any pain and a ten (10) means that you had the worst pain you have ever experienced or that you could not do the activity because of pain.

		_		_							
RATE YOUR PAIN: Sample Scale #	No Pain ⁰	1	2	3	4	5	6	7	8	9 V	10 Vorst Ever
At rest	0	1	2	3	4	5	6	7	8	9	10
When doing a task with a repeated wrist movement	0	1	2	3	4	5	6	7	8	9	10
When lifting a heavy object	0	1	2	3	4	5	6	7	8	9	10
When it is at its worst	0	1	2	3	4	5	6	7	8	9	10
How often do you have pain?	0 Never	1	2	3	4	5	6	7	8	9	10 Alwavs

2. FUNCTION

A. SPECIFIC ACTIVITIES Rate the amount of difficulty you experienced performing each of the items listed below - over the past week, by circling the number that describes your difficulty on a scale of 0-10. A zero (0) means you did not experience any difficulty and a ten (10) means it was so difficult you were unable to do it at all.

Sample scale →	0 No Difficulty	, ¹	2	3	4	5	6	7	8	9	10 Unable To Do
Turn a door knob using my affected hand	0	1	2	3	4	5	6	7	8	9	10
Cut meat using a knife in my affected hand	0	1	2	3	4	5	6	7	8	9	10
Fasten buttons on my shirt	0	1	2	3	4	5	6	7	8	9	10
Use my affected hand to push up from a chair	0	1	2	3	4	5	6	7	8	9	10
Carry a 10lb object in my affected hand	0	1	2	3	4	5	6	7	8	9	10
Use bathroom tissue with my affected hand	0	1	2	3	4	5	6	7	8	9	10
B. USUAL ACTIVITIES Rate the amount of difficulty you experienced performing your usual activities in each of the areas listed below, over the past week, by dircing the number that best describes your difficulty on a scale of 0-10. By "usual activities", we mean the activities you performed before you started having a problem with your wrist. A zero (0) means that you did not experience any difficulty and a ten (10) means it was so difficulty ou were unable to do any of your usual activities.											
Personal care activities (dressing, washing)	0	1	2	3	4	5	6	7	8	9	10
Household work (deaning, maintenance)	0	1	2	3	4	5	6	7	8	9	10

reisonal care acevises (di essilig, washing)	0		~	•		•			0	9	10	
Household work (deaning, maintenance)	0	1	2	3	4	5	6	7	8	9	10	
Work (your job or usual everyday work)	0	1	2	3	4	5	6	7	8	9	10	
Recreational activities	0	1	2	3	4	5	6	7	8	9	10	

C JC MacDermid

Appendix 6 STEADI falls risk assessment

Four Things You Can Do to Prevent Falls:

① Speak up.

- Talk openly with your healthcare provider about fall risks and prevention. Ask your doctor or pharmacist to review your medicines.
- ② Keep moving. Begin an exercise program to improve your leg strength and balance.
- ③ Get an annual eye exam. Replace eyeglasses as needed.
- ④ Make your home safer. Remove clutter and tripping hazards.

1 in 4 people 65 and older falls each year.

Falls can lead to a loss of independence, but they are preventable.

Learn More

Contact your local community or senior center for information on exercise, fall prevention programs, and options for improving home safety, or visit:

• go.usa.gov/xN9XA • www.stopfalls.org

For more information, visit www.cdc.gov/s This brochure was produced in collaboration with the following organizations: VA Greater Los Angeles Healthcare System, Geriatric Research Education & Clinical Center (GRECC), and the Fall Prevention Center of Excellence

2017

CDC

Centers for Disease Control and Prevention National Center for Injury Prevention and Control

STEAD opping Elderly eaths & Injuries

Stay

prevention.

Independent

Learn more about fall

Check Your Risk for Falling

Circle "Yes" or "No" for each statement below			Why it matters		
Yes (2)	No (0)	I have fallen in the past year.	People who have fallen once are likely to fall again.		
Yes (2)	No (0)	I use or have been advised to use a cane or walker to get around safely.	People who have been advised to use a cane or walker may already be more likely to fall.		
Yes (1)	No (0)	Sometimes I feel unsteady when I am walking.	Unsteadiness or needing support while walking are signs of poor balance.		
Yes (1)	No (0)	I steady myself by holding onto furniture when walking at home.	This is also a sign of poor balance.		
Yes (1)	No (0)	I am worried about falling.	People who are worried about falling are more likely to fall.		
Yes (1)	No (0)	I need to push with my hands to stand up from a chair.	This is a sign of weak leg muscles, a major reason for falling.		
Yes (1)	No (0)	I have some trouble stepping up onto a curb.	This is also a sign of weak leg muscles.		
Yes (1)	No (0)	I often have to rush to the toilet.	Rushing to the bathroom, especially at night, increases your chance of falling.		
Yes (1)	No (0)	I have lost some feeling in my feet.	Numbness in your feet can cause stumbles and lead to falls.		
Yes (1)	No (0)	I take medicine that sometimes makes me feel light-headed or more tired than usual.	Side effects from medicines can sometimes increase your chance of falling.		
Yes (1)	No (0)	I take medicine to help me sleep or improve my mood.	These medicines can sometimes increase your chance of falling.		
Yes (1)	No (0)	l often feel sad or depressed.	Symptoms of depression, such as not feeling well or feeling slowed down, are linked to falls.		
Total		Add up the number of points for each "yes" answer. Discuss this brochure with your doctor.	If you scored 4 points or more, you may be at risk for falling.		

This checklist was developed by the Greater Los Angeles VA Geriatric Research Education Clinical Center and affiliates and is a validated fall risk self-assessment tool (Rubenstein et al. J Safety Res; 2011: 42(6):493-499). Adapted with permission of the authors.

Appendix 7 R packages

ggplot2	graphs
gplots	plots of group means with confidence intervals
ICC.Sample.Size	ICC power calculation
lme4	linear mixed effects models
MASS	negative binomial regression
meta	meta-analysis
mgcv	generalized additive models
mice	multivariate imputation by chained equations
pbkrtest	Kenward Roger approximation
psych	ICC calculation
sandwich	robust standard errors

Appendix 8 National regression model results

<u>Under 50</u>

IRR of fracture rate for deprivation quintiles are compared with quintile 1 (most deprived), ethnic groups are compared to white patients and male gender with female gender

Factor	IRR	95% Cl	95% Cl	P value
		iower innit	upper minit	
Quintile 2	0.84	0.82	0.86	p<0.001
Quintile 3	0.77	0.75	0.79	p<0.001
Quintile 4	0.81	0.79	0.83	p<0.001
Quintile 5	0.75	0.73	0.77	p<0.001
Asian	0.18	0.17	0.20	p<0.001
Black	0.10	0.09	0.12	p<0.001
Other	0.45	0.42	0.49	p<0.001
Male	1.35	1.33	1.37	p<0.001

50 and over

IRR of fracture rate for the gender/ethnicity deprivation quintiles are compared with quintile 1 (most deprived), ethnic group results for men are compared to results for women of the same ethnic group

Factor	IRR	95% Cl lower limit	95% Cl upper limit	P value
Quintile 2 - White women	0.94	0.92	0.96	0.002
Quintile 3 - White women	0.86	0.84	0.88	p<0.001
Quintile 4 - White women	0.90	0.88	0.92	p<0.001
Quintile 5 - White women	0.90	0.88	0.92	p<0.001
Quintile 2 - Asian women	0.68	0.59	0.77	0.003
Quintile 3 - Asian women	0.80	0.70	0.92	0.112
Quintile 4 - Asian women	0.77	0.66	0.91	0.11
Quintile 5 - Asian women	0.79	0.67	0.93	0.146
Quintile 2 - Black women	1.03	0.78	1.38	0.904
Quintile 3 - Black women	1.81	1.33	2.48	0.057
Quintile 4 - Black women	2.78	1.96	3.94	0.004
Quintile 5 - Black women	4.32	3.08	6.07	p<0.001
Quintile 2 - Other women	1.13	0.95	1.36	0.489
Quintile 3 - Other women	1.67	1.39	1.99	0.004
Quintile 4 - Other women	1.64	1.36	1.98	0.009
Quintile 5 - Other women	2.22	1.86	2.66	p<0.001
Quintile 2 - White men	0.82	0.78	0.86	0.008
Quintile 3 - White men	0.73	0.70	0.77	0.001
Quintile 4 - White men	0.69	0.66	0.73	p<0.001
Quintile 5 - White men	0.72	0.68	0.76	p<0.001
Quintile 2 - Asian men	0.63	0.56	0.72	0.003
Quintile 3 - Asian men	0.69	0.60	0.79	0.112

Quintile 4 - Asian men	0.70	0.59	0.82	0.11
Quintile 5 - Asian men	0.71	0.60	0.84	0.146
Quintile 2 - Black men	0.97	0.73	1.29	0.904
Quintile 3 - Black men	1.56	1.14	2.13	0.057
Quintile 4 - Black men	2.50	1.76	3.55	0.004
Quintile 5 - Black men	3.89	2.77	5.46	p<0.001
Quintile 2 - Other men	0.82	0.68	0.98	0.489
Quintile 3 - Other men	0.73	0.61	0.88	0.004
Quintile 4 - Other men	0.69	0.57	0.84	0.009
Quintile 5 - Other men	0.72	0.60	0.86	p<0.001
White men	0.23	0.23	0.24	p<0.001
Asian men	2.51	2.27	2.79	p<0.001
Black men	2.13	1.69	2.70	0.001
Other men	1.45	1.26	1.67	0.007

Appendix 9	Surgical	fixation	rate in	Leicestershire
------------	----------	----------	---------	----------------

Factor	IRR	95% Cl lower limit	95% Cl upper limit	P value
Quintile 2	1.79	1.43	2.24	0.009
Quintile 3	1.17	0.91	1.51	0.529
Quintile 4	1.07	0.84	1.37	0.775
Quintile 5	1.43	1.15	1.78	0.106
Age under 50	0.57	0.48	0.68	0.001
Male	0.84	0.68	1.03	0.391
White	1.74	1.40	2.15	0.01
Intra-articular DRF	1.56	1.24	1.97	0.057
Injury outdoors	1.57	1.22	2.03	0.075
Quintile 2 - Intra DRF	0.76	0.57	1.01	0.334
Quintile 3 - Intra DRF	1.30	0.96	1.78	0.393
Quintile 4 - Intra DRF	1.61	1.20	2.16	0.105
Quintile 5 - Intra DRF	0.82	0.62	1.08	0.477
Under 50 - Outdoors	2.72	2.22	3.33	p<0.001
Male - Intra DRF	1.51	1.26	1.82	0.024
Male - Outdoors	1.18	0.96	1.43	0.414
White - Outdoors	0.72	0.56	0.92	0.188

References

1. Anonymous. Influence of employments on public health. Lancet. 1843:657-61.

2. Townsend P. Deprivation. J Soc Policy. 1987;16(2):125-46.

3. Booth C. Life and Labour of the People, vol. 1: Williams and Norgate; 1889.

4. Rowntree BS. Poverty: A study of town life: Macmillan; 1901.

5. Townsend P. Poverty in the United Kingdom: a survey of household resources and standards of living: Univ of California Press; 1979.

6. Townsend P. Poverty as relative deprivation: resources and style of living. Wedderburn (ed), Poverty, Inequality and Class Structure. 1974.

7. Mack J, Lansley S. Poor britain: Allen & Unwin London; 1985.

8. Gordon D, Mack J, Lansley S, Main G, Nandy S, Patsios D, et al. The impoverishment of the UK. PSE UK first results. Living standards. 2013.

9. Noble M, Wright G, Smith G, Dibben C. Measuring multiple deprivation at the smallarea level. Environment and planning A. 2006;38(1):169-85.

10. Peake MD. Deprivation, distance and death in lung cancer. Thorax. 2015;70(2):108-9.

11. Riaz SP, Horton M, Kang J, Mak V, Luchtenborg M, Moller H. Lung cancer incidence and survival in England: an analysis by socioeconomic deprivation and urbanization. J Thorac Oncol. 2011;6(12):2005-10.

12. Coleman MP, Babb P, Sloggett A, Quinn M, De Stavola B. Socioeconomic inequalities in cancer survival in England and Wales. Cancer. 2001;91(S1):208-16.

13. Berglund A, Holmberg L, Tishelman C, Wagenius G, Eaker S, Lambe M. Social inequalities in non-small cell lung cancer management and survival: a population-based study in central Sweden. Thorax. 2010;65(4):327-33.

14. Elliss-Brookes L, McPhail S, Ives A, Greenslade M, Shelton J, Hiom S, et al. Routes to diagnosis for cancer–determining the patient journey using multiple routine data sets. Br J Cancer. 2012;107(8):1220.

McPhail S, Elliss-Brookes L, Shelton J, Ives A, Greenslade M, Vernon S, et al.
 Emergency presentation of cancer and short-term mortality. Br J Cancer. 2013;109(8):2027.
 Barakat K, Stevenson S, Wilkinson P, Suliman A, Ranjadayalan K, Timmis AD.

Socioeconomic differentials in recurrent ischaemia and mortality after acute myocardial infarction. Heart. 2001;85(4):390-4.

17. Kaplan GA, Salonen JT. Socioeconomic conditions in childhood and ischaemic heart disease during middle age. BMJ. 1990;301(6761):1121-3.

18. Lynch JW, Kaplan GA, Salonen JT. Why do poor people behave poorly? Variation in adult health behaviours and psychosocial characteristics by stages of the socioeconomic lifecourse. Soc Sci Med. 1997;44(6):809-19.

19. Litonjua AA, Carey VJ, Weiss ST, Gold DR. Race, socioeconomic factors, and area of residence are associated with asthma prevalence. Pediatr Pulmonol. 1999;28(6):394-401.

20. Evans JMM, Newton RW, Ruta DA, MacDonald TM, Morris AD. Socio-economic status, obesity and prevalence of Type 1 and Type 2 diabetes mellitus. Diabet Med. 2000;17(6):478-80.

21. Ellis MP, French JJ, Charnley RM. Acute pancreatitis and the influence of socioeconomic deprivation. Br J Surg. 2009;96(1):74-80.

22. Dowswell T, Towner E. Social deprivation and the prevention of unintentional injury in childhood: a systematic review. Health Educ Res. 2002;17(2):221-37.

23. Dunn L, Henry J, Beard D. Social deprivation and adult head injury: a national study. J Neurol Neurosurg Psychiatry. 2003;74(8):1060-4.

24. Wright J, Kariya A. Aetiology of assault with respect to alcohol, unemployment and social deprivation: a Scottish accident and emergency department case-control study. Injury. 1997;28(5-6):369-72.

25. Tennant A. Admission to hospital following head injury in England: incidence and socio-economic associations. BMC Public Health. 2005;5:21.

26. MacKenzie EJ, Bosse MJ, Kellam JF, Burgess AR, Webb LX, Swiontkowski MF, et al. Characterization of patients with high-energy lower extremity trauma. J Orthop Trauma. 2000;14(7):455-66.

27. Curtis EM, van der Velde R, Moon RJ, van den Bergh JPW, Geusens P, de Vries F, et al. Epidemiology of fractures in the United Kingdom 1988–2012: variation with age, sex, geography, ethnicity and socioeconomic status. Bone. 2016;87:19-26.

28. Cauley JA, Chalhoub D, Kassem AM, Fuleihan GE-H. Geographic and ethnic disparities in osteoporotic fractures. Nature Reviews Endocrinology. 2014;10(6):338-51.

29. Bacon WE, Hadden WC. Occurrence of hip fractures and socioeconomic position. J Aging Health. 2000;12(2):193-203.

30. Benetou V, Orfanos P, Feskanich D, Michalsson K, Pettersson-Kymmer U, Ahmed LA, et al. Education, marital status, and risk of hip fractures in older men and women: the CHANCES project. Osteoporos Int. 2015;26(6):1733-46.

31. Kanis JA, McCloskey EV, Johansson H, Cooper C, Rizzoli R, Reginster JY. European guidance for the diagnosis and management of osteoporosis in postmenopausal women. Osteoporos Int. 2013;24(1):23-57.

32. Brennan SL, Holloway KL, Williams LJ, Kotowicz MA, Bucki-Smith G, Moloney DJ, et al. The social gradient of fractures at any skeletal site in men and women: data from the Geelong Osteoporosis Study Fracture Grid. Osteoporos Int. 2015;26(4):1351-9.

33. Jones S, Johansen A, Brennan J, Butler J, Lyons RA. The effect of socioeconomic deprivation on fracture incidence in the United Kingdom. Osteoporosis international : a journal established as result of cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA. 2004;15(7):520-4.

34. Brennan J, Johansen A, Butler J, Stone M, Richmond P, Jones S, et al. Place of residence and risk of fracture in older people: a population-based study of over 65-year-olds in Cardiff. Osteoporos Int. 2003;14(6):515-9.

35. Aitken SA, Duckworth AD, Clement ND, McQueen MM. The relationship between social deprivation and the incidence of adult fractures. The Journal of Bone & Joint Surgery. 2013;95(6):e32.

36. Court-Brown CM, Brydone A. Social deprivation and adult tibial diaphyseal fractures. Injury. 2007;38(7):750-4.

37. Quah C, Boulton C, Moran C. The influence of socioeconomic status on the incidence, outcome and mortality of fractures of the hip. The Journal of bone and joint surgeryBritish volume. 2011;93(6):801-5.

38. Brennan SL, Pasco JA, Urquhart DM, Oldenburg B, Hanna F, Wluka AE. The association between socioeconomic status and osteoporotic fracture in population-based adults: a systematic review. Osteoporos Int. 2009;20(9):1487-97.

39. Menon MR, Walker JL, Court-Brown CM. The epidemiology of fractures in adolescents with reference to social deprivation. The Journal of bone and joint surgeryBritish volume. 2008;90(11):1482-6.

40. Stark AD, Bennet GC, Stone DH, Chishti P. Association between childhood fractures and poverty: population based study. BMJ. 2002;324(7335):457.

41. Lyons RA, Delahunty AM, Heaven M, McCabe M, Allen H, Nash P. Incidence of childhood fractures in affluent and deprived areas: population based study. BMJ. 2000;320(7228):149.

42. Bridgman S, Wilson R. Epidemiology of femoral fractures in children in the West Midlands region of England 1991 to 2001. Bone & Joint Journal. 2004;86(8):1152-7.

43. Hinton RY, Lincoln A, Crockett MM, Sponseller P, Smith G. Fractures of the femoral shaft in children. Incidence, mechanisms, and sociodemographic risk factors. J Bone Joint Surg Am. 1999;81(4):500-7.

44. Rennie L, Court-Brown CM, Mok JYQ, Beattie TF. The epidemiology of fractures in children. Injury. 2007;38(8):913-22.

45. Kendrick D, Smith S, Sutton AJ, Mulvaney C, Watson M, Coupland C, et al. The effect of education and home safety equipment on childhood thermal injury prevention: meta-analysis and meta-regression. Inj Prev. 2009;15(3):197-204.

46. White D, Raeside R, Barker D. Road accidents and children living in disadvantaged areas: a literature review. The Scottish Executive Central Unit, Edinburgh. 1999.

47. Towner EML, Jarvis SN, Walsh SSM, Aynsley-Green A. Measuring exposure to injury risk in schoolchildren aged 11-14. BMJ. 1994;308(6926):449-52.

48. Odgers CL, Moffitt TE, Tach LM, Sampson RJ, Taylor A, Matthews CL, et al. The protective effects of neighborhood collective efficacy on British children growing up in deprivation: a developmental analysis. Dev Psychol. 2009;45(4):942.

49. Clement ND, Duckworth AD, Wickramasinghe NR, McQueen MM. Does socioeconomic status influence the epidemiology and outcome of distal radial fractures in adults? Eur J Orthop Surg Traumatol. 2017:1-8.

50. Aitken SA, Ralston SH, McQueen MM. The relationship of fall-related fractures to social deprivation. Osteoporos Int. 2011;22(4):1211-8.

51. Ong T, Tan W, Marshall L, Sahota O. The relationship between socioeconomic status and fracture in a fracture clinic setting: Data from the Nottingham Fracture Liaison Service. Injury. 2015;46(2):366-70.

52. Horton TC, Dias JJ, Burke FD. Social deprivation and hand injury. The Journal of hand surgery, European volume. 2007;32(3):256-61.

53. Anakwe RE, Aitken SA, Cowie JG, Middleton SD, Court-Brown CM. The epidemiology of fractures of the hand and the influence of social deprivation. The Journal of hand surgery, European volume. 2011;36(1):62-5.

54. Duckworth AD, Clement ND, Jenkins PJ, Will EM, Court-Brown CM, McQueen MM. Socioeconomic deprivation predicts outcome following radial head and neck fractures. Injury. 2012;43(7):1102-6.

55. Varenna M, Binelli L, Zucchi F, Ghiringhelli D, Gallazzi M, Sinigaglia L. Prevalence of osteoporosis by educational level in a cohort of postmenopausal women. Osteoporos Int. 1999;9(3):236-41.

56. Pearson D, Taylor R, Masud T. The relationship between social deprivation, osteoporosis, and falls. Osteoporosis international : a journal established as result of cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA. 2004;15(2):132-8.

57. Araujo AB, Yang M, Suarez EA, Dagincourt N, Abraham JR, Chiu G, et al. Racial/ethnic and socioeconomic differences in bone loss among men. J Bone Miner Res. 2014;29(12):2552-60.

58. West J, Hippisley-Cox J, Coupland CAC, Price GM, Groom LM, Kendrick D, et al. Do rates of hospital admission for falls and hip fracture in elderly people vary by socio-economic status? Public Health. 2004;118(8):576-81.

59. Gribbin J, Hubbard R, Smith C, Gladman J, Lewis S. Incidence and mortality of falls amongst older people in primary care in the United Kingdom. QJM : monthly journal of the Association of Physicians. 2009;102(7):477-83.

60. Balarajan R, Yuen P, Machin D. Deprivation and general practitioner workload. BMJ. 1992;304(6826):529-34.

61. Fleming D, Crombie D, Cross K. An examination of practice referral rates in relation to practice structure, patient demography and case mix. Health Trends. 1991;23(3):100-4.

62. Haynes R. Inequalities in health and health service use: evidence from the General Household Survey. Soc Sci Med. 1991;33(4):361-8.

63. Benzeval M, Judge K. The determinants of hospital utilisation: implications for resource allocation in England. Health Econ. 1994;3(2):105-16.

64. Majeed FA, Cook DG, Anderson HR, Hilton S, Bunn S, Stones C. Using patient and general practice characteristics to explain variations in cervical smear uptake rates. BMJ. 1994;308(6939):1272-6.

65. McCormick A, Charlton J, Fleming D. Assessing health needs in primary care. Morbidity study from general practice provides another source of information. BMJ: British Medical Journal. 1995;310(6993):1534.

66. Cowling TE, Cecil EV, Soljak MA, Lee JT, Millett C, Majeed A, et al. Access to primary care and visits to emergency departments in England: a cross-sectional, population-based study. PLoS One. 2013;8(6):e66699.

67. Cooper C, Campion G, Melton Lr. Hip fractures in the elderly: a world-wide projection. Osteoporos Int. 1992;2(6):285-9.

68. Cook DJ, Guyatt GH, Adachi JD, Clifton J, Griffith LE, Epstein RS, et al. Quality of life issues in women with vertebral fractures due to osteoporosis. Arthritis & Rheumatology. 1993;36(6):750-6.

69. Ahmed S, Elmantaser M. Secondary osteoporosis. Calcium and Bone Disorders in Children and Adolescents. 16: Karger Publishers; 2009. p. 170-90.

70. Raisz LG. Pathogenesis of osteoporosis: concepts, conflicts, and prospects. The Journal of clinical investigation. 2005;115(12):3318-25.

71. Seeman E, Delmas PD. Bone quality—the material and structural basis of bone strength and fragility. N Engl J Med. 2006;354(21):2250-61.

72. Berntsen GK, Fonnebo V, Tollan A, Sogaard AJ, Magnus JH. Forearm bone mineral density by age in 7,620 men and women: the Tromso study, a population-based study. Am J Epidemiol. 2001;153(5):465-73.

73. Organization WH. Assessment of fracture risk and its application to screening for postmenopausal osteoporosis: report of a WHO study group [meeting held in Rome from 22 to 25 June 1992]. 1994.

74. Van Staa TP, Dennison EM, Leufkens HGM, Cooper C. Epidemiology of fractures in England and Wales. Bone. 2001;29(6):517-22.

75. Lorentzon M, Cummings SR. Osteoporosis: the evolution of a diagnosis. J Intern Med. 2015;277(6):650-61.

76. British Orthopaedic A. The care of patients with fragility fracture. London: British Orthopaedic Association. 2007:8-11.

77. Mauck KF, Clarke BL, editors. Diagnosis, screening, prevention, and treatment of osteoporosis. Mayo Clin Proc; 2006: Elsevier.

78. Hernlund E, Svedbom A, Ivergard M, Compston J, Cooper C, Stenmark J, et al. Osteoporosis in the European Union: medical management, epidemiology and economic burden. A report prepared in collaboration with the International Osteoporosis Foundation (IOF) and the European Federation of Pharmaceutical Industry Associations (EFPIA). Arch Osteoporos. 2013;8:136. 79. Clunie G, Stephenson S. Implementing and running a fracture liaison service: an integrated clinical service providing a comprehensive bone health assessment at the point of fracture management. Journal of Orthopaedic Nursing. 2008;12(3):159-65.

Department of H. Falls and Fractures: Effective interventions in health and social care.
 2009.

81. National Osteoporosis S. 25th anniversary report - A fragile future. 2011.

82. Leal J, Gray AM, Prieto-Alhambra D, Arden NK, Cooper C, Javaid MK, et al. Impact of hip fracture on hospital care costs: a population-based study. Osteoporos Int. 2016;27(2):549-58.

83. Osnes E, Lofthus C, Meyer H, Falch J, Nordsletten L, Cappelen I, et al. Consequences of hip fracture on activities of daily life and residential needs. Osteoporos Int. 2004;15(7):567-74.
84. National Hip Fracture Database extended report 2014 2014 [Available from:

http://www.nhfd.co.uk/20/hipfractureR.nsf/vwcontent/2014reportPDFs/\$file/NHFD2014ExtendedReport.pdf?

85. Scaf-Klomp W, Van Sonderen E, Sanderman R, Ormel J, Kempen GI. Recovery of physical function after limb injuries in independent older people living at home. Age Ageing. 2001;30(3):213-9.

86. Owen RA, Melton LJ, 3rd, Ilstrup DM, Johnson KA, Riggs BL. Colles' fracture and subsequent hip fracture risk. Clin Orthop Relat Res. 1982(171):37-43.

87. Robinson CM, Royds M, Abraham A, McQueen MM, Court-Brown CM, Christie J. Refractures in patients at least forty-five years old. a prospective analysis of twenty-two thousand and sixty patients. Journal of Bone & Joint Surgery - American Volume. 2002;84-A(9):1528-33.

88. Porthouse J, Birks YF, Torgerson DJ, Cockayne S, Puffer S, Watt I. Risk factors for fracture in a UK population: a prospective cohort study. QJM : monthly journal of the Association of Physicians. 2004;97(9):569-74.

89. Gehlbach S, Saag KG, Flahive J, Anderson FA, Hooven F, Investigators G. Previous fractures increase risk for subsequent fractures at multiple sites: global longitudinal study of osteoporosis in women. Bone. 2010;47:S180.

90. Cuddihy MT, Gabriel SE, Crowson CS, O'Fallon WM, Melton LJ, 3rd. Forearm fractures as predictors of subsequent osteoporotic fractures. Osteoporos Int. 1999;9(6):469-75.

91. Crandall CJ, Hovey KM, Cauley JA, Andrews CA, Curtis JR, Wactawski-Wende J, et al. Wrist Fracture and Risk of Subsequent Fracture: Findings from the Women's Health Initiative Study. J Bone Miner Res. 2015.

92. Haentjens P, Autier P, Collins J, Velkeniers B, Vanderschueren D, Boonen S. Colles fracture, spine fracture, and subsequent risk of hip fracture in men and women. A metaanalysis. Journal of Bone & Joint Surgery - American Volume. 2003;85-A(10):1936-43.

93. Chen CW, Huang TL, Su LT, Kuo YC, Wu SC, Li CY, et al. Incidence of subsequent hip fractures is significantly increased within the first month after distal radius fracture in patients older than 60 years. The Journal of Trauma and Acute Care Surgery. 2013;74(1):317-21.

94. Lauritzen JB, Schwarz P, McNair P, Lund B, Transbol I. Radial and humeral fractures as predictors of subsequent hip, radial or humeral fractures in women, and their seasonal variation. Osteoporos Int. 1993;3(3):133-7.

95. Cranney A, Guyatt G, Krolicki N, Welch V, Griffith L, Adachi JD, et al. A meta-analysis of etidronate for the treatment of postmenopausal osteoporosis. Osteoporos Int. 2001;12(2):140-51.

96. Ettinger B, Black DM, Mitlak BH, Knickerbocker RK, Nickelsen T, Genant HK, et al. Reduction of vertebral fracture risk in postmenopausal women with osteoporosis treated with raloxifene: results from a 3-year randomized clinical trial. JAMA. 1999;282(7):637-45. 97. Larsen ER, Mosekilde L, Foldspang A. Vitamin D and calcium supplementation prevents osteoporotic fractures in elderly community dwelling residents: a pragmatic population-based 3-year intervention study. J Bone Miner Res. 2004;19(3):370-8.

98. O'Neill TW, Cooper C, Finn JD, Lunt M, Purdie D, Reid DM, et al. Incidence of distal forearm fracture in British men and women. Osteoporosis international : a journal established as result of cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA. 2001;12(7):555-8.

99. Burge RT, Worley D, Johansen A, Bhattacharyya S, Bose U. The Cost of osteoporotic fractures in the UK: Projections for 2000-2020. J Med Econ. 2001;4(1-4):51-62.

100. Shauver MJ, Yin H, Banerjee M, Chung KC. Current and future national costs to medicare for the treatment of distal radius fracture in the elderly. The Journal of hand surgery. 2011;36(8):1282-7.

101. Donaldson LJ, Cook A, Thomson RG. Incidence of fractures in a geographically defined population. J Epidemiol Community Health. 1990;44(3):241-5.

102. Court-Brown CM, Caesar B. Epidemiology of adult fractures: a review. Injury. 2006;37(8):691-7.

103. Baron JA, Karagas M, Barrett J, Kniffin W, Malenka D, Mayor M, et al. Basic epidemiology of fractures of the upper and lower limb among Americans over 65 years of age. Epidemiology. 1996;7(6):612-8.

104. Diamantopoulos AP, Rohde G, Johnsrud I, Skoie IM, Hochberg M, Haugeberg G. The epidemiology of low-and high-energy distal radius fracture in middle-aged and elderly men and women in Southern Norway. PLoS One. 2012;7(8):e43367.

105. Bentohami A, Bosma J, Akkersdijk GJM, van Dijkman B, Goslings JC, Schep NWL. Incidence and characteristics of distal radial fractures in an urban population in The Netherlands. Eur J Trauma Emerg Surg. 2014;40(3):357-61.

106. Flinkkil T, Sirni K, Hippi M, Hartonen S, Ruuhela R, Ohtonen P, et al. Epidemiology and seasonal variation of distal radius fractures in Oulu, Finland. Osteoporos Int. 2011;22(8):2307-12.

107. Brogren E, Petranek M, Atroshi I. Incidence and characteristics of distal radius fractures in a southern Swedish region. BMC Musculoskelet Disord. 2007;8(1):48.

108. Thompson PW, Taylor J, Dawson A. The annual incidence and seasonal variation of fractures of the distal radius in men and women over 25 years in Dorset, UK. Injury. 2004;35(5):462-6.

109. Jaglal SB, Weller I, Mamdani M, Hawker G, Kreder H, Jaakkimainen L, et al. Population trends in BMD testing, treatment, and hip and wrist fracture rates: are the hip fracture projections wrong? J Bone Miner Res. 2005;20(6):898-905.

110. Lofthus CM, Frihagen F, Meyer HE, Nordsletten L, Melhuus K, Falch JA. Epidemiology of distal forearm fractures in Oslo, Norway. Osteoporos Int. 2008;19(6):781-6.

111. McLellan AR, Wolowacz SE, Zimovetz EA, Beard SM, Lock S, McCrink L, et al. Fracture liaison services for the evaluation and management of patients with osteoporotic fracture: a cost-effectiveness evaluation based on data collected over 8 years of service provision. Osteoporos Int. 2011;22(7):2083.

112. Bhattacharyya T, Millham FH. Relationship between weather and seasonal factors and trauma admission volume at a Level I trauma center. Journal of Trauma and Acute Care Surgery. 2001;51(1):118-22.

113. Gill M, Goldacre MJ. Seasonal variation in hospital admission for road traffic injuries in England: analysis of hospital statistics. Injury prevention : journal of the International Society for Child and Adolescent Injury Prevention. 2009;15(6):374-8.

114. Falch JA. Epidemiology of fractures of the distal forearm in Oslo, Norway. Acta Orthop. 1983;54(2):291-5.

115. Wareham K, Johansen A, Stone MD, Saunders J, Jones S, Lyons RA. Seasonal variation in the incidence of wrist and forearm fractures, and its consequences. Injury. 2003;34(3):219-22.

116. Jacobsen SJ, Sargent DJ, Atkinson EJ, O'Fallon WM, Melton Iii LJ. Contribution of weather to the seasonality of distal forearm fractures: a population-based study in Rochester, Minnesota. Osteoporos Int. 1999;9(3):254-9.

117. Cooper C, Atkinson EJ, Jacobsen SJ, O'Fallon WM, Melton IIILJ. Population-based study of survival after osteoporotic fractures. Am J Epidemiol. 1993;137(9):1001-5.

118. Campbell AJ, Spears GFS, Borrie MJ, Fitzgerald JL. Falls, elderly women and the cold. Gerontology. 1988;34(4):205-8.

119. Hull P, Baraza N, Whalley H, Brewster M, Costa M. Dorsally displaced fractures of the distal radius—a study of preferred treatment options among UK trauma and orthopaedic surgeons. Hand Surg. 2010;15(03):185-91.

120. Dias J, Wray C, Jones J, Gregg P. The value of early mobilisation in the treatment of Colles' fractures. Bone & Joint Journal. 1987;69(3):463-7.

121. National Institute of Clinical Excellence. Fractures (non-complex):assessment and management Fractures: diagnosis, management and follow-up of fracturesNICE Guideline NG38 2016 [Available from: <u>https://www.nice.org.uk/guidance/ng38/evidence/full-guideline-pdf-2358460765</u>.

122. Leung F, Zhu L, Ho H, Lu W, Chow S. Palmar plate fixation of AO type C2 fracture of distal radius using a locking compression plate—a biomechanical study in a cadaveric model. J Hand Surg Am. 2003;28(3):263-6.

123. Obert L, Rey P-B, Uhring J, Gasse N, Rochet S, Lepage D, et al. Fixation of distal radius fractures in adults: a review. Orthopaedics & Traumatology: Surgery & Research. 2013;99(2):216-34.

124. Koval KJ, Harrast JJ, Anglen JO, Weinstein JN. Fractures of the distal part of the radius. The evolution of practice over time. Where's the evidence? The Journal of bone and joint surgeryAmerican volume. 2008;90(9):1855-61.

125. Handoll HH, Huntley JS, Madhok R. Different methods of external fixation for treating distal radial fractures in adults. The Cochrane Library. 2008.

126. Costa ML, Achten J, Plant C, Parsons NR, Rangan A, Tubeuf S, et al. UK DRAFFT: a randomised controlled trial of percutaneous fixation with Kirschner wires versus volar locking-plate fixation in the treatment of adult patients with a dorsally displaced fracture of the distal radius. Health technology assessment (Winchester, England). 2015;19(17):vi.

127. Costa ML, Jameson SS, Reed MR. Do large pragmatic randomised trials change clinical practice?: assessing the impact of the Distal Radius Acute Fracture Fixation Trial (DRAFFT). The bone & joint journal. 2016;98-B(3):410-3.

128. Goehre F, Otto W, Schwan S, Mendel T, Vergroesen PP, Lindemann-Sperfeld L. Comparison of palmar fixed-angle plate fixation with K-wire fixation of distal radius fractures (AO A2, A3, C1) in elderly patients. The Journal of hand surgery, European volume. 2014;39(3):249-57.

129. Tubeuf S, Yu G, Achten J, Parsons NR, Rangan A, Lamb SE, et al. Cost effectiveness of treatment with percutaneous Kirschner wires versus volar locking plate for adult patients with a dorsally displaced fracture of the distal radius: analysis from the DRAFFT trial. The bone & joint journal. 2015;97-B(8):1082-9.

130. Kreder HJ, Hanel DP, McKee M, Jupiter J, McGillivary G, Swiontkowski MF. Consistency of AO fracture classification for the distal radius. J Bone Joint Surg Br. 1996;78(5):726-31.
131. Johnson P, Szabo R. Angle measurements of the distal radius: a cadaver study. Skeletal

Radiol. 1993;22(4):243-6.

132. Salling N, Andreasen TS, Foldager-Jensen AD, Tromborg H, Srensen AS, Krabsen K, et al. National clinical guideline on the treatment of distal radial fractures. 2016.

133. Lichtman DM, Bindra RR, Boyer MI, Putnam MD, Ring D, Slutsky DJ, et al. Treatment of distal radius fractures. J Am Acad Orthop Surg. 2010;18(3):180-9.

134. Brogren E, Hofer M, Petranek M, Wagner P, Dahlin LB, Atroshi I. Relationship between distal radius fracture malunion and arm-related disability: a prospective population-based cohort study with 1-year follow-up. BMC Musculoskelet Disord. 2011;12(1):9.

135. McQueen MM, Hajducka C. Redisplaced unstable fractures of the distal radius. Bone & Joint Journal. 1996;78(3):404-9.

136. Wilcke MKT, Abbaszadegan H, Adolphson PY. Patient-perceived outcome after displaced distal radius fractures: a comparison between radiological parameters, objective physical variables, and the DASH score. J Hand Ther. 2007;20(4):290-9.

137. Finsen V, Rod O, Rød K, Rajabi B, Alm-Paulsen PS, Russwurm H. The relationship between displacement and clinical outcome after distal radius (Colles') fracture. Journal of Hand Surgery (European Volume). 2013;38(2):116-26.

138. Chung KC, Kotsis SV, Kim HM. Predictors of functional outcomes after surgical treatment of distal radius fractures. The Journal of hand surgery. 2007;32(1):76-83.

139. Plant CE, Parsons NR, Costa ML. Do radiological and functional outcomes correlate for fractures of the distal radius? Bone Joint J. 2017;99(3):376-82.

140. Grewal R, MacDermid JC. The risk of adverse outcomes in extra-articular distal radius fractures is increased with malalignment in patients of all ages but mitigated in older patients. The Journal of hand surgery. 2007;32(7):962-70.

141. Knirk JL, Jupiter JB. Intra-articular fractures of the distal end of the radius in young adults. J Bone Joint Surg Am. 1986;68(5):647-59.

142. Catalano LW, Cole RJ, Gelberman RH, Evanoff BA, Gilula LA, Borrelli J. Displaced intraarticular fractures of the distal aspect of the radius. Long-term results in young adults after open reduction and internal fixation. J Bone Joint Surg Am. 1997;79(9):1290-302.

143. Goldfarb CA, Rudzki JR, Catalano LW, Hughes M, Borrelli J. Fifteen-year outcome of displaced intra-articular fractures of the distal radius. The Journal of hand surgery. 2006;31(4):633-9.

144. Wagner WF, Tencer AF, Kiser P, Trumble TE. Effects of intra-articular distal radius depression on wrist joint contact characteristics. The Journal of hand surgery. 1996;21(4):554-60.

145. Trumble TE, Schmitt SR, Vedder NB. Factors affecting functional outcome of displaced intra-articular distal radius fractures. The Journal of hand surgery. 1994;19(2):325-40.

146. Kreder HJ, Hanel DP, Agel J, McKee M, Schemitsch EH, Trumble TE, et al. Indirect reduction and percutaneous fixation versus open reduction and internal fixation for displaced intra-articular fractures of the distal radius. Bone & Joint Journal. 2005;87(6):829-36.

147. Forward DP, Davis TR, Sithole JS. Do young patients with malunited fractures of the distal radius inevitably develop symptomatic post-traumatic osteoarthritis? The Journal of bone and joint surgeryBritish volume. 2008;90(5):629-37.

148. Kopylov P, Johnell O, Redlund-Johnell I, Bengner U. Fractures of the distal end of the radius in young adults: a 30-year follow-up. The Journal of Hand Surgery: British & European Volume. 1993;18(1):45-9.

149. Fanuele J, Koval KJ, Lurie J, Zhou W, Tosteson A, Ring D. Distal radial fracture treatment: what you get may depend on your age and address. The Journal of Bone and Joint SurgeryAmerican volume. 2009;91(6):1313.

150. Chung KC, Shauver MJ, Yin H, Kim HM, Baser O, Birkmeyer JD. Variations in the use of internal fixation for distal radial fracture in the United States medicare population. The Journal of Bone and Joint SurgeryAmerican volume. 2011;93(23):2154.

151. Walenkamp MMJ, Mulders MAM, Goslings JC, Westert GP, Schep NWL. Analysis of variation in the surgical treatment of patients with distal radial fractures in the Netherlands. Journal of Hand Surgery (European Volume). 2017;42(1):39-44.
152. Care NHSR. NHS atlas of variation in healthcare. Reducing unwarranted variation to increase value and improve quality. 2011.

153. Donabedian A. Evaluating the quality of medical care. The Milbank memorial fund quarterly. 1966;44(3):166-206.

154. Birkmeyer JD, Dimick JB, Birkmeyer NJ. Measuring the quality of surgical care: structure, process, or outcomes? J Am Coll Surg. 2004;198(4):626-32.

155. Support UDSC. InFuse 2011 2011 [Available from: <u>http://infusecp.mimas.ac.uk/</u>.
156. The England Indices of Deprivation 2015 [Available from:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/465791/Eng lish_Indices_of_Deprivation_2015_-_Statistical_Release.pdf.

157. National Office of Statistics. Population Estimates 2017 [Available from:

https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/populationestimates - datasets.

158. McLennan D BH, Noble M, Davies J, Garrat E, Dibben C. The English Indices of Deprivation 2010 2011 [Available from:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/6871/1871 208.pdf.

159. Ng CY, McQueen MM. What are the radiological predictors of functional outcome following fractures of the distal radius? Bone & Joint Journal. 2011;93(2):145-50.

160. Cole RJ, Bindra RR, Evanoff BA, Gilula LA, Yamaguchi K, Gelberman RH. Radiographic evaluation of osseous displacement following intra-articular fractures of the distal radius: reliability of plain radiography versus computed tomography. The Journal of hand surgery. 1997;22(5):792-800.

161. Plant CE, Hickson C, Hedley H, Parsons NR, Costa ML. Is it time to revisit the AO classification of fractures of the distal radius? Bone Joint J. 2015;97(6):818-23.

162. ELDRIDGE R. The metacarpal index: a useful aid in the diagnosis of the Marfan syndrome. Arch Intern Med. 1964;113(2):248-54.

163. Barnett E, Nordin B. The radiological diagnosis of osteoporosis: a new approach. Clin Radiol. 1960;11(3):166-74.

164. Haara M, Heliovaara M, Impivaara O, Arokoski JP, Manninen P, Knekt P, et al. Low metacarpal index predicts hip fracture: a prospective population study of 3,561 subjects with 15 years of follow-up. Acta Orthop. 2006;77(1):9-14.

165. Bouxsein ML, Palermo L, Yeung C, Black DM. Digital X-ray radiogrammetry predicts hip, wrist and vertebral fracture risk in elderly women: a prospective analysis from the study of osteoporotic fractures. Osteoporosis international : a journal established as result of cooperation between the European Foundation for Osteoporosis and the National Osteoporosis Foundation of the USA. 2002;13(5):358-65.

166. Lu Y, Genant HK, Shepherd J, Zhao S, Mathur A, Fuerst TP, et al. Classification of osteoporosis based on bone mineral densities. J Bone Miner Res. 2001;16(5):901-10.

167. Mokkink LB, Terwee CB, Patrick DL, Alonso J, Stratford PW, Knol DL, et al. The COSMIN checklist for assessing the methodological quality of studies on measurement properties of health status measurement instruments: an international Delphi study. Qual Life Res. 2010;19(4):539-49.

168. Zou G. Sample size formulas for estimating intraclass correlation coefficients with precision and assurance. Stat Med. 2012;31(29):3972-81.

169. Organisation WH. International Statistical Classification of Diseases and Related Health Problems 10th Revision 2010 [Available from:

http://apps.who.int/classifications/icd10/browse/2010/en.

170. Digital N. OPCS Classification of Interventions and Procedures 2016 [Available from: http://content.digital.nhs.uk/isce/publication/scci0084. 171. Office M. MIDAS: UK Daily Temperature Data 2006 [Available from:

http://catalogue.ceda.ac.uk/uuid/1bb479d3b1e38c339adb9c82c15579d8.

172. Digital N. Supplementary files 2014 [Available from:

http://content.digital.nhs.uk/article/5317/2014-Supplementary-information-files.

173. Charlson M, Szatrowski TP, Peterson J, Gold J. Validation of a combined comorbidity index. J Clin Epidemiol. 1994;47(11):1245-51.

174. MacDermid JC, Turgeon T, Richards RS, Beadle M, Roth JH. Patient rating of wrist pain and disability: a reliable and valid measurement tool. J Orthop Trauma. 1998;12(8):577-86.

175. R B. EuroQol: the current state of play. Pubmed: Euroquol Brooks.

176. Kanis J, Johnell O, Odén A, Johansson H, McCloskey E. FRAX[™] and the assessment of fracture probability in men and women from the UK. Osteoporos Int. 2008;19(4):385-97.
177. Excellence NIfHaC. Falls in older people Quality standard [QS86] 2015 [Available from: https://www.nice.org.uk/guidance/qs86.

178. Perell KL, Nelson A, Goldman RL, Luther SL, Prieto-Lewis N, Rubenstein LZ. Fall risk assessment measures: an analytic review. The Journals of Gerontology Series A: Biological Sciences and Medical Sciences. 2001;56(12):M761-M6.

179. Rubenstein LZ, Vivrette R, Harker JO, Stevens JA, Kramer BJ. Validating an evidencebased, self-rated fall risk questionnaire (FRQ) for older adults. J Safety Res. 2011;42(6):493-9.

180. Aging ODo. STEADY U Ohio 2017 [Available from: <u>http://aging.ohio.gov/steadyu</u>.

181. Prevention CfDCa. STEADI (Stopping elderly accidents, deaths and injuries) 2017 [Available from: <u>https://www.cdc.gov/steadi/pdf/Stay_Independent_brochure-a.pdf</u>.

182. McCullagh P. Nelder. JA (1989), Generalized Linear Models. CRC Monographs on Statistics & Applied Probability, Springer Verlag, New York. 1973.

183. von Bortkiewicz L. Das gesetz der kleinen zahlen: BG Teubner; 1898.

184. Cameron AC, Trivedi PK. Regression analysis of count data: Cambridge university press; 2013.

185. Judd CM, Westfall J, Kenny DA. Treating stimuli as a random factor in social psychology: A new and comprehensive solution to a pervasive but largely ignored problem. J Pers Soc Psychol. 2012;103(1):54.

186. Hastie T, Tibshirani R. Generalized additive models: some applications. Journal of the American Statistical Association. 1987;82(398):371-86.

187. Finsen V, Benum P. Colles' fracture as an indicator of increased risk of hip fracture. An epidemiological study. Ann Chir Gynaecol. 1987;76(2):114-8.

188. Osteosynthese International 2011 Annual Meeting. Injury. 2011;42.

189. Adami S, Libanati C, Boonen S, Cummings SR, Ho PR, Wang A, et al. Denosumab treatment in postmenopausal women with osteoporosis does not interfere with fracture-healing: Results from the FREEDOM trial. Journal of Bone and Joint Surgery - Series A. 2012;94(23):2113-9.

190. Ahmed LA, Emaus N, Berntsen GK, Bjornerem A, Fonnebo V, Jorgensen L, et al. Bone loss and the risk of non-vertebral fractures in women and men: The tromso study. Osteoporos Int. 2010;21(9):1503-11.

191. Astrand J, Nilsson J, Thorngren KG. Screening for osteoporosis reduced new fracture incidence by almost half: A 6-year follow-up of 592 fracture patients from an osteoporosis screening program. Acta Orthop. 2012;83(6):661-5.

192. Barrett-Connor E, Siris ES, Wehren LE, Miller PD, Abbott TA, Berger ML, et al. Osteoporosis and fracture risk in women of different ethnic groups. J Bone Miner Res. 2005;20(2):185-94.

193. Barrett-Connor E, Nielson CM, Orwoll E, Bauer DC, Cauley JA, Osteoporotic FG. Epidemiology of rib fractures in older men: Osteoporotic Fractures in Men (MrOS) prospective cohort study. BMJ. 2010;340. 194. Barrett-Connor E, Nielson C, Ensrud K, Orwoll E. Wrist fracture incidence, risk factors, and associations with subsequent fractures in older men. J Bone Miner Res. 2012;27.

195. Bauer DC, Gluer CC, Cauley JA, Vogt TM, Ensrud KE, Genant HK, et al. Broadband ultrasound attenuation predicts fractures strongly and independently of densitometry in older women: A prospective study. Arch Intern Med. 1997;157(6):629-34.

196. Bensen R, Adachi JD, Papaioannou A, Ioannidis G, Olszynski WP, Sebaldt RJ, et al. Evaluation of easily measured risk factors in the prediction of osteoporotic fractures. BMC Musculoskelet Disord. 2005;6.

197. Bjornerem A, Ahmed LA, Jorgensen L, Stormer J, Joakimsen RM. Breastfeeding protects against hip fracture in postmenopausal women: The Tromso study. J Bone Miner Res. 2011;26(12):2843-50.

198. Black DM, Arden NK, Palermo L, Pearson J, Cummings SR. Prevalent vertebral deformities predict hip fractures and new vertebral deformities but not wrist fractures. J Bone Miner Res. 1999;14(5):821-8.

199. Black DM, Palermo L, Woo C, Ensrud K, Cummings SR, Hochberg M, et al. Use of fracture history to predict long term risk of specific fracture types: Is any non-vertebral fracture history an adequate predictor? Bone. 2009;44.

200. Bliuc D, Nguyen TV, Eisman JA, Center JR. The impact of nonhip nonvertebral fractures inn elderly women and men. J Clin Endocrinol Metab. 2014;99(2):415-23.

201. Bohannon AD, Hanlon JT, Landerman R, Gold DT. Association of race and other potential risk factors with nonvertebral fractures in community-dwelling elderly women. Am J Epidemiol. 1999;149(11):1002-9.

202. Byrne C, O'Connor M, Spencer P, Duggan J. Catching a break: Identifying patients at risk of further fracture. Ir J Med Sci. 2013;182.

203. Cabana F, Roux S, Beaulieu MC, Beaulieu M, Boire G. Incidence of subsequent fractures according to site of initial non-vertebral fractures and treatment status: The first 3 years of the optimus initiative. Annals of the Rheumatic Disease. 2013;71.

204. Curtis JR, Arora T, Matthews RS, Taylor A, Becker DJ, Colon-Emeric C, et al. Is withholding osteoporosis medication after fracture sometimes rational? A Comparison of the risk for second fracture versus death. J Am Med Dir Assoc. 2010;11(8):584-91.

205. Dholakia S, Thilagarajah M, Singh R. Are wrist fractures still a useful predictor of future hip fracture in 2011? Osteoporos Int. 2011;22:S655.

206. Endres HG, Dasch B, Maier C, Lungenhausen M, Smektala R, Trampisch HJ, et al. Diagnosis and treatment of osteoporosis in postmenopausal women with distal radius fracture in Germany. Curr Med Res Opin. 2007;23(9):2171-81.

207. Ettinger B, Ray GT, Pressman AR, Gluck O. Limb fractures in elderly men as indicators of subsequent fracture risk. Arch Intern Med. 2003;163(22):2741-7.

208. Haentjens P, Johnell O, Kanis JA, Bouillon R, Cooper C, Lamraski G, et al. Evidence from data searches and life-table analyses for gender-related differences in absolute risk of hip fracture after Colles' or spine fracture: Colles' fracture as an early and sensitive marker of skeletal fragility in white men. J Bone Miner Res. 2004;19(12):1933-44.

209. Hodsman AB, Leslie WD, Tsang JF, Gamble GD. 10-Year probability of recurrent fractures following wrist and other osteoporotic fractures in a large clinical cohort: An analysis from the Manitoba bone density program. Arch Intern Med. 2008;168(20):2261-7.

210. Huang TL, Chen CW. Colles' fracture as a risk factor for subsequent hip fractures: An Asian population based study. Osteoporos Int. 2013;24.

211. Lesnyak O, Ershova O, Belova K, Gladkova E, Sinitsina O, Ganert O, et al. Epidemiology of fracture in the Russian Federation and the development of a FRAX model. Archives of Osteoporosis. 2012;7(1-2):67-73.

212. Luthje P, Helkamaa T, Kaukonen JP, Nurmi-Luthje I, Kataja M. A long-term follow-up of 221 hip fracture patients in southeastern Finland: Analysis of survival and prior or subsequent fractures. Arch Gerontol Geriatr. 2012;54(3):e299.

213. Minicelli MG, Pistillo P, Cerocchi I, Baldi J, Meglio GBD, Tarantino U. Should awrist fracture and the number of previous fractures indicate higher priority for osteoporosis investigation? Osteoporos Int. 2011;22:S378.

214. Oyen J, Gjesdal CG, Brudvik C, Hove LM, Apalset EM, Gulseth HC, et al. Low-energy distal radius fractures in middle-aged and elderly men and women--the burden of osteoporosis and fracture risk : A study of 1794 consecutive patients. Osteoporos Int. 2010;21(7):1257-67.

215. Ruan WD, Wang P, Ma XL, Ge RP, Zhou XH. Analysis on the risk factors of second fracture in osteoporosis-related fractures. Chinese Journal of Traumatology - English Edition. 2011;14(2):74-8.

216. Rudenka EV, Romanov GN, Rudenka AV, Adamenka A. Fractures in patients with osteoporosis. Osteoporos Int. 2011;22.

217. Runge M. Diagnosing fall risk in the elderly. Ther Umsch. 2002;59(7):351-8.

218. Sakuma M, Endo N, Oinuma T, Miyasaka D, Oguma Y, Imao K, et al. Incidence of osteoporotic fractures in Sado, Japan in 2010. J Bone Miner Metab. 2014;32(2):200-5.

219. Schousboe JT, Fink HA, Taylor BC, Stone KL, Hillier TA, Nevitt MC, et al. Association between self-reported prior wrist fractures and risk of subsequent hip and radiographic vertebral fractures in older women: a prospective study. J Bone Miner Res. 2005;20(1):100-6.

220. Smith MG, Dunkow P, Lang DM. Treatment of osteoporosis: missed opportunities in the hospital fracture clinic. Ann R Coll Surg Engl. 2004;86(5):344-6.

221. Song X, Shi N, Badamgarav E, Kallich J, Varker H, Lenhart G, et al. Cost burden of second fracture in the US Health System. Bone. 2011;48(4):828-36.

222. Tan M, Kek PC, Zhang RF, Lim A, Chandran M. Demographics and characteristics of singaporean patients with fragility fractures recruited into an osteoporosis disease management program. Endocr Rev. 2011;32(3).

223. Warriner AH, Patkar NM, Yun H, Delzell E. Minor, Major, Low-Trauma, and hightrauma fractures: What are the subsequent fracture risks and how do they vary? Current Osteoporosis Reports. 2011;9(3):122-8.

224. Collins GS, Mallett S, Altman DG. Predicting risk of osteoporotic and hip fracture in the United Kingdom: prospective independent and external validation of QFractureScores. BMJ (Clinical research ed). 2011;342:d3651.

225. Graafmans WC, Ooms ME, Bezemer PD, Bouter LM, Lips P. Different risk profiles for hip fractures and distal forearm fractures: a prospective study. Osteoporos Int. 1996;6(6):427-31.

226. Mehta SP, MacDermid JC, Richardson J, MacIntyre NJ, Grewal R. A Structured Literature Synthesis to Identify Measures for Screening for the Risk of Adverse Outcomes in Individuals Following Distal Radius Fracture. Critical Reviews[™] in Physical and Rehabilitation Medicine. 2014;26(3-4).

227. Morin SN, Lix LM, Leslie WD. The importance of previous fracture site on osteoporosis diagnosis and incident fractures in women. J Bone Miner Res. 2014;29(7):1675-80.

228. Nazrun AS, Tzar MN, Mokhtar SA, Mohamed IN. A systematic review of the outcomes of osteoporotic fracture patients after hospital discharge: morbidity, subsequent fractures, and mortality. Ther Clin Risk Manag. 2014;10:937.

229. Ratti C, Vulcano E, La Barbera G, Canton G, Murena L, Cherubino P. The incidence of fragility fractures in Italy. Aging Clin Exp Res. 2013;25(1):13-4.

230. Van Staa TP, Leufkens HGM, Cooper C. Does a fracture at one site predict later fractures at other sites? A British cohort study. Osteoporos Int. 2002;13(8):624-9.

231. van Helden S, Cals J, Kessels F, Brink P, Dinant GJ, Geusens P. Risk of new clinical fractures within 2 years following a fracture. Osteoporos Int. 2006;17(3):348-54.

232. Mallmin H, Ljunghall S, Persson I, Naessén T, Krusemo U-B, Bergström R. Fracture of the distal forearm as a forecaster of subsequent hip fracture: a population-based cohort study with 24 years of follow-up. Calcif Tissue Int. 1993;52(4):269-72.

233. McKay SD, MacDermid JC, Roth JH, Richards RS. Assessment of complications of distal radius fractures and development of a complication checklist. The Journal of hand surgery. 2001;26(5):916-22.

234. Kreder HJ, Hanel DP, McKee M, Jupiter J, McGillivary G, Swiontkowski MF. X-ray film measurements for healed distal radius fractures. The Journal of hand surgery. 1996;21(1):31-9.

235. O'Malley MP, Rodner C, Ritting A, Cote MP, Leger R, Stock H, et al. Radiographic interpretation of distal radius fractures: visual estimations versus digital measuring techniques. Hand. 2014;9(4):488-93.

236. Abramo A, Kopylov P, Tagil M. Evaluation of a treatment protocol in distal radius fractures: a prospective study in 581 patients using DASH as outcome. Acta Orthop. 2008;79(3):376-85.

237. Watson NJ, Asadollahi S, Parrish F, Ridgway J, Tran P, Keating JL. Reliability of radiographic measurements for acute distal radius fractures. BMC Med Imaging. 2016;16(1):44.

238. Katz MA, Beredjiklian PK, Bozentka DJ, Steinberg DR. Computed tomography scanning of intra-articular distal radius fractures: does it influence treatment? J Hand Surg Am. 2001;26(3):415-21.

239. Naor E, Di Segni V, Robin G, Makin M, Menczel J. Intra-observer variability in the determination of the metacarpal cortical index. Br J Radiol. 1972;45(531):213-7.

240. Dequeker J. Quantitative radiology: radiogrammetry of cortical bone. Br J Radiol. 1976;49(587):912-20.

241. Ishikawa K, Ohta T. Radial and metacarpal bone mineral density and calcaneal quantitative ultrasound bone mass in normal Japanese women. Calcif Tissue Int. 1999;65(2):112-6.

242. Hyldstrup L, Nielsen SP. Metacarpal index by digital X-ray radiogrammetry: normative reference values and comparison with dual X-ray absorptiometry. Journal of clinical densitometry : the official journal of the International Society for Clinical Densitometry. 2001;4(4):299-306.

243. Cummings SR, Melton LJ. Epidemiology and outcomes of osteoporotic fractures. Lancet. 2002;359(9319):1761-7.

244. Huachou Z, Kitazawa A, Kushida K, Nagano A. Longitudinal study of age- and menopause-related metacarpal index changes in Japanese adult females. J Clin Densitom. 2001;4(1):43-9.

245. Seo HJ, Kim SG, Kim CS. Risk factors for bone mineral density at the calcaneus in 40-59 year-old male workers: a cross-sectional study in Korea. BMC Public Health. 2008;8:253.
246. Kroger H, Tuppurainen M, Honkanen R, Alhava E, Saarikoski S. Bone mineral density and risk factors for osteoporosis--a population-based study of 1600 perimenopausal women. Calcif Tissue Int. 1994;55(1):1-7.

247. Clayton RA, Gaston MS, Ralston SH, Court-Brown CM, McQueen MM. Association between decreased bone mineral density and severity of distal radial fractures. J Bone Joint Surg Am. 2009;91(3):613-9.

248. Excellence NIfHaC. Osteoporosis: fragility fracture risk. Costing report. Implementing NICE guidance 2012 [Available from: <u>http://healthdocbox.com/Senior_Health/67279098-Osteoporosis-fragility-fracture-risk-costing-report-implementing-nice-guidance.html</u>.

249. Drake MT, Clarke BL, Khosla S. Bisphosphonates: mechanism of action and role in clinical practice. Mayo Clin Proc. 2008;83(9):1032-45.

250. Kimura K. Bilateral asymmetry of cross-sectional radiogrametric and microdensitometric measurements in the second metacarpal. Okajimas Folia Anat Jpn. 1990;67(1):39-45.

251. Gittoes N, McLellan A, Cooper A, Dockery F, Davenport G, Goodwin V. Effective secondary prevention of fragility fractures: clinical standards for fracture liaison services. National Osteoporosis Society, Camerton. 2015.

252. Miller SW, Evans JG. Fractures of the distal forearm in Newcastle: an epidemiological survey. Age Ageing. 1985;14(3):155-8.

253. Kaukonen JP. Fractures of the distal forearm in the Helsinki district. Ann Chir Gynaecol. 1985;74(1):19-21.

254. Bengnér U, Johnell O. Increasing incidence of forearm fractures. A comparison of epidemiologic patterns 25 years apart. Acta Orthop Scand. 1985;56(2):158.

255. Melton LJ, 3rd, Amadio PC, Crowson CS, O'Fallon WM. Long-term trends in the incidence of distal forearm fractures. Osteoporos Int. 1998;8(4):341-8.

256. Association BO. The Care of patients with fragility fracture. 2007.

257. Ralis ZA. Epidemics of fractures during periods of snow and ice. Br Med J (Clin Res Ed). 1986;293(6545):484.

258. Johansen A, Evans RJ, Stone MD, Richmond PW, Lo SV, Woodhouse KW. Fracture incidence in England and Wales: a study based on the population of Cardiff. Injury. 1997;28(9-10):655-60.

259. Skinner J, Weinstein JN, Sporer SM, Wennberg JE. Racial, ethnic, and geographic disparities in rates of knee arthroplasty among Medicare patients. N Engl J Med. 2003;349(14):1350-9.

260. Wennberg JE. Unwarranted variations in healthcare delivery: implications for academic medical centres. BMJ: British Medical Journal. 2002;325(7370):961.

261. Shain M, Roemer MI. Hospital costs relate to the supply of beds. J Occup Environ Med. 1959;1(9):518.

262. Hiscock R, Bauld L, Amos A, Platt S. Smoking and socioeconomic status in England: the rise of the never smoker and the disadvantaged smoker. Journal of Public Health. 2012;34(3):390-6.

263. Freeman A, Tyrovolas S, Koyanagi A, Chatterji S, Leonardi M, Ayuso-Mateos JL, et al. The role of socio-economic status in depression: results from the COURAGE (aging survey in Europe). BMC Public Health. 2016;16(1):1098.

264. O'Loughlin JL, Robitaille Y, Boivin JF, Suissa S. Incidence of and risk factors for falls and injurious falls among the community-dwelling elderly. Am J Epidemiol. 1993;137(3):342-54.
265. Tinetti ME, Speechley M. Prevention of falls among the elderly. N Engl J Med. 1989;320(16):1055-9.

266. Tinetti ME. Factors associated with serious injury during falls by ambulatory nursing home residents. J Am Geriatr Soc. 1987;35(7):644-8.

267. Health Do. National Service Framework for Older People. 2001.

268. Gillespie LD, Robertson MC, Gillespie WJ, Lamb SE, Gates S, Cumming RG, et al. Interventions for preventing falls in older people living in the community. Cochrane Database Syst Rev. 2009;2(CD007146).

269. Costello E, Edelstein JE. Update on falls prevention for community-dwelling older adults: review of single and multifactorial intervention programs. J Rehabil Res Dev. 2008;45(8):1135-52.

270. Jackson G, Thornley S, Woolston J, Papa D, Bernacchi A, Moore T. Reduced acute hospitalisation with the healthy housing programme. J Epidemiol Community Health. 2011;65(7):588-93.

271. Leipzig RM, Cumming RG, Tinetti ME. Drugs and falls in older people: a systematic review and meta-analysis: I. Psychotropic drugs. J Am Geriatr Soc. 1999;47(1):30-9.

272. Ray WA, Thapa PB, Gideon P. Benzodiazepines and the risk of falls in nursing home residents. J Am Geriatr Soc. 2000;48(6):682-5.

273. Robbins AS, Rubenstein LZ, Josephson KR, Schulman BL, Osterweil D, Fine G. Predictors of falls among elderly people. Results of two population-based studies. Arch Intern Med. 1989;149(7):1628-33.

274. Friedman SM, Munoz B, West SK, Rubin GS, Fried LP. Falls and fear of falling: which comes first? A longitudinal prediction model suggests strategies for primary and secondary prevention. J Am Geriatr Soc. 2002;50(8):1329-35.

275. Guideline for the prevention of falls in older persons. American Geriatrics Society, British Geriatrics Society, and American Academy of Orthopaedic Surgeons Panel on Falls Prevention. J Am Geriatr Soc. 2001;49(5):664-72.

276. Lawlor DA, Patel R, Ebrahim S. Association between falls in elderly women and chronic diseases and drug use: cross sectional study. BMJ. 2003;327(7417):712-7.

277. Gregg EW, Pereira MA, Caspersen CJ. Physical activity, falls, and fractures among older adults: a review of the epidemiologic evidence. J Am Geriatr Soc. 2000;48(8):883-93.

278. Arfken CL, Lach HW, Birge SJ, Miller JP. The prevalence and correlates of fear of falling in elderly persons living in the community. Am J Public Health. 1994;84(4):565-70.

279. Graafmans WC, Ooms ME, Hofstee HM, Bezemer PD, Bouter LM, Lips P. Falls in the elderly: a prospective study of risk factors and risk profiles. Am J Epidemiol. 1996;143(11):1129-36.

280. Feder G, Cryer C, Donovan S, Carter Y. Guidelines for the prevention of falls in people over 65. The Guidelines' Development Group. BMJ. 2000;321(7267):1007-11.

281. Tinetti ME, Mendes de Leon CF, Doucette JT, Baker DI. Fear of falling and fall-related efficacy in relationship to functioning among community-living elders. J Gerontol. 1994;49(3):M140-7.

282. Ebrahim S, Thompson PW, Baskaran V, Evans K. Randomized placebo-controlled trial of brisk walking in the prevention of postmenopausal osteoporosis. Age Ageing. 1997;26(4):253-60.

283. Skelton DA. Effects of physical activity on postural stability. Age Ageing. 2001;30 Suppl 4:33-9.

284. Tinetti ME, Speechley M, Ginter SF. Risk factors for falls among elderly persons living in the community. N Engl J Med. 1988;319(26):1701-7.

285. Bath PA, Morgan K. Differential risk factor profiles for indoor and outdoor falls in older people living at home in Nottingham, UK. Eur J Epidemiol. 1999;15(1):65-73.

286. Kelsey JL, Procter-Gray E, Hannan MT, Li W. Heterogeneity of falls among older adults: implications for public health prevention. Am J Public Health. 2012;102(11):2149-56.

287. Li W, Keegan THM, Sternfeld B, Sidney S, Quesenberry Jr CP, Kelsey JL. Outdoor falls among middle-aged and older adults: a neglected public health problem. Am J Public Health. 2006;96(7):1192-200.

288. Keegan TH, Kelsey JL, King AC, Quesenberry CP, Jr., Sidney S. Characteristics of fallers who fracture at the foot, distal forearm, proximal humerus, pelvis, and shaft of the tibia/fibula compared with fallers who do not fracture. Am J Epidemiol. 2004;159(2):192-203.

289. Skelton D, Todd C. What are the main risk factors for falls amongst older people and what are the most effective interventions to prevent these falls? How should interventions to prevent falls be implemented. World Health Organization, Europe. 2004.

290. Foundation JR. UK Poverty 2017 2017 [Available from:

https://www.jrf.org.uk/report/uk-poverty-2017.

291. Rabar S, Lau R, O'Flynn N, Li L, Barry P, Guideline Development G. Risk assessment of fragility fractures: summary of NICE guidance. BMJ. 2012;345:e3698.

292. Marsh D, Åkesson K, Beaton DE, Bogoch ER, Boonen S, Brandi ML, et al. Coordinatorbased systems for secondary prevention in fragility fracture patients. Osteoporos Int. 2011;22(7):2051-65.

293. Mitchell PJ. Fracture liaison services: the UK experience. Osteoporos Int. 2011;22(3):487-94.

294. Edwards BJ, Bunta AD, Simonelli C, Bolander M, Fitzpatrick LA. Prior fractures are common in patients with subsequent hip fractures. Clin Orthop Relat Res. 2007;461:226-30.
295. Cooper C, Mitchell P, Kanis JA. Breaking the fragility fracture cycle. Osteoporos Int. 2011;22(7):2049-50.

296. Lyles KW, Colón-Emeric CS, Magaziner JS, Adachi JD, Pieper CF, Mautalen C, et al. Zoledronic acid and clinical fractures and mortality after hip fracture. N Engl J Med. 2007;357(18):1799-809.

297. Silverman SL, Watts NB, Delmas PD, Lange JL, Lindsay R. Effectiveness of bisphosphonates on nonvertebral and hip fractures in the first year of therapy: the risedronate and alendronate (REAL) cohort study. Osteoporos Int. 2007;18(1):25-34.

298. Majumdar SR, Johnson JA, McAlister FA, Bellerose D, Russell AS, Hanley DA, et al. Multifaceted intervention to improve diagnosis and treatment of osteoporosis in patients with recent wrist fracture: a randomized controlled trial. CMAJ : Canadian Medical Association journal = journal de l'Association medicale canadienne. 2008;178(5):569-75.

299. Majumdar SR, Lier DA, Rowe BH, Russell AS, McAlister FA, Maksymowych WP, et al. Cost-effectiveness of a multifaceted intervention to improve quality of osteoporosis care after wrist fracture. Osteoporos Int. 2011;22(6):1799-808.

300. Meunier PJ, Roux C, Seeman E, Ortolani S, Badurski JE, Spector TD, et al. The effects of strontium ranelate on the risk of vertebral fracture in women with postmenopausal osteoporosis. N Engl J Med. 2004;350(5):459-68.

301. Skelton D, Neil F. NHS Greater Glasgow and Clyde Strategy for Osteoporosis and Falls
 Prevention 2006–2010: an evaluation 2007–2009. Glasgow Caledonian University, Glasgow.
 2009.

302. Johansen A, Wakeman R, Boulton C, Plant F, Roberts J, Williams A. National hip fracture database: national report 2013. London: Royal College of Physicians. 2013.

303. Roberts D, de Souza V. A venue-based analysis of the reach of a targeted outreach service to deliver opportunistic community NHS Health Checks to 'hard-to-reach'groups. Public Health. 2016;137:176-81.

304. Woringer M, Cecil E, Watt H, Chang K, Hamid F, Khunti K, et al. Evaluation of community provision of a preventive cardiovascular programme-the National Health Service Health Check in reaching the under-served groups by primary care in England: cross sectional observational study. BMC Health Serv Res. 2017;17(1):405.